#### FRANCHISE DISCLOSURE DOCUMENT



Mrs. Fields Franchising, LLC
a Delaware limited liability company
1717 S. 4800 W.
Salt Lake City, Utah 84104
(801) 412-8890
https://www.mrsfields.com/franchise/
Generalquestions@famousbrandsintl.com

We offer single and multiple unit franchises to own and operate Mrs. Fields-branded retail outlets in two different formats: a store and a kiosk. Both formats involve the sale of Mrs. Fields-branded cookies and related products.

The total investment necessary to begin operation of a retail store ranges from \$311,500 to \$495,850. This includes \$35,000 that must be paid to us or our affiliate. The total investment necessary to begin operation of a retail kiosk ranges from \$188,860 to \$290,310. This includes \$35,000 that must be paid to us or our affiliate. If you sign an Area Development Agreement for multiple units, you also must pay us \$12,500 for each Store you commit to develop, which amount will be credited against the Initial Franchise Fee.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.** 

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact our Development Department, at 1717 S. 4800 W., Salt Lake City, Utah 84104, (801) 412-8890 or <a href="mailto:Generalquestions@famousbrandsintl.com">Generalquestions@famousbrandsintl.com</a>.

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at <a href="https://www.ftc.gov">www.ftc.gov</a> for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: April 30, 2025

#### **How to Use This Franchise Disclosure Document**

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

| QUESTION                                                                          | WHERE TO FIND INFORMATION                                                                                                                                                                                                                                 |  |  |
|-----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| How much can I earn?                                                              | Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit F. |  |  |
| How much will I need to invest?                                                   | Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.                                                                  |  |  |
| Does the franchisor have the financial ability to provide support to my business? | Item 21 or Exhibit A includes financial statements. Review these statements carefully.                                                                                                                                                                    |  |  |
| Is the franchise system stable, growing, or shrinking?                            | Item 20 summarizes the recent history of the number of company-owned and franchised outlets.                                                                                                                                                              |  |  |
| Will my business be the only Mrs. Fields business in my area?                     | Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.                                                                                                             |  |  |
| Does the franchisor have a troubled legal history?                                | Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.                                                                                                                      |  |  |
| What's it like to be a Mrs. Fields franchisee?                                    | Item 20 or Exhibit F lists current and former franchisees. You can contact them to ask about their experiences.                                                                                                                                           |  |  |
| What else should I know?                                                          | These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.                                                   |  |  |

#### What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

**Business model can change**. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

**Renewal**. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

#### **Some States Require Registration**

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit G.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

#### Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. Out-of-State Dispute Resolution. The franchise agreement requires you to resolve disputes with the franchisor by arbitration and/or litigation only in Delaware. Out-of-state arbitration or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to arbitrate or litigate with the franchisor in Delaware than in your own state.
- 2. **Spousal Liability.** Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

#### STATE OF MICHIGAN DISCLOSURE NOTICE

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in the Michigan Franchise Act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise before the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than 5 years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on the terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
  - (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.
  - (ii) The fact that the proposed transferee is a competitor of the franchisor or sub-franchisor.

- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of these assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

# THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to: Michigan Attorney General's Office, Consumer Protection Division, Attn: Franchise Section, G. Mennen Williams Building – 1<sup>st</sup> Floor, 525 West Ottawa Street, Lansing, Michigan 48933, Telephone Number: (517) 335-7567.

THE MICHIGAN NOTICE APPLIES ONLY TO FRANCHISEES WHO ARE RESIDENTS OF MICHIGAN OR LOCATE THEIR FRANCHISES IN MICHIGAN.

Note: Despite paragraph (f) above, we intend, and we and you agree to fully enforce the arbitration provisions of the Franchise Agreement. We believe that paragraph (f) is unconstitutional and cannot preclude us from enforcing these arbitration provisions.

#### TABLE OF CONTENTS

| <u>Item</u>     |                                                          | <u>Page</u> |
|-----------------|----------------------------------------------------------|-------------|
| ITEM 1 THE F    | RANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES  | 1           |
| ITEM 2 BUSIN    | VESS EXPERIENCE                                          | 3           |
| ITEM 3 LITIG    | ATION                                                    | 4           |
| ITEM 4 BANK     | RUPTCY                                                   | 4           |
| ITEM 5 INITIA   | AL FEES                                                  | 5           |
| ITEM 6 OTHE     | R FEES                                                   | 6           |
| ITEM 7 ESTIM    | IATED INITIAL INVESTMENT                                 | 9           |
| ITEM 8 RESTI    | RICTIONS ON SOURCES OF PRODUCTS AND SERVICES             | 12          |
|                 | CHISEE'S OBLIGATIONS                                     |             |
| ITEM 10 FINA    | NCING                                                    | 15          |
| ITEM 11 FRAN    | NCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND |             |
|                 | NING.                                                    | 15          |
|                 | RITORY                                                   |             |
|                 | DEMARKS                                                  |             |
|                 | ENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION            |             |
|                 | GATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE     | 20          |
|                 | NCHISE BUSINESS                                          | 27          |
|                 | TRICTIONS ON WHAT THE FRANCHISEE MAY SELL                |             |
|                 | EWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION      |             |
|                 | LIC FIGURES.                                             |             |
|                 | NCIAL PERFORMANCE REPRESENTATIONS                        |             |
|                 | LETS AND FRANCHISEE INFORMATION                          |             |
|                 | NCIAL STATEMENTS                                         |             |
|                 | TRACTS                                                   |             |
|                 | EIPTS                                                    |             |
| 1121112511261   | 31 10                                                    |             |
|                 |                                                          |             |
| <b>EXHIBITS</b> |                                                          |             |
|                 |                                                          |             |
| Exhibit A –     | Financial Statements and Guarantee of Performance        |             |
| Exhibit B –     | Franchise Agreement                                      |             |
| Exhibit C –     | Area Development Agreement                               |             |
| Exhibit D –     | Form Deposit Agreement                                   |             |
| Exhibit E –     | Table of Contents of Operations Manual                   |             |
| Exhibit F –     | Lists of Current and Former Franchisees                  |             |
| Exhibit G –     | State Administrators / Agents for Service of Process     |             |
| Exhibit H –     | State-Specific Addenda                                   |             |
| Exhibit I –     | Sample Release Form                                      |             |
| Exhibit J –     | Disclosure Acknowledgment Agreement                      |             |
| Exhibit K –     | State Effective Dates and Receipts                       |             |

## ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

To simplify the language in this disclosure document, "we" means Mrs. Fields Franchising, LLC, the franchisor, and "you" means the individual, corporation, or other entity that buys a franchise and signs a Franchise Agreement. If you are a corporation or an entity, a reference to "you" also includes your owners because your owners must sign a personal guaranty and assumption of your obligations.

#### The Franchisor

We conduct business under the name Mrs. Fields Franchising, LLC, and various trade names, including "Mrs. Fields" and "Mrs. Fields Cookies." We are a Delaware limited liability company formed on February 4, 2004, and our principal business address is 1717 S. 4800 W., Salt Lake City, Utah 84104. We disclose our agents for service of process in Exhibit G.

We offer, sell and service franchises for Stores (as defined below). We do not currently own or operate any Stores. We have offered franchises for Stores from March 2004 through March 2015 and from January 2017 to the present. We have never offered franchises for any other concepts, but we may do so in the future.

#### Our Parent, Affiliates and Predecessors

We are a wholly owned subsidiary of Mrs. Fields Franchising Holdco, LLC ("MFF Holdco"). MFF Holdco is a wholly owned subsidiary of Famous Brands, LLC ("FB"), which was formerly Mrs. Fields Famous Brands, LLC and previously owned and operated Stores, but currently owns none. FB is a wholly owned subsidiary of Famous Brands International, LLC ("FBI"), which was formerly Mrs. Fields Original Cookies, Inc. but which converted to a limited liability company on June 26, 2020 and changed its name on June 7, 2024. FBI is a wholly owned subsidiary of Famous Brands International Parent, LLC ("FBIP"), which is a wholly owned subsidiary of FB Intermediate Holdco, LLC ("Holdco"), which is a wholly owned subsidiary of Famous Brands International Holdings, LLC ("FBH"). Each of MFF Holdco, FB and FBI shares our principal business address. The principal business address for each of FBIP, Holdco and FBH is 1330 Avenue of the Americas, 16<sup>th</sup> Floor, New York, NY 10019. On October 19, 2023, FBIP acquired Mrs. Fields Franchising, LLC and TCBY Systems, LLC (the "Transaction"). As a result, we are now owned and controlled, directly and indirectly, by FBIP, Holdco and FBH. Subsequent to the Transaction, on August 1, 2024, FB completed a restructuring in which MFF Holdco was established and FB transferred all of its interest in us to MFF Holdco.

Our affiliate, TCBY Systems, LLC ("TCBY"), is a Delaware limited liability company formed on May 30, 2000, and it shares our principal business address. TCBY sold franchises for TCBY-branded frozen yogurt stores from June 2000 to March 2015. It again began selling TCBY-branded franchises in January 2017 and has continued to do so since then. As of December 31, 2024 (the end of our most recent fiscal year), TCBY was the franchisor of 113 TCBY-branded outlets. In addition, from June 2016 through September 2017, TCBY offered franchises to own and operate a TCBY®-branded vending machine. The vending machines sell and dispense assorted TCBY®-branded frozen yogurt products and toppings to individual customers. As of December 31, 2024, TCBY had 2 vending machines in operation, each of which is franchised. Through September 2011, TCBY also previously offered the opportunity to enter into an area director franchise. TCBY currently has 1 area director in operation. Other than granting renewal franchises, TCBY currently has no plans to offer franchises for vending machines or area directors in the future.

Our affiliate, Famous Brands Franchising, LLC ("FBF") sold franchises under both the Mrs. Fields and TCBY brands from April 2015 to December 2016. In January 2017, FBF assigned all of its interests in the franchises (i) for TCBY-branded stores to TCBY and (ii) for Stores to us. FBF shares our principal business address. As of the date of this disclosure document, FBF does not offer, operate or own franchises in any line of business.

Other than as disclosed above, no parents, predecessors, or affiliates are required to be disclosed in this Item. Other than as disclosed above, none of our parents or affiliates has owned or offered franchises in any line of business or conducted the type of business that is the subject of the franchise.

#### Description of the Franchise Offered

We currently offer franchises for retail stores ("Stores") and Kiosks (as defined below) identified by the Marks (as defined below), which offer those products we or our affiliates periodically require or approve, including specialty snacks and other bakery items, desserts, and beverages such as cookies, brownies, cakes, muffins, bagels, croissants, cinnamon rolls, sticky buns, chocolate, chocolate-covered fruit, dried fruit, nuts, candies, popcorn and coffee ("Approved Products"). Unless otherwise noted, the term "Store" refers to both Stores and Kiosks. The term "Kiosks" refers to franchised Stores with a smaller footprint than average Stores, typically between 100 and 300 square feet. Kiosks are typically a free-standing structure, generally located in corridors of traditional malls, strip malls, and lifestyle centers. All Stores are identified by the "Marks," which include trademarks, trade names, service marks, logos and other commercial symbols that we periodically authorize franchisees to use to identify their Stores and the Approved Products, including the trademarks and service marks Mrs. Fields® and Mrs. Fields Cookies®, and the associated trade dress. Certain franchisees have been permitted to operate TCBY-branded outlets in conjunction with Stores as part of a co-branded location. These franchisees enter into a separate franchise agreement with TCBY.

You will acquire a franchise to develop, own and operate a single Store by signing a franchise agreement (a "Franchise Agreement"). A copy of our current form of Franchise Agreement is attached to this disclosure document as <a href="Exhibit B">Exhibit B</a>. If you enter into a Franchise Agreement, you will be authorized to use the "System" in the operation of your Store. The System includes our business formats, signs, equipment, methods, procedures, designs, layouts, standards, and specifications, including the use of the Marks.

You also may enter into an area development agreement (an "Area Development Agreement") under which you may acquire the exclusive right to open a certain number of Stores over a defined period of time in a defined area, as we determine, on the basis of the market potential and the size of the designated area. A copy of our current form of Area Development Agreement is attached to this disclosure document as <a href="Exhibit C">Exhibit C</a>. The term of your Area Development Agreement generally will not be longer than 3-5 years, depending on the number of Stores you agree to open, and generally will require you to open your first Store within 12 months from the date of your Area Development Agreement and open your other Stores as we determine based on the number of Stores you agree to open, usually within a 5 year period.

#### Market and Competition

The market for your business is the general public. Sales at Stores tend to be seasonal, with the highest sales typically occurring during November and December. The difference in sales from season to season varies depending upon the climate where a particular Store is located. The retail cookie industry is mature and highly competitive.

Stores compete with all other stores, restaurants and retail outlets that offer baked goods, desserts and snack items. You will also compete with convenience stores, quick service restaurants, fast food chains and internet catalog sellers. The market for the products is well developed and competitive.

Some of these competitive stores or outlets are or may be owned, managed, franchised or licensed by us or one of our affiliates. Your competition includes other existing Mrs. Fields franchisees that may operate, or have the right to operate, Stores under different formats or who have entered into agreements with us that contain terms significantly different than those in the Franchise Agreement. Competition may also include sales made by one of our affiliates through alternative channels of distribution, such as Internet sales, gifting, and vending machines. We and our affiliates may enter into co-branding arrangements and these activities may compete with you.

#### Laws and Regulations

In addition to laws and regulations that apply to business generally, Stores may be subject to federal, state, and local laws and regulations pertaining to food labeling, sanitation, and weights and measurement. You should also be aware of federal, state, and local employment laws and regulations, specifically including minimum age and wage requirements. Local law requirements vary by location.

Some states may require franchisees to obtain restaurant, business, occupational, food products, and miscellaneous licenses. Some states also have laws regarding who may secure these licenses. You may also have to obtain health licenses and to comply with health laws and regulations that apply to restaurant and food product sales establishments. You should inquire about these laws and regulations.

We strongly suggest that you consult with an attorney, consultant and/or financial advisor regarding such laws and regulations prior to purchasing a franchise from us. Applicable laws and regulations are subject to change.

#### ITEM 2 BUSINESS EXPERIENCE

#### James Carnrite: Interim Chief Executive Officer

Mr. Carnrite has been our Interim Chief Executive Officer since April 2025. Mr. Carnrite also serves as Interim Chief Executive Officer of TCBY and has been the President of JKC Supply Chain Services in Wellington, Florida since November 2022. Prior to that, Mr. Carnrite was the Vice President Operations for Twin Star International from August 2017 to November 2022 in Boca Raton, Florida. Mr. Carnrite primarily works from New York, New York.

#### Richard T. Hankins: Senior Director of Development and Real Estate

Mr. Hankins is employed by our parent, FB, and has been our, TCBY's and FB's Senior Director of Development and Real Estate since June 2015. Mr. Hankins primarily works from Bellwood, Pennsylvania.

#### Brian Mooney: Senior Director of Franchise Operations

Brian Mooney has been our and TCBY's Senior Director of Franchise Operations since October 2016, working primarily from Virginia Beach, Virginia.

#### Corie King: Director of Marketing

Corie King has been our and TCBY's Director of Marketing since March 2017, working primarily from Broomfield, Colorado.

#### Stacey Bertke: Director of Supply Chain Planning

Stacey Bertke has been our and TCBY's Director of Supply Chain Planning since August 2022. From July 2020 to July 2022, Ms. Bertke served as Director of Purchasing for Schaffer Partners in Cleveland, Ohio. From April 2020 to June 2020, Ms. Bertke was between positions. Prior to that, Ms. Bertke served as Senior Buyer, General Merchandise Manager, Divisional Merchandise Manager, Buyer and Product Manager for Enesco Properties LLC from October 2009 to March 2020 in Richmond Heights, Ohio. Ms. Bertke primarily works from Cleveland, Ohio.

#### Chris Sabido: Director of Training, Talent & Culture

Chris Sabido has been our and TCBY's Director of Training, Talent & Culture since December 2023. From December 2019 to November 2023, Mr. Sabido was General Manager for Whole Foods Market in Winter Park, Florida. From February 2017 to December 2019, Mr. Sabido was Associate General Manager for Whole Foods Market in Gainesville, Florida. Mr. Sabido primarily woks from Orlando, Florida.

#### Dawn Capobianco: Controller

Dawn Capobianco has been our and TCBY's Controller since May 2024. From April 2023 to February 2024, Ms. Capobianco was Financial Controller for Partners Coffee in Brooklyn, New York. Prior to that, Ms. Capobianco was Senior Manager Finance at Pomptonian Food Service from November 2023 to March 2023. From August 2022 to October 2022, Ms. Capobianco was between positions. From January 2022 to July 2022, Ms. Capobianco was Controller for JJ Nuts in Fairfield, New Jersey. From May 2021 to December 2021, Ms. Capobianco was between positions. Prior to that, Ms. Capobianco was Vice President of Finance and Human Resources at Innovare Medical Media in Branchburg, New Jersey from September 2017 to April 2021.

## ITEM 3 LITIGATION

No litigation is required to be disclosed in this Item.

### ITEM 4 BANKRUPTCY

No bankruptcy is required to be disclosed in this Item.

### ITEM 5 INITIAL FEES

We offer an individual unit franchise and territory development rights under a multiple unit or area development franchise.

#### Individual Unit Franchise Agreement

The "Initial Franchise Fee" for an individual unit franchise is \$35,000 for each Store. You will pay the Initial Franchise Fee, after applying the Deposit described below (if applicable), when you sign the Franchise Agreement.

The Initial Franchise Fee is \$25,000 if you are an honorably discharged veteran of the United States military or if you are an existing franchisee. We also offer a first responders discount for law enforcement officers, firefighters, and emergency medical technicians who have at least 10 years of experience. Qualified first responders will be eligible for a discount of 20% of the Initial Franchise Fee. This discount does not apply to a renewal fee. If you choose to operate a TCBY-branded outlet in conjunction with your Store as part of a co-branded location, you will be eligible for a discount of 50% of the Initial Franchise Fee for the TCBY-branded outlet. We may modify or eliminate these discounted Initial Franchise Fee programs at any time.

Except as described above, the Initial Franchise Fee is uniform and deemed fully earned by us when we sign the Franchise Agreement and is not refundable under any circumstances.

During fiscal year 2024, we charged Initial Franchise Fees ranging from \$0 to \$35,000.

Before signing the Franchise Agreement, you will enter into a deposit agreement with us in the form attached as  $\underline{\text{Exhibit D}}$ . Under the Deposit Agreement, you must make a deposit payment of \$10,000 ("Deposit") in connection with assistance we will provide to you in locating a site. If you then enter into a Franchise Agreement with us, we will apply the Deposit toward the Initial Franchise Fee. If we do not approve you as an applicant, we will refund the Deposit. The Deposit is otherwise not refundable.

#### Area Development Agreement

If you sign an Area Development Agreement, you will pay a lump sum "Development Fee" when you sign the Area Development Agreement. Unless you are an existing franchisee (see below), the Development Fee is calculated at the rate of \$12,500 for each Store that you are authorized to develop after the first franchise. In addition to the Development Fee, you must sign a Franchise Agreement and pay the Initial Franchise Fee for your first Store at the same time you sign the Area Development Agreement. You will be required to sign a Franchise Agreement for each additional Store and pay the remaining portion of the Initial Franchise Fee of \$12,500 at least 4 months before each Store is scheduled to open or when you sign the lease for the Store, whichever occurs first. For example, if you enter into an Area Development Agreement to establish your first Store and four additional Stores, you will sign a Franchise Agreement and pay an Initial Franchise Fee of \$35,000 for your first Store and a Development Fee of \$50,000 for the right to develop four additional Stores. Development Fees will be applied to the Initial Franchise Fees for each additional Store opened under your Area Development Agreement. The Development Fee is not refundable under any circumstances.

If you are a current Mrs. Fields franchisee, you will pay a Development Fee at the rate of \$12,500 (one-half of the Initial Franchise Fee of \$25,000) for each Store that you are authorized to develop when you sign the Area Development Agreement. You will be required to sign a Franchise Agreement for each

Store and pay the remaining portion of the Initial Franchise Fee of \$12,500 at least 4 months before your Store is scheduled to open or before you sign the lease for the Store, whichever occurs first.

# ITEM 6 OTHER FEES

| Type of Fee <sup>1</sup>         | Amount                                                                                                                                                                                                                                     | <b>Due Date</b>                                                                                                                                                  | Remarks                                                                                                                                                                                                                                                                                                                                                            |
|----------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Royalty <sup>2</sup>             | 6% of Gross<br>Revenue (as defined<br>below)                                                                                                                                                                                               | Weekly on or before<br>the close of business<br>on Wednesday of each<br>week based on your<br>Store's Gross<br>Revenue for the<br>immediately<br>preceding week. |                                                                                                                                                                                                                                                                                                                                                                    |
| Brand Fund<br>Contribution       | 3% of Gross<br>Revenue                                                                                                                                                                                                                     | Same as royalty fee                                                                                                                                              | See Note 3.                                                                                                                                                                                                                                                                                                                                                        |
| Local Store<br>Marketing         | Monthly minimum of \$2,000, beginning at month 3 of operations and continuing for 9 months after opening.  After the first year of operations, you must spend a minimum of 2% of Gross Revenue or \$1,000 per month, whichever is greater. | Monthly                                                                                                                                                          | See Note 4.                                                                                                                                                                                                                                                                                                                                                        |
| Technology Fee                   | None currently, but<br>may be charged in<br>the future (up to the<br>greater of \$100 per<br>month or 1% of<br>Gross Revenue).                                                                                                             | Monthly                                                                                                                                                          | We reserve the right to charge you a technology fee that we may use to research, develop, maintain, implement, update, upgrade or use System technology. We may adjust the amount of the Technology Fee once per calendar year upon 60 days' notice to you; provided, however that we will not increase the potential maximum fee more than 10% per calendar year. |
| Training Fee                     | None currently, but may be charged in the future                                                                                                                                                                                           | When incurred                                                                                                                                                    | We may charge a fee for certain training programs, including the training of replacement managers.                                                                                                                                                                                                                                                                 |
| Refresher/Additional<br>Training | Our then-current fee<br>for such training;<br>currently, \$500 per<br>day per person plus<br>travel expenses (up<br>to \$750 per day per<br>person)                                                                                        | When incurred                                                                                                                                                    | We may require you and/or previously trained and experienced managers and employees to attend periodic refresher or additional courses at the times and locations we designate.                                                                                                                                                                                    |

| Type of Fee <sup>1</sup>       | Amount                                                                                                        | <b>Due Date</b>                                                                                                | Remarks                                                                                                                                                                                                                                                                                                   |
|--------------------------------|---------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Computer System Compliance Fee | The greater of \$500 per month, or the then current monthly rate of the software license.                     | When incurred                                                                                                  | If you do not comply with the required Computer<br>System that we specify, you will be assessed a<br>Computer System Compliance Fee.                                                                                                                                                                      |
| Delinquent Opening<br>Fee      | \$1,000 per month                                                                                             | When incurred                                                                                                  | If you fail to obtain our approval of and secure the Premises for your Store within the 6-month period, at our sole discretion we may charge you a \$1,000 per month Delinquent Opening Fee until you execute a lease, or terminate the Franchise Agreement.                                              |
| Late Payment Fee               | \$100 for each delinquent payment                                                                             | When the delinquent payment is due.                                                                            |                                                                                                                                                                                                                                                                                                           |
| Late Reporting Fee             | \$100 for each<br>delinquent report                                                                           | When the delinquent report is due and continuing to be due for each period that the report remains delinquent. |                                                                                                                                                                                                                                                                                                           |
| Interest Expenses              | A rate equal to the<br>lesser of the highest<br>rate allowed under<br>applicable law or<br>1.5% per month     | Payable at the same time you make a delinquent payment.                                                        | Payable on any amount overdue to us.                                                                                                                                                                                                                                                                      |
| New Supplier<br>Testing        | The costs we incur in evaluating a proposed supplier                                                          | When incurred                                                                                                  | Due if you request our approval of a new proposed supplier.                                                                                                                                                                                                                                               |
| Advertising<br>Cooperative     | Not currently applicable but may be established in the future (and determined by members of the Cooperative). | If established, the timing of contributions will be established by members of the Cooperative.                 | If established, the amount of required contributions to the Cooperative will be determined by a majority vote of the Stores that are members of the Cooperative, with each member Store having one vote, regardless if the member is a franchisee-owned or company-owned Store (subject to our approval). |
| Audit                          | Cost of financial audit                                                                                       | 15 days after receipt of audit or inspection report.                                                           | You must pay the costs of the audit or inspection only if the audit results show an understatement of Gross Revenue of more than 2% or if the need for an audit was a result of your default under the Franchise Agreement in failing to provide records and reports in a timely manner.                  |
| Transfer Fee                   | 50% of then-current<br>Initial Franchise Fee<br>for existing<br>franchisees                                   | Payable before or upon final closing of transfer.                                                              | We will not charge a transfer fee if the transfer is of ownership interests among your existing Entity Owners (as defined in Item 15) and the names and identity of all Entity Owners remains the same following the transfer.                                                                            |
| Renewal Fee                    | 20% of then-current<br>Initial Franchise Fee                                                                  | Payable upon renewal                                                                                           | You must meet other conditions to renew the Franchise Agreement.                                                                                                                                                                                                                                          |
| Interim Management<br>Fees     | 10% of Gross<br>Revenue during the<br>period of<br>management                                                 | As incurred                                                                                                    | Incurred if we elect to manage your Store pending our purchase of that Store, or we assume management of your Store in the case of your voluntary abandonment.                                                                                                                                            |

| Type of Fee <sup>1</sup>                                                             | Amount                                                                                                                                             | <b>Due Date</b>                                                                      | Remarks                                                                                                                                                                                                                                              |
|--------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Costs and Attorneys'<br>Fees;<br>Indemnification                                     | Will vary under circumstances                                                                                                                      | Upon occurrence                                                                      | If we or an affiliate prevail in any proceeding or litigation against you, you must pay the costs and attorneys' fees incurred. You and each of your Entity Owners also have indemnification obligations to us and our affiliates.                   |
| Lost Revenue<br>Damages                                                              | Will vary under the circumstance                                                                                                                   | Within 15 days of our termination of your Franchise Agreement based on your default. | Payable only if we terminate your Franchise Agreement for your breach. See Note 5.                                                                                                                                                                   |
| National Convention<br>and Regional<br>Meetings Fee                                  | A reasonable fee for each of your attendees, although, currently, we do not charge you this fee. Up to \$1,000 per attendee.                       | Upon demand                                                                          | You or at least one of your Entity Owners (if you are an Entity) and, when we request, the manager of Your Store and/or an approved trainer if you are a multi-unit franchisee must attend all mandatory national conventions and regional meetings. |
| Fee for Failing to<br>Attend Mandatory<br>National Convention<br>or Regional Meeting | A reasonable fee for you and each of your required representatives, although, currently, we do not charge you this fee. Up to \$1,000 each person. | Upon demand                                                                          |                                                                                                                                                                                                                                                      |

#### General Comments:

- 1. All fees are uniformly imposed (unless otherwise noted), payable to us or our affiliate and are non-refundable except as explained below. If we or our affiliates do not actually receive your payments on the due date, they will be deemed delinquent.
- 2. We will determine from time to time the method by which you must pay royalties and all other fees and amounts due us or our affiliates under the Franchise Agreement or relating to your Store. We currently require payment of all such amounts by pre-authorized electronic bank transfer, and you must sign all documents necessary to allow us to initiate debit entries and/or credit correction entries to your designated bank account for that purpose.
- 3. "Gross Revenue" means the aggregate of all revenue, in whatever form (including the value of all barter, exchange, trade or other credit), attributed to the operation of your Store (whether or not in compliance with this Agreement) based on the gross (undiscounted) price of all products and services sold at or through your Store and all other revenue sources, but excluding all federal, state or municipal sales, use, or service taxes collected from customers and paid to the appropriate taxing authority. Gift certificate, gift card and similar program payments will be included in Gross Revenue consistent with the policies periodically stated in the Operations Manual. Gross Revenue also includes all insurance proceeds you receive for loss of business due to a casualty or a similar event.
- 4. You must spend at least the minimum required amount on approved local marketing and promotion of your Store ("Local Store Marketing"). You must provide us with a quarterly and annual statement of your Local Store Marketing expenditures. If you do not spend the minimum required amount on Local Store Marketing, you must deposit with us the difference between what you did spend and the minimum required amount for deposit in the Brand Fund.

5. You will be charged lost revenue damages if we terminate the Franchise Agreement as a result of your breach, at an amount equal to the net present value of the Royalty and Brand Fund Contributions that would have become due had the Franchise Agreement not been terminated, from the date of termination to the earlier of: (a) 5 years following the date of termination, or (b) the scheduled expiration of the term of the Franchise Agreement (the "Measurement Period"). Lost Revenue Damages will be calculated as follows: (1) the number of calendar months in the Measurement Period, multiplied by (2) the aggregate of the Royalty and Brand Fund Contributions multiplied by (3) the average monthly Gross Revenue of your Store during the 12 full calendar months immediately preceding the termination date; provided, that if as of the termination date, your Store has not been operating for at least 12 months, the average monthly Gross Revenue of all Stores operating under the same Marks during the entirety of our fiscal year immediately preceding the termination date. You will pay us Lost Revenue Damages within 15 days after the Franchise Agreement is terminated, or on any later date that we determine.

ITEM 7
ESTIMATED INITIAL INVESTMENT

|                                                          | Amount                   |                           | Method of                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | To Whom                                       |
|----------------------------------------------------------|--------------------------|---------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------|
| Type of Expenditure                                      | Store Kiosk              |                           | Payment                                                                                                                             | When Due                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Payment is Made                               |
| Initial Franchise Fee (Note 1)                           | \$35,000                 | \$35,000                  | \$10,000 paid upon signing the Deposit Agreement and balance due upon signing the Eranchise Agreement Agreement Agreement Agreement |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Us                                            |
| Travel and living expenses while training (Note 2)       | \$2,000 -<br>\$3,000     | \$2,000 -<br>\$3,000      | Lump sum, as incurred                                                                                                               | As incurred during training                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Suppliers                                     |
| Real estate lease (Note 3)                               | Note 3                   | Note 3                    | Note 3                                                                                                                              | Note 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Note 3                                        |
| Improvements and Equipment (Note 4)                      | \$230,000 -<br>\$387,300 | \$107,360 to<br>\$184,660 | As agreed with suppliers                                                                                                            | As incurred                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Suppliers                                     |
| Opening Product and Soft Goods<br>Inventory (Note 5)     | \$5,000 -<br>\$10,000    | \$5,000 -<br>\$10,000     | As agreed with suppliers                                                                                                            | As incurred                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Suppliers                                     |
| Grand opening promotion, if opening a new store (Note 6) | \$10,000                 | \$10,000                  | Lump sum                                                                                                                            | As incurred                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Suppliers                                     |
| Local Store Marketing (Note 7)                           | \$2,000                  | \$2,000                   | As agreed with suppliers                                                                                                            | \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) \( \) |                                               |
| Deposits and other prepaid expenses (Note 8)             | \$4,000 -<br>\$5,000     | \$4,000 -<br>\$5,000      | Lump sum                                                                                                                            | Before opening                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Landlord and other suppliers                  |
| Professional fees<br>(Note 9)                            | \$9,000 -<br>\$10,000    | \$9,000 -<br>\$10,000     | Lump sum or as arranged by providers                                                                                                | As incurred                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Attorneys, accountants, and other consultants |
| Insurance (3 months) (Note 10)                           | \$2,500 -<br>\$3,500     | \$2,500 -<br>\$3,500      | Lump sum or<br>installments, as<br>determined by<br>insurance carriers                                                              | Before or upon<br>signing of<br>Franchise<br>Agreement                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Insurance carriers                            |
| Coffee preparation and serving equipment                 | \$2,500 -<br>\$10,000    | \$2,500 -<br>\$10,000     | Monthly                                                                                                                             | Monthly                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | Suppliers                                     |
| Computer hardware and software (Note 11)                 | \$1,500 -<br>\$8,050     | \$1,500 -<br>\$5,150      | Lump sum                                                                                                                            | As incurred                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Suppliers                                     |

|                                          | Amount                    |                           | Method of             |             | To Whom                       |
|------------------------------------------|---------------------------|---------------------------|-----------------------|-------------|-------------------------------|
| Type of Expenditure                      | Store                     | Kiosk                     | Payment               | When Due    | Payment is Made               |
| Additional funds (3 months)<br>(Note 12) | \$8,000 -<br>\$12,000     | \$8,000 -<br>\$12,000     | Lump sum, as incurred | As incurred | Employees and other suppliers |
| Totals                                   | \$311,500 to<br>\$495,850 | \$188,860 to<br>\$290,310 |                       |             |                               |

#### General Comments:

This table estimates the initial investment required to develop a single Store or a Kiosk. All payments you make to us or an affiliate are non-refundable unless otherwise stated. Payments you make to parties other than us or an affiliate may be refundable at the option of the other party. Neither we nor an affiliate offers any financing, directly or indirectly, to you.

#### NOTES:

- 1. The Initial Franchise Fee is \$25,000 if you are an existing franchisee or if you participate in our veterans' discount program. In addition, qualifying first responders receive a discount of 20% of the Initial Franchise Fee.
- 2. You are responsible for any incidental expenses that you and your personnel incur while attending our initial training program, including car rental, gas, airline tickets, meals, hotel room, entertainment, salaries, and benefits.
- 3. If you do not currently own adequate space, you must lease the space for your Store. Typical locations are shopping malls and strip shopping centers. The average Store size ranges from 600 to 900 square feet. The average size of a Kiosk is typically 100 to 300 square feet but may vary widely. Typical locations are located in malls and lifestyle centers. You will make rental payments to the landlord and the amount of your monthly rental payments will vary greatly from site to site and are affected by a number of factors, including location, size, visibility, accessibility, and competitive market conditions, but will generally range between \$65 and \$270 per square foot. In addition to rental payments, your lease may require other payments to the landlord, such as payments for shopping center or building operating expenses, common area maintenance expenses, food court expenses, merchants' association assessments, and other miscellaneous assessments. Your lease may also require you to spend a certain amount on advertising and promotion for your Store. Because these payments vary widely from lease to lease, we cannot estimate the amount you may be required to pay for these or other similar items.
- 4. These estimates include construction costs (labor and material) for typical tenant improvements and remodeling to prepare a site for operation of a Store or Kiosk as well as estimated costs for necessary trade fixtures, such as display cases, signage, counters and work tables, and equipment, such as ovens, refrigerators, beverage dispensers, coffee preparation and dispensing equipment, small wares and cash registers. At select Stores, we may permit the build-out of drive-through services, and the estimates in this Item 7 include the costs to build a drive-through at Stores. The estimates also include construction management costs, general conditions, builder's risk/liability insurance and financing costs. If you develop a new store, we will provide you with prototypical plans and specifications at no additional cost to you, but you must also employ and pay an architect or engineer to prepare a site plan and other construction documents to adapt these plans and specifications to city, state and local building codes and to the specific site chosen for your Store. We may require you to use a commercial contractor that we approve and our designated architect, and, as such, this amount also includes our approved vendor's fee in assisting you with the site design and architectural plans of the premises of the Store (the "Premises"), which we estimate to be between \$6,000 and \$14,000. These estimates do not include lease costs. Your actual construction costs

will depend on numerous factors, such as the condition of the Premises, duration of the building process (delays), union labor requirements, contractors' fees, signage, availability of materials and equipment, interest rates, and the insurance coverage you choose. A modular, prefabricated Kiosk store is a free-standing Store that is operated within a mall, shopping center or other site, but is not a permanent, in-line structure. It is, however, usually operated under a long-term lease. The modular unit is assembled using prefabricated components built off-site, often at a savings over traditional construction methods. It is typically most appropriate for high-traffic center court or corridors within larger metropolitan malls. If you find a site that we approve for installation of this type of unit, you must purchase components from a vendor or vendors we designate.

- 5. This estimate includes supplies, opening inventory, accounting forms and systems, soft goods, such as napkins, cups, and other paper goods, utensils, packaging materials and other required items. The costs will vary depending upon your inventory levels and storage space.
- 6. You must spend a minimum of \$10,000 on a grand opening advertising and promotional program that we mandate or approve.
- 7. The Local Store Marketing obligations begin at month 3 of operations. This amount is the minimum you may spend during month 3 of operations.
- 8. You may be required to pay a security deposit under your real estate lease and other deposits for utilities and insurance premiums. Lease security deposits are typically due upon signing and can potentially be refundable if you do not default on your lease. Your lease may also require you to pay the last month's rent in advance. Deposits for utility services are typically required at the time the service is applied for and may or may not be refundable. You must confirm all of the specific deposits required. The amount for licenses and permits can vary significantly, and you should verify specific amounts with local authorities.
- 9. You should have an attorney to review the real estate lease or sublease, the franchise documents, or to assist in forming a corporation, partnership, or limited liability company. You may also retain an accountant for advice in establishing and operating your franchise business and filing necessary tax forms and returns. This amount also includes our approved vendor's fees to assist you in securing a lease for the Premises of the Store, which we estimate to be between \$3,000 and \$6,000.
- 10. You must obtain and maintain certain types and amounts of insurance. Insurance costs depend on policy limits, types of policies, nature and value of physical assets, Gross Revenue, number of employees, wages paid, square footage, location, business contents, and other factors bearing on risk exposure. Insurance providers may require an annual payment, semi-annual installments or monthly installments. This estimate contemplates monthly installments for 3 months. These estimates also include workers' compensation insurance, which are based on wages, which will vary from state to state. You should review the rates in the state in which you are opening the Store for an estimate of the premium you will be required to pay.
- 11. The computer hardware and software costs for a Store includes initial computer hardware and installation costs, three months of POS software, gift card and internet costs for one (for the low range) to two terminals (for the high range). The range for computer hardware and software costs for a Kiosk covers only the costs associated with one terminal.
- 12. This range represents the additional funds we estimate you will require to operate the Store for the first 3 months. These figures include estimated payroll costs. However, they do not include the salary for the store manager, on the assumption that you will manage the Store. We have based the estimates provided in the table above upon our and our franchisees' experience establishing and operating Stores.

You should review these figures carefully with a business adviser before making any decision to enter into a Franchise Agreement.

## ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

#### **Development of Your Store**

In developing and operating your Store, you must: (i) use only the fixtures, furnishings, equipment and signs that we require and have approved for Stores as meeting our System Standards; (ii) place or display at the Premises (interior and exterior) only the signs, emblems, lettering, logos and display materials that we approve in writing; (iii) use the computer equipment and operating software, including any point-of-sale or electronic cash register ("Computer System") that we specify from time to time; and (iv) purchase the products and services only from those vendors, distributors, suppliers and producers that we approve or specify.

#### Approved Suppliers; Standards and Specifications

As described above in this Item 8, we have approved designated suppliers as the only permitted providers of certain items, including certain products, services (such as credit card processing), signs, furnishings, the Approved Products, the Computer System and fixtures and equipment. We develop the specifications and standards for these items. We may issue to you or to our approved suppliers (except as we deem necessary for purposes of production) the specifications of these items. If we do issue these specifications to you or modify the specifications, it will be through the Operations Manual. We will provide to you a list of approved suppliers in writing, but do not provide you a list of criteria in connection with how we approve suppliers.

You may not deviate from our approved products and services without our prior written approval. If you request our approval of a new supplier and we choose to evaluate your proposed supplier, we may require you to reimburse us the reasonable costs we incur in making this evaluation, which will be a flat rate based on our actual out-of-pocket expenses to evaluate the proposed supplier. We will advise you of our decision on your request within a reasonable time (typically within 60 days) after our receipt of the information we request from you or from the proposed supplier regarding the supplier's qualifications. We may impose limits on the number of approved suppliers. We may elect to withhold approval of a supplier or revoke approval of a supplier, if, at any time, the supplier fails to meet our criteria.

You may use in your Local Store Marketing, including any social media activities, only those materials that we provide or make available to your or for which we have given our prior written approval. You must submit your proposed materials for our review at least 15 days before your proposed use. We may provide you with copies of advertising, marketing and promotional formats and materials for use in your Store, which we have prepared using Brand Fund Contributions we have collected from Stores. While we expect to provide some of these materials without charge, we may charge you for these materials or require that you pay certain shipping and related expenses associated with delivering them to you.

You must use our designated real estate broker to evaluate potential sites, unless we otherwise agree in writing. In addition, we may require you to use our designated architect to assist you with the site design and architectural plans for the Premises.

You must contract with our designated vendor for the technology platform that will collect monthly financial statements for the Store. You must submit monthly, quarterly and annual financial statements to us through the technology platform using a standardized chart of accounts. If you fail to provide the required financial statements, we may require you to use one of our preferred bookkeepers at your expense.

Neither we nor any of our affiliates is currently an approved or designated supplier, but we may be so in the future. None of our officers currently owns an interest in any approved or designated supplier.

#### Gift Card and Loyalty Programs

You must, at your expense, participate in, and honor all provisions of any gift card or loyalty programs we establish, and use any designated providers we specify for such programs.

#### Insurance

At your expense, you must purchase and maintain for each Store you operate: (1) comprehensive general liability insurance with minimum limits of \$1,000,000 per occurrence and \$2,000,000 annual aggregate; (2) workers' compensation, employer's liability and other insurance to meet the greater of all applicable statutory requirements or the then-current minimum levels of coverage we require; (3) commercial property insurance including, at a minimum, fire, vandalism, theft, burglary, and extended coverage with limits of at least 100% replacement value of the Store premises, fixtures, equipment and inventory; (4) business interruption/time element coverage in the amounts we require either as a component of or an endorsement to a commercial property insurance policy; (5) automobile liability insurance, including personal injury, wrongful death and property damage, with limits of at least \$1,000,000 per occurrence; (6) data privacy insurance; (7) employment practices liability insurance; and (8) an umbrella liability coverage of \$2,000,000 each occurrence and \$4,000,000 in the aggregate. We periodically may, with prior written notice to you, increase the minimum liability protection requirements, and require different or additional kinds of insurance to reflect inflation or changes in standards of liability. Your general liability policy must name us and our affiliates and any other person that we designate as an additional insured and must meet any other requirements that we designate. If you own more than one Store, (1) we may require you to obtain an umbrella liability policy; and (2) we may allow you to obtain single policy for such Stores provided that we periodically will determine the levels of insurance coverage that you must obtain and other requirements that you must satisfy for each Store covered by such policy.

#### Required Franchisee Purchases; Rebates

We estimate that the cost of required purchases of products, supplies, fixtures, furnishings, equipment, signs and leases from approved suppliers or otherwise will represent, for Stores, 80% to 90% of your overall purchases of those items in establishing your Store, and 65% to 75% of your overall purchases of those items in operating your Store.

During our fiscal year ended December 31, 2024, we received revenues of \$577,614 as a result of franchisee purchases of products and services. This amount represents 23.2% of our total revenue from all sources during our last fiscal year, which was \$2,487,808. During our fiscal year ended December 31, 2024, our affiliates did not derive any revenue from our franchisees' required purchases of products or services.

In the future, we may continue to receive and our affiliates may receive rebates or other payments from distributors, suppliers and other service providers based (directly or indirectly) on sales to franchisees. These payments have ranged or may range from less than 1% to 15% or more of the amount of those purchases by franchisees.

We or our affiliates may negotiate purchase arrangements, including prices and terms for the benefit of franchisees, with designated and approved suppliers for Stores. We do not participate in any purchasing or distribution cooperatives. We do not provide material benefits to franchisees based on their purchase of particular products or services.

#### ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

|    | Obligation                                                            | Continuin Agreement                                                                                | Disclosure Document          |
|----|-----------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|------------------------------|
|    | Obligation Site selection and                                         | Section in Agreement Sections 4.1 and 4.2 of Franchise Agreement;                                  | Item                         |
| a. | acquisition/lease                                                     | Section 4 of Area Development Agreement                                                            | Items 6, 7, 8, 11 and 12     |
| b. | Pre-opening purchases/leases                                          | Sections 4.2, 4.3, 4.4, 4.6 and 7.2 of Franchise Agreement                                         | Items 5, 6, 7, and 8         |
| c. | Site development and other pre-opening requirements                   | Sections 4.3, 4.4, 4.5, and 7.2 of Franchise Agreement;<br>Section 4 of Area Development Agreement | Items 6, 7 and 11            |
| d. | Initial and ongoing training                                          | Section 5 of Franchise Agreement                                                                   | Items 6, 7 and 11            |
| e. | Opening                                                               | Section 4.5; Section 4 of Area Development Agreement                                               | Items 5, 6, 7 and 11         |
| f. | Fees                                                                  | Section 6; Section 2 of Area Development Agreement                                                 | Items 5, 6, 7 and 11         |
| g. | Compliance with System<br>Standards and<br>policies/Operations Manual | Sections 5.1, 5.3 and 7.2 of Franchise Agreement                                                   | Items 6, 7, 8, 11, 15 and 16 |
| h. | Marks and proprietary information                                     | Section 10; Section 9 of Area Development Agreement                                                | Items 8, 13, 14 and 17       |
| i. | Restrictions on products/services offered                             | Sections 2.3, 4.4 and 7.2 of Franchise Agreement                                                   | Items 1, 8, 14 and 16        |
| j. | Warranty and customer service requirements                            | Sections 7.2 and 7.6 of Franchise Agreement                                                        | Item 11                      |
| k. | Territorial development and sales quotas                              | Section 5 of Area Development Agreement                                                            | Item 12                      |
| 1. | Ongoing product/service purchases                                     | Sections 2.3, 4.4 and 7.2 of Franchise Agreement                                                   | Items 8 and 11               |
| m. | Maintenance, appearance, and remodeling requirements                  | Section 4.3, 4.4 and 7.2 of Franchise Agreement                                                    | Items 11, 13 and 17          |
| n. | Insurance                                                             | Section 7.8 of Franchise Agreement                                                                 | Items 6, 7 and 11            |
| 0. | Advertising                                                           | Sections 4.6 and 9 of Franchise Agreement                                                          | Items 5, 6, 7, 11 and 13     |
| p. | Indemnification                                                       | Section 15.4 of Franchise Agreement                                                                | Item 6                       |
| q. | Owner's participation/<br>management/staffing                         | Sections 7.2, 7.3 and 7.11 of Franchise Agreement                                                  | Items 11 and 15              |
| r. | Records and reports                                                   | Sections 7.5 and 8 of Franchise Agreement                                                          | Not applicable               |
| s. | Inspections and audits                                                | Sections 7.5 and 8 of Franchise Agreement                                                          | Item 6                       |
| t. | Transfer                                                              | Section 12; Section 8 of Area Development Agreement                                                | Item 17                      |
| u. | Renewal                                                               | Section 3.1 of Franchise Agreement                                                                 | Item 17                      |
| v. | Post-termination obligations                                          | Section 14; Section 7 of Area Development Agreement                                                | Item 17                      |
| w. | Non-competition covenants                                             | Section 11                                                                                         | Item 17                      |
| х. | Dispute resolution                                                    | Section 17; Section 10 of Area Development Agreement                                               | Item 17                      |

#### ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee your note, lease or any other obligation.

#### ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

#### **Assistance Before Opening:**

Before you open your Store, we or an affiliate will:

- 1. Approve or disapprove a site that you propose for your Store. (Franchise Agreement, Section 4.1)
- 2. Approve or disapprove the Lease (as defined below) that you propose for the Premises. (Franchise Agreement, Section 4.2)
- 3. If you are developing a new Store, provide you with prototypical plans and specifications on which to model the plans to build out your Store. (Franchise Agreement, Section 4.3(a))
- 4. Provide you, through the Operations Manual and other materials, the standards and specifications for the fixtures, furnishings, equipment (including computer hardware and software) and signs that you must use. (Franchise Agreement, Sections 4.4 and 5.1)
- 5. Provide an initial training program for you, or if you are an entity, your Entity Owner (as defined in Item 15) and for your initial store manager. (Franchise Agreement, Section 5.2(a))

#### **Assistance During Operation:**

During the operation of your Store, we or an affiliate will:

- 1. Provide electronic access to the Operations Manual. (Franchise Agreement, Section 5.1)
- 2. Provide training for any existing or replacement store managers, as explained below. (Franchise Agreement, Section 5.2(b))
  - 3. Provide additional or refresher training programs. (Franchise Agreement, Section 5.2(c))
  - 4. Administer the Brand Fund. (Franchise Agreement, Section 9.1)
- 5. Provide advertising and marketing services to you as explained below. (Franchise Agreement, Section 9.3)
- 6. We may periodically advise you regarding various aspects of the operation of your Store. (Franchise Agreement, Section 5.3)
- 7. We may periodically set a maximum or minimum price that you may charge for products and services offered by your Store. (Franchise Agreement, Section 7.9)

#### **Site Selection**

We will identify the Premises on Schedule 3 of the Franchise Agreement at the time we sign the Franchise Agreement. If at the time of signing the Franchise Agreement we have not approved and you have not secured the Premises for your Store, and you subsequently obtain our approval of and secure the Premises for your Store, we and you will sign Alternative Schedule 3 identifying the Premises for your Store. If we do not approve your proposed site within 30 days of receiving the required information, it is deemed disapproved. Generally, we do not own the Premises of the Store or lease it to you. If you enter into an Area Development Agreement, we and you will have agreed to a Development Schedule which identify the number of Stores you will develop, and the time frame and the area in which the Stores will be developed.

You must use our designated real estate broker when you are evaluating potential sites, unless we agree otherwise in writing. You must locate and obtain our written acceptance of the Premises before you sign a lease or sublease for or begin construction of the Premises. Our approval of the Premises is for our own internal purposes and will, to a large extent, be based on information you provide us with respect to the demographic and commercial characteristics of the proposed Premises. We consider several factors when evaluating a proposed site, including the demographics of the area surrounding the proposed site, the neighborhood, the size of the site, the parking at the site and traffic patterns around the site. We may, at our discretion, assist you in identifying and securing possession of the Premises, including by identifying one or more sites that we believe to be available and that might meet our criteria. We will endeavor to approve or disapprove the Premises within 30 days of receiving all information we request regarding a proposed site from you.

Once we have confirmed, in writing, our approval of your proposed Premises, it will be your responsibility to secure the right to occupy the Premises under an agreement that allows you to develop and operate your Store (the "Lease"). You may not execute the Lease until we have accepted its terms. Our approval will be conditioned on the landlord's agreement, either in the body of the Lease or in a separate addendum or rider, with certain provisions that we determine to be important for the continuity and goodwill of the Mrs. Fields brand and for our position as franchisor, including:

- (a) notice of, and the right (but not the obligation) to cure, your default of the Lease;
- (b) agreement that, without the landlord's consent, you can assign to us, our affiliate, or another franchisee your interest under the Lease; and
- (c) a requirement that the landlord disclose to us, upon our request, sales and other information that you furnish to it.

You must provide us with a copy of the fully executed Lease within 15 days after its execution. We will endeavor to approve the Lease within 10 days of receiving it from you, but our failure to do so does not mean we have approved it or that you may execute it. The final Lease is always subject to our approval. If you do not open your Store within 12 months following the date of the Franchise Agreement, we may terminate the Franchise Agreement.

You must do all things necessary to construct and develop your Store in accordance with our System Standards, the Lease, and applicable law. We will furnish you with prototypical plans and specifications for a Store, including requirements for exterior and interior materials and finishes, dimensions, design, image, interior layout, decor, fixtures, equipment, signs, furnishings and color scheme. You must comply with these plans and specifications, as adapted, with our prior written approval, to accommodate the particular shape and dimensions of the Premises and to ensure that the plans and specifications comply with applicable ordinances, building codes and permit requirements and with lease

requirements and restrictions. You will submit construction plans and specifications to us for our approval before construction of your Store is commenced, and to submit all revised plans and specifications to us for our approval during the course of construction. In constructing and outfitting your Store, you may use only those materials and items (including furniture, fixtures and equipment) we designate, and you may secure those items from and use only those providers (including an architect and general contractor) that we designate or approve from time to time. We do not deliver or install any of the required furniture, fixtures, or equipment, but we reserve the right to install signage at your expense. On completion of construction, you will provide us with a set of "as built" plans and specifications. You will assume all risk relating to the construction and development of your Store.

#### Time to Open

If you are developing a new Store, and you already have an approved Premises, we estimate that it will take between 90 to 150 days between the date you sign the Franchise Agreement or first pay consideration for your franchise, and the date you open your Store. If you do not have an approved Premises, then this estimate could be increased by up to 6 months. The interval may vary depending upon factors such as the weather, the location and condition of the site, your ability to obtain any necessary financing and building, zoning or other permits and approvals, construction delays, completion of required training and so forth. You will not open your Store for business until we approve it for opening. We will not approve it for opening until all required pre-opening training has been completed to our satisfaction, all amounts then owed to use and our affiliates have been paid, you have provided us with a copy of (and we have accepted) your fully executed Lease, and you have provided us with a certificate of insurance issued by your insurance provider confirming that you have procured and fully paid for the insurance required under the Franchise Agreement. If you sign an Area Development Agreement, you must complete development and open all of your Stores according to the Development Schedule.

#### **Operations Manual**

To facilitate the training and guidance we provide and to assist you in developing and operating your Store, we will provide you with access to a manual or series of manuals, memos, newsletters and other written communications for your use solely in developing and operating your Store (the "Operations Manual"). The Operations Manual contain both mandatory and suggested specifications, standards and operating procedures, including our "System Standards," which are the operating procedures, standards, requirements and specifications, whether contained in the Operations Manual or elsewhere, which we may improve, further develop or modify from time to time and which are mandatory in nature so as to comprise the requirements to be followed by Stores, and the use of the Marks in connection therewith. The Operations Manual may be modified by us from time to time to reflect changes in the image, specifications, standards, procedures, Approved Products, System, and System Standards. You may not at any time copy any part of the Operations Manual, either physically or electronically. If there is a dispute relating to the contents of the Operations Manual, the master copy we maintain at our corporate headquarters will control. The Operations Manual and any passwords that we provide to access it electronically form part of our Confidential Information (as defined in Item 14) and, as such, are subject to the same restrictions as other Confidential Information regarding its use, disclosure and return. As of the date of this disclosure document, the Operations Manual is available online only and consists of two components, a decorating manual and a Mrs. Fields manual, which together contain approximately 501 pages. The table of contents for the Operations Manual is attached as Exhibit E hereto.

#### **Computer System**

We require you to obtain our specified Computer System to track sales and facilitate reporting to us. We require the Computer System be purchased from us or a designated supplier. The hardware

components include 1 to 3 terminals each with a credit card terminal, receipt printer, cash drawer and router and scale (for self-serve concepts only). You must use our designated credit card processor.

Our modification of specifications for the Computer System's components may require you to incur costs to purchase, lease and/or license new or modified computer hardware and/or software, obtain service and support for the Computer System during the term of the Franchise Agreement and pay any related connection fees. We may require you in the future to purchase additional or different components of the Computer System, including computer hardware and software and connection and other related services, from a supplier or suppliers we designate, which may include us or our affiliates. You will incur the costs of obtaining the computer hardware and software comprising the Computer System (or additions or modifications). There are no contractual limitations to the frequency and cost of the obligation to upgrade and maintain the Computer System. You are required to establish and maintain a valid email address and authorize us to communicate with you by this method at such address. We may independently access the information and data you collect and gather using any Computer System or other data collection equipment (such as an electronic cash register) we require for your Store, and there is no contractual limit on our right to do so. For defective equipment, products, software or other items purchased by you, you must deal directly with that manufacturer. We are not responsible for claims or damages incurred by you as a result of you using our Computer System. The estimated cost for the purchase of the POS hardware is approximately \$3,500 - 4,500 and the software company currently charges ongoing fees depending on number of terminals and other auxiliary equipment you may add-on. The software subscription cost is approximately \$1,020 - \$2,300 per year. Neither we nor our affiliate have any contractual obligation to maintain, repair, update or upgrade the Computer System.

If you are buying an existing Store with an unapproved cash register system or that does not otherwise comply with our required Computer System and Brand Standards, we will require you or the seller to purchase and install the approved POS system as well as other approved Computer System components in the Store prior to transfer as a condition of our consent to the transfer. If you are remodeling an existing Store with an older cash register or POS system, we will require you to purchase and install our then-current POS system and approved Computer System components.

If you do not comply with the required Computer System that we specify, you will be assessed a Computer System Compliance Fee in the amount of the greater of \$500 per month, or the then current monthly rate of the software license. In addition, your failure to comply with the required Computer System that we specify will constitute a default under Section 13.1(m) of the Franchise Agreement. To avoid any misunderstanding, any deviation from our required Computer System is subject to our prior written approval.

#### **Training**

We will provide you with access to our initial training program on the operation of Stores (the "Initial Training Program"). Before opening your Store for operation, you, and at least one of your Entity Owners (if you are an Entity), and your initial store manager must successfully complete, to our satisfaction, the Initial Training Program. We currently offer much of the Initial Training Program through our online Learning Management System ("LMS"). We offer the Initial Training Program on a rolling and as-needed basis. Our training materials are updated periodically, but may include worksheets, web-based programs, PowerPoint presentations and other materials we specify. You may not allow your Store to be managed by a person who has not completed, to our satisfaction, the Initial Training Program or who we have not otherwise certified to do so. We will not charge an additional fee for up to 2 people to attend or participate in the Initial Training Program.

If you replace your initial store manager, the replacement manager will also be required to complete, to our satisfaction, all phases of the Initial Training Program. We may charge a fee for the training for subsequent managers, which you must pay at least 10 days prior to beginning of training.

With the exception of the Initial Training Program, we may require that you provide initial and/or ongoing training for your other employees using our online library of training modules on our LMS. The online training program may include tests that each trainee must pass before he or she will be deemed to have successfully completed the training program.

We may require you and/or your previously trained and experienced representatives to attend periodic refresher training programs and training programs or courses regarding specific products or processes. We may charge fees for these additional or refresher training courses. Our current fee for additional or refresher training is \$500 per day, per person, plus additional travel and living expenses.

You or at least one of your Entity Owners (if you are an Entity) and, when we request, the Store manager (and/or an approved trainer if you are a multi-unit franchisee) must attend all national conventions and regional meetings that we designate as mandatory. National conventions and regional meetings will take place at the time and locations we designate, and we may charge you a reasonable fee for each of your attendees. Currently, we do not charge a fee, but you will be responsible for any travel and lodging expenses for each of your attendees. We may also charge you a reasonable fee if you or any of your required representatives fails to attend any national convention or regional meeting we designate as mandatory.

You must arrange for and pay all of your and your other representatives' other costs of attending or participating in the Initial Training Program and all other training programs we offer. We will determine the format, content, method of delivery and location of each training program.

The Initial Training Program will include classroom, in-store/on-the-job, computer-based training through the LMS and self-study and is described below:

#### TRAINING PROGRAM

| Subject                                                                              | Hours of Classroom<br>Training | Hours of On the<br>Job Training | Location                                                |
|--------------------------------------------------------------------------------------|--------------------------------|---------------------------------|---------------------------------------------------------|
| Orientation: welcome, overview, company history, vision and values, mission          | 1                              | 0                               | LMS, Salt Lake City, Utah or other designated locations |
| Customer Service: sampling, upselling, complaint handling                            | 1                              | 4                               | LMS, Salt Lake City, Utah or other designated locations |
| Equipment: use and care                                                              | 5                              | 4                               | LMS, Salt Lake City, Utah or other designated locations |
| Product: cookies, brownies, beverages and decorating                                 | 4                              | 4                               | LMS, Salt Lake City, Utah or other designated locations |
| Marketing: local, national, social media                                             | 2                              | 1                               | LMS, Salt Lake City, Utah or other designated locations |
| People: hiring, training, coaching                                                   | 2                              | 1                               | LMS, Salt Lake City, Utah or other designated locations |
| Regulatory compliance: food safety, OSHA compliance                                  | 1                              | 1                               | LMS, Salt Lake City, Utah or other designated locations |
| Financial: POS usage, sales reporting, budgeting                                     | 4                              | 1                               | LMS, Salt Lake City, Utah or other designated locations |
| Administrative: ordering, receiving, inventory, NSO process, scheduling, maintenance | 1                              | 1                               | LMS, Salt Lake City, Utah or other designated locations |

| Subject                                                   | Hours of Classroom<br>Training | Hours of On the<br>Job Training | Location                                                |
|-----------------------------------------------------------|--------------------------------|---------------------------------|---------------------------------------------------------|
| Meet with personnel from the Franchisee Support<br>Center | 1                              | 0                               | LMS, Salt Lake City, Utah or other designated locations |
| Final knowledge assessments                               | 2                              | 0                               | LMS, Salt Lake City, Utah or other designated locations |
| TOTALS                                                    | 24                             | 17                              |                                                         |

The training program is overseen by our Director of Training, Talent & Culture, who currently is Chris Sabido. Chris has been our Director of Training, Talent & Culture since December 2023 and has over 30 years of experience in the food and service industry.

Training will also be administered by our Training & Operations Support Manager, who currently is Darren Schmitt. Darren has been our Training & Operations Support Manager since February 2024 and has over 4 years of experience in the food and service industry.

In addition, training will be administered by our Operations and Training Manager, who currently is Kim Browning. In total, Kim has worked for us and our affiliates for approximately 30 years and most recently resumed her role with us in April 2017. Kim has over 30 years of experience in the food and service industry.

#### **Marketing and Advertising**

#### **Brand Fund and Brand Fund Contributions**

We operate a National Brand Fund (the "Brand Fund") to generally advertise and promote the Stores and the Marks in the System. You will pay us a monthly national Brand Fund contribution of 3% of your Store's Gross Revenues (the "Brand Fund Contribution"). We will deposit the Brand Fund Contribution in the Brand Fund that we manage through a separate account. Any Store owned by us or an affiliate will contribute to the Brand Fund on the same basis as franchised Stores. Some franchisees may not be required under their franchise agreement to contribute at the same rate.

We will, without any fiduciary obligation to you, direct or cause our designee to direct the Brand Fund. We will determine the creative concepts, materials and endorsements used and the geographic, market and media placement and allocation. The expenses that may be funded by the Brand Fund Contributions we collect include preparing and producing video, audio, and written materials and electronic media; developing, implementing, and maintaining a website for the System and Stores and related strategies; administering regional and multi-regional marketing and advertising programs, including purchasing trade journal, direct mail, and other media advertising and using advertising, promotion, and marketing agencies and other advisors to provide assistance; administering online advertising and marketing campaigns (including search engine, social media, email, and display ad campaigns); developing and maintaining application software designed to run on computers and similar devices, including tablets, smartphones and other mobile devices, as well as any evolutions or "next generations" of any such devices; implementing a loyalty program or other marketing programs designed to encourage the patronage of Stores; and reimbursing us or our designees for the reasonable salaries and benefits of personnel who manage and administer the Brand Fund, including administrative costs, travel and other expenses reasonably related to administering or directing the Brand Fund and its programs. We may use Brand Fund Contributions to prepare, furnish or offer for sale to you advertising, marketing and promotional formats and materials for your use in your own local marketing efforts.

The Brand Fund Contributions we collect will be accounted for separately from our other funds, and we may segregate them from our other operating funds. We will prepare an annual, unaudited statement

of collections and expenses of the Brand Fund, and you may obtain a copy of such statement by making a written request to us. The Brand Fund is intended for general promotion. We do not guaranty any particular result or that our expenditure of Brand Fund Contributions or the conduct of the Brand Fund will directly, indirectly, or proportionally benefit your Store or have any relationship to the amount of Brand Fund Contributions you pay. We are not obligated to spend any amount on advertising in any proximity of your Premises or territory. You agree to have your Store participate, in accordance with the rules and requirements we establish for the Brand Fund.

During our fiscal year ended December 31, 2024, Brand Fund Contributions were used as follows: (i) 61% on production, printing, and merchandising; and (ii) 39% on administrative expenses (including salaries of advertising/marketing executives and employees of ours or our affiliates).

The Brand Fund has not been used, and we do not intend to use it, for advertising principally directed at the sale of franchises. Currently, there is no advertising council composed of franchisees, though there may be one in the future.

#### **Local Marketing**

In addition to paying Brand Fund Contributions and participating in the Brand Fund, you must spend at least \$2,000 per month for marketing purposes in your Territory, beginning 3 months after opening your Store and continuing for 9 months after opening. You must use the Local Store Marketing on approved local marketing and promotion of your Store. You must provide us with a quarterly and annual statement of your Local Store Marketing expenditures. If you do not spend the minimum required amount on Local Store Marketing, you must deposit with us the difference between what you did spend and the minimum required amount for deposit in the Brand Fund. We will deposit such amounts into the Brand Fund to be spent on local marketing and promotional activities in, near or attempting to target the geographic area of your Store, including marketing and promotional activities on the Internet. You may use in your Local Store Marketing, including any social media activities, only those materials that we provide or make available to you or for which we have given our prior written approval. You must submit your proposed materials for our review at least 15 days prior to your proposed use. After the first year of operations, you must spend each month the greater of (a) 2% of Gross Revenue, or (b) \$1,000. Your Local Store Marketing must include the creation of an annual marketing plan, which we must approve.

We may also require you to participate in a social media technology platform ("Social Media Technology Platform"). A portion of your Local Store Marketing will be designated for the Social Media Technology Platform. We intend to use the Brand Fund to pay the remaining cost of the Social Media Technology Platform. The Social Media Platform will allow you to upload templates of social media content to the Social Media Technology Platform to be posted on social media sites such as Instagram, Facebook and other sites.

#### **Grand Opening Advertising**

If you are developing a new Store or Kiosk, you must spend a minimum of \$10,000 on a grand opening advertising and promotional program that we require or approve. You must submit a grand opening plan containing details about your planned grand opening promotion, and obtain our approval of the plan, at least 60 days before your opening date.

#### **Advertising Materials**

We may develop and market special mandatory promotional items for Stores and require you to maintain a representative inventory of these promotional items to meet public demand. The source of the advertising material is generally in-house advertising resources. We will make these items available to you

at our direct cost plus a reasonable mark-up and any shipping, handling and storage charges. If you do not place minimum orders of products and other items necessary for a mandatory promotion or product roll-out by a certain date, we may send, or direct suppliers to send, an automatic shipment of a specified minimum quantity of such products and items to you, and you must accept and pay for them upon receipt. We may also conduct coupon or discount promotions. In such case, we may require you to accept coupons or discounts that are issued or communicated by us or our affiliates and presented at your Store by your customers. You may receive certain compensation for these coupons when you tender them to us based on our System Standards.

#### **Advertising Cooperatives**

We may designate any geographical area for purposes of establishing a regional advertising and promotional cooperative ("Cooperative"), and to determine whether a Cooperative is applicable to your Store. If a Cooperative exists or is established for the area in which your Store is located, you must join and participate in the activities of the Cooperative. Each Cooperative will be organized and governed in a form and manner, and will commence and cease operations on the dates we determine. If a Cooperative is formed in your market, we will make available to you any documents governing the operation of your Cooperative, including any rules relating to membership. Each Cooperative will be organized for the exclusive purpose of administering regional advertising programs and standardizing advertising materials for use by the members in local advertising. No promotional or advertising plans or materials may be used by a Cooperative or furnished to its members without our prior approval. We will determine the amount and timing of the required contributions to the Cooperative and all activities and other decisions of the Cooperative with input from the Stores that are members of the Cooperative. All required contributions of the Cooperative will be payable to the Cooperative. If a Cooperative is established, Company-owned Stores may be required to contribute to the Cooperative at the same rate as franchisee-owned Stores. All such contributions and decisions will be subject to our approval, which we will not unreasonably withhold. Contributions will be payable to the Cooperative or us or our designee if we are administering the accounting and vendor payments of the Cooperative.

#### **Internet Use**

You may not advertise your Store or the Approved Products over the internet (or any other form of electronic commerce) or establish a related site on the World Wide Web without our prior written consent. Your general conduct on the internet (or any other form of electronic commerce) and specifically your use of the Marks is subject to the provisions of the Franchise Agreement and our Brand Standards. You will follow our policies and procedures as they may be communicated to you periodically in the Operations Manual or otherwise regarding the use of social media and similar methods of communication. You must obtain our prior written consent prior to your use of the Marks or any content using the Marks on social media, in any form available now or in the future, including but not limited to Facebook, Instagram and Twitter, and your failure to obtain such consent will constitute a breach of the Franchise Agreement. We also have the right to monitor, edit and/or delete any social media content you may post that does not meet our Brand Standards. We require you to have access to the internet from the Premises and to submit all sales activity through the Computer System through a pre-approved router and internet connection. We also require you establish and maintain a valid email address and authorize us to communicate with you by this method at such address.

#### **Pricing**

We may periodically set a maximum or minimum price that you may charge for products and services offered by your Store. If we impose such a maximum or minimum price for any product or service, you may charge any price for the product or service up to and including our designated maximum price or down to and including our designated minimum price. The designated maximum and minimum prices for

the same product or service may, at our option, be the same. For any product or service for which we do not impose a maximum or minimum price, we may require you to comply with an advertising policy adopted by us which will prohibit you from advertising any price for a product or service that is different than our suggested retail price. Although you must comply with any advertising policy we adopt, you will not be prohibited from selling any product or service at a price above or below the suggested retail price unless we impose a maximum price or minimum price for such product or service where permitted by law.

#### ITEM 12 TERRITORY

#### **Franchise Agreement**

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. Except for those rights you may receive under an Area Development Agreement, you receive no minimum territory and you must obtain our approval of the location of your Store. You are not granted an exclusive area or protected territory around the Premises within which we or our affiliates agree not to issue franchises or operate competing businesses. Our affiliate, TCBY Systems, LLC, may operate or franchise others to operate outlets located near Stores. TCBY outlets offer frozen and non-frozen dessert and treat items, such as cakes and pies, sorbet, smoothies and fresh yogurt. The principal trademarks used by TCBY Systems, LLC are TCBY® and THE COUNTRY'S BEST YOGURT®. One or more of these outlets may already be located near your proposed Store, or may be established near the Store in the future. Additionally, we or our affiliates may establish, acquire or co-brand new concepts operating under different trademarks and those outlets may be located near the Store. We have no system or method for resolving conflicts between our franchisees and franchisees of these competing concepts. As stated in Item 1, we share our principal address with TCBY, and we have no plans to maintain physically separate offices or training facilities from TCBY.

We and our affiliates may, anywhere in the world, issue franchises, grant licenses for products or services, and conduct any other business or activities, with or without the Marks and in any type of distribution channel, including the internet, regardless of whether such activities are competitive with your Store. We have not yet established other franchises or company-owned Stores that sell or lease similar products or services under a different trademark, but we may do so in the future. There are no restrictions on these activities and neither we nor our affiliates will be required to provide you with any compensation for, or any right to participate in, any such activities.

You may not operate your Store at any site other than the Premises without our prior written approval. In addition, you may only offer and sell finished Approved Products that have been approved for sale over the counter to retail customers from your Store, and may not sell Approved Products or any materials, supplies, or inventory bearing the Marks at any other location or through any alternative channel of distribution without our prior written consent. "Alternative channels of distribution" include the operation of a food cart or kiosk, sales through the Internet (or any other form of electronic commerce), vending machine sales, and mail order and telephone sales. You may, however, (i) offer and sell Approved Products as part of off-site catering events and company account programs, provided you or a third-party local delivery company deliver (and do not engage a major air/ground shipping carrier to deliver) Approved Products that meet System Standards for freshness the same day they are made and the sales are not part of a mail order program; (ii) offer samples of Approved Products at or directly in front of your Store or other locations near your Store as approved by your landlord; or (iii) upon our prior written approval, offer and sell Approved Products from a table, kiosk or cart at satellite locations that we approve. You may not sell to anyone any materials, supplies, or inventory used to prepare any Approved Products. You may not sell any Approved Products to any person or entity purchasing the Approved Products for resale.

Your Franchise Agreement does not grant you the ability to relocate, any options, rights of first refusal or similar rights to acquire additional franchises.

#### **Area Development Agreement**

If you enter into an Area Development Agreement, you will receive certain protected rights to develop more than one Store within a designated geographic area (the "Designated Area") to be described in Exhibit A attached to the Area Development Agreement. The size of the Designated Area will vary, depending on the number of Stores you intend to open, the population density, and the demographics in the area in which you desire to operate. The Designated Area is described by a written description or map in an attachment to your Area Development Agreement and shows the metropolitan area, zip codes, cities, counties, or other political subdivisions or market areas designated by physical boundaries like streets, highways or physical landforms and is usually determined by density of population and demographics. We will not establish another franchised or company-owned Store or Kiosk in the Designated Area so long as you meet the Development Schedule, satisfy our minimum capital requirements to develop the Store, comply with all other provisions described in the Area Development Agreement and you otherwise comply with the provisions of each related Franchise Agreement. As described above, however, we have certain rights under Area Development Agreement and each Franchise Agreement to sell products and services using the Marks or other marks using similar or dissimilar channels of distribution in the Designated Area. If you do not comply with the Development Schedule and the Area Development Agreement, we may terminate the Area Development Agreement or terminate your protected rights to develop Stores in the Designated Area and grant individual or multiple unit franchises within the Designated Area to third parties.

#### ITEM 13 TRADEMARKS

Under the Franchise Agreement, we grant you the right to operate a Store under the Marks. The following are the principal Marks registered on the Principal Register of the U.S. Patent and Trademark Office ("USPTO"):

| Principal Trademark | U.S. Registration No. | Principal/Supplemental<br>Register of the United States<br>Patent and Trademark Office | Date of Registration                     |
|---------------------|-----------------------|----------------------------------------------------------------------------------------|------------------------------------------|
| Mrs. Tields         | 4,546,222             | Principal                                                                              | June 10, 2014                            |
| Mrs. Tields         |                       | Principal                                                                              | April 3, 2012                            |
| Mrs. Fields         | 3,806,710             | Principal                                                                              | June 22, 2010<br>Renewed: March 24, 2021 |

| Principal Trademark | U.S. Registration No. | Principal/Supplemental Register of the United States Patent and Trademark Office | Date of Registration                            |
|---------------------|-----------------------|----------------------------------------------------------------------------------|-------------------------------------------------|
| Mrs.Fields          | 3,384,301             | Principal                                                                        | February 19, 2008<br>Renewed: November 1        |
| Mrs Fields          | 3,384,300             | Principal                                                                        | February 19, 2008<br>Renewed: November 16, 2017 |
| Mrs. Tields         | 1,983,184             | Principal                                                                        | July 2, 1996<br>Renewed: June 24, 2016          |
| Mrs. Tields         | 1,299,149             | Principal                                                                        | October 2, 1984<br>Renewed: May 21, 2014        |

Mrs. Fields Gifting and Licensing, LLC ("MFGL") owns the Mrs. Fields trademarks. MFBI has filed all required affidavits with the USPTO. There are no currently effective material determinations of the USPTO, Trademark Trial and Appeal Board, the Trademark Administrator of any state, or any court, nor are there any pending infringements, opposition or cancellation proceedings or any pending material litigation, involving the MRS. FIELDS Mark described above.

Your right to use the Marks is derived solely from the Franchise Agreement and is limited to your conduct of business in compliance with the Franchise Agreement and all applicable standards, specifications, operating procedures and rules that we require. We derive our right to use and sublicense the Marks pursuant to a Trademark License Agreement dated October 4, 2023 that we have entered into with MFGL. The term of that license agreement began on October 4, 2023, and will continue to run for 99 years unless otherwise terminated. MFGL may terminate the Trademark License Agreement if we become insolvent, challenge MFGL's ownership of the Marks, or fail to cure any material breach within 30 days. If the Trademark License Agreement terminates, any then-existing franchisees will be required to cease using the Marks immediately. Other than this license agreement with MFGL, there are no agreements currently in effect which significantly limit our rights to use or franchise the use of the Marks.

You must use the applicable Marks as the sole identification of your Store, and you must identify yourself as the independent owner in the manner we require. You may not use any Mark as part of any corporate or trade name or with any prefix, suffix or other modifying words, terms, designs or symbols (other than logos franchised to you under the Franchise Agreement), or in any modified form, including on any sites on the Internet or World Wide Web, as an Internet domain name, or as part of an electronic mail

address without prior written approval from us. You must follow all other policies and procedures relating to the Marks, as contained in the Operations Manual. You must display all applicable Marks in the manner we require, and you must use the registration symbol "®" in using any of the registered Marks. You must refrain from any business or marketing practice which may be injurious to our business and the good will associated with the Marks or Stores. We may require you to modify or discontinue use of any Mark or use one or more additional or substitute trade or service marks if we determine that it becomes advisable at any time. You must comply with our directions in this regard. We are not required to reimburse you for your expenses in modifying or discontinuing the use of a Mark and substituting a different Mark or for any loss of goodwill associated with any modified or discontinued Mark or for any expenditures made by you to promote a modified or substitute Mark.

You must immediately notify us of any apparent infringement of or challenge to your use of any Mark or claim by any person of any rights in any Mark, and you must not communicate with any person other than us or our counsel about the infringement, challenge or claim. We and our affiliates may take the action we deem appropriate and control exclusively any litigation, USPTO proceeding or any other administrative or court proceeding concerning any Mark. You must sign any instruments and documents, render assistance and do those things as, in the opinion of our legal counsel, may be necessary or advisable to protect and maintain our interests in any litigation or USPTO or other proceeding or otherwise to protect and maintain our interests in the Marks.

We will take the action we think appropriate. We will indemnify you against and reimburse you for all damages for which you are held liable in any proceeding arising out of your authorized use of any Mark and for all costs you reasonably incur in defending any claim brought against you or any proceeding in which you are named as a party, if you have timely notified us of the claim or proceeding and have otherwise complied with the requirements of the Franchise Agreement. At our option, we and our affiliates are entitled to defend and control the defense of any proceeding arising out of your authorized use of any Mark. We do not know of any infringing uses that could materially affect your use of the Marks.

# ITEM 14 PATENTS, COPYRIGHTS, AND PROPRIETARY INFORMATION

Except as noted below, we and our affiliates do not own any patents or copyrights that are material to the Franchise. As of the date of this disclosure document, there are no patents or copyrights registered or pending, and no patent applications that are material to the Franchise.

We and our affiliates claim copyrights in the Operations Manual, construction plans, specifications and materials, printed advertising, promotional, sales, training and management materials and in related items you will use in operating your Franchise. We and our affiliates have not registered these copyrights with the U.S. Registrar of Copyrights.

There are currently no effective determinations of the U.S. Copyright Office or any court regarding any of the copyrights. There are no agreements currently in effect which significantly limit our rights to use or franchise the copyrighted materials. Also, there are no superior prior rights or infringing uses actually known to us which could materially affect your use of the copyrighted materials in any state.

Your right to use the copyrights is derived solely from the Franchise Agreement and is limited to your conduct of business in compliance with the Franchise Agreement and all applicable standards, specifications, operating procedures and rules that we require. We may require you to modify or discontinue use of any of the materials in which we claim copyrights if we determine that it becomes advisable at any time. In that case, you must comply with our directions to modify or discontinue the use of those materials within a reasonable time after notice from us.

You must immediately notify us if you learn that any person may be using our copyrighted materials without our consent or authorization. You must also immediately notify us of any challenge to your use of any copyright or claim by any person of any rights in any copyright. You must not communicate with any person other than us or our counsel about any challenge or claim to any copyright. We and our affiliates may take the action we deem appropriate and the right to control exclusively any litigation, U.S. Copyright Office proceeding or any other administrative proceeding concerning any copyright. You must sign any instruments and documents, render assistance and do those things as, in the opinion of our legal counsel, may be necessary or advisable to protect and maintain our interests in any litigation or Copyright Office or other proceeding or otherwise to protect and maintain our interests in the copyrights.

We will compensate and reimburse you for all damages for which you are held liable in any proceeding arising out of your authorized use of any copyright and for all costs you reasonably incur in defending any claim brought against you or any proceeding in which you are named as a party, if you have timely notified us of the claim or proceeding and have complied with your obligations under the Franchise Agreement. At our option, we or our affiliates are entitled to defend and control the defense of any proceeding arising out of your use of any copyright.

We and our affiliates also own the Confidential Information (as defined below) and claim copyrights in the Confidential Information. Portions of the Confidential Information required in the operation of your business will be communicated to you. However, you will not acquire any interest in any Confidential Information, other than the right to utilize Confidential Information disclosed to you in operating your Store during the term of the Franchise Agreement. You may only use the Confidential Information as outlined in the Franchise Agreement. "Confidential Information" includes any information relating to the Approved Products or the development or operation of Stores, including site selection criteria; trade secrets; recipes and methods for the preparation of Approved Products; methods, techniques, formats, standards, specifications, systems, procedures, sales and marketing techniques and knowledge of and experience in the development and operation of Stores; marketing programs for Stores; knowledge of specifications for and suppliers of certain Approved Products, materials, supplies, equipment, furnishings and fixtures; customer data; and knowledge of the operating results and financial performance of Stores, including your Store.

We and our affiliates will own and have the perpetual right to use and authorize other Stores to use, and you must fully and promptly disclose to us, all ideas, concepts, formulas, recipes, methods and techniques relating to the development and/or operation of a cookie bakery, dessert or retail snack food business and the products those businesses sell or might sell, including those that are conceived or developed by you and/or your employees during the term of the Franchise Agreement. You agree that we shall have the perpetual right to use and authorize other to use such ideas, concepts, methods and techniques and, if incorporated in the System for the development and/or operation of Stores, such ideas, concepts, methods and techniques shall become our sole and exclusive property without any consideration to you. You must not, however, test, offer or sell any new products without our prior written consent.

# ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

We recommend that you participate personally in the direct operation of your Store, although the Franchise Agreement does not specifically obligate you to do so. However, you must either manage your Store yourself, or use a full time "on Premises" manager. The manager need not have an equity interest in the franchisee entity to act as manager. We do not impose any restrictions on who you hire as manager. Both you (or one of your Entity Owners, if you are an entity) and the manager of your Store must complete all phases of our training program to our satisfaction and must participate in all other activities required to

open your Store. Replacement managers must also satisfactorily complete all phases of our training program.

If you are an entity, each Entity Owner (as defined below) must guarantee your obligations under the Franchise Agreement by signing the Guaranty attached to the Franchise Agreement, a copy of which is included in <a href="Exhibit B">Exhibit B</a> to the Franchise Agreement. We also will require your Entity Owner's spouse to sign the Guaranty.

We may require certain key employees (including you and any Entity Owner or manager) to sign a confidentiality agreement. In addition, we may require each manager of a Store to agree to the non-competition covenants described in Item 17 of this disclosure document.

An Entity Owner means, with respect to an entity, any shareholder owning directly or beneficially five percent (5%) or more of any class of securities of the entity; any general partner or co-venturer in the entity; any partner in a limited liability partnership or member in a limited liability company owning directly or beneficially 5% or more of the ownership interests in the limited liability partnership or limited liability company; the trustees or administrators of any trust or estate; and any beneficiary of a trust or estate owning, directly or beneficially, five percent (5%) or more of the interests in the trust or estate. If any Entity Owner within the scope of this definition is itself an entity (including an Entity Owner that is an Entity Owner because of this sentence), the term "Entity Owner" also includes Entity Owners (as defined in the preceding sentence) in the entity. It is the intent of this definition to "trace back" and include within the definition of Entity Owner each natural person owning the requisite interests to qualify as Entity Owners.

# ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

In operating your Store, you must use and offer for sale all of, and only, the Approved Products and any other products or services that we mandate or approve from time to time for you to sell at the Premises. The Approved Products that you initially are authorized to offer at your Store are explained in the Operations Manual (as defined in Item 11). In the future, we may change or add to the Approved Products that you are authorized to offer at the Premises and notify you of such changes or additions, as we determine, through references to the Operations Manual, bulletins and other written materials, electronic computer messages, telephonic conversations, and/or consultations at our offices or at your Store. There are no limits on our right to make such changes.

You may not use your Store for any purpose other than the operation of a Store in compliance with the Franchise Agreement. You may not: (a) offer Approved Products or materials, supplies, or inventory bearing the Marks at any site other than the Premises of your Store, (b) offer for sale any materials, supplies or inventory used in the preparation of any of the Approved Products; or (c) sell any Approved Products to any person or entity purchasing the Approved Products for resale. You may only sell Approved Products to retail customers. We otherwise do not impose any restrictions regarding the customers to whom you may sell authorized products or services.

# ITEM 17 RENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION

#### THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

|    | Provision                                 | Section in Franchise<br>Agreement | Summary                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|----|-------------------------------------------|-----------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| a. | Length of the franchise term              | Sections 1(q) and 2.1             | 10 years from the date the Franchise Agreement is fully executed.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| b. | Renewal or extension of the term          | Section 3.1                       | You may renew for 1 additional 10-year term subject to conditions.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| c. | Requirements for you to renew or extend   | Section 3.1                       | Provide at least 90 but no more than 180 days prior notice; sign then current Franchise Agreement (which may contain materially different terms and conditions than your original Franchise Agreement); Franchisee and their Affiliates and respective agents must execute a general release of all claims; refurbish and remodel the Premises at our request; remain in good standing with us during the initial term; satisfy all monetary obligations; retain the Premises for the renewal term; pay a renewal fee; follow our then-current renewal process, including any requirements for additional training and delivery of certain financial statements and records.                                                                                                                                                               |
| d. | Termination by you                        | N/A                               | Not applicable (subject to state law).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| e. | Termination by us without cause           | N/A                               | Not applicable.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| f. | Termination by us with cause              | Section 13                        | We may terminate for cause.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| g. | "Cause" defined -<br>curable defaults     | Sections 13.1 and 13.2            | You will have the following cure periods for the following defaults: (1) you have 48 hours to cure failure to comply with certain System Standards and health requirements; (2) you have 10 days to cure failure to make payments; and (3) you have 30 days to cure any other default of Franchise Agreement or any other agreement with us or an affiliate not listed above or in "17.h." below.                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| h. | "Cause" defined –<br>non-curable defaults | Section 13                        | We may terminate the Franchise Agreement immediately, upon written notice but without an opportunity to cure for: insolvency/bankruptcy, unauthorized transfer, material misrepresentation or omission in franchise application and other adverse developments, unauthorized use of Marks or Confidential Information, abandonment, breach of lease, understatements of Gross Revenue; failure to pay taxes, failure of two consecutive store inspections; two or more breaches within a 12 month period or three breaches over the term of the Franchise Agreement, financing defaults, defaults of other agreements with us or our Affiliates, failure to secure the Store Premises, possession or use of unauthorized products, failure to complete training or failure to open the Store for business within the required time period. |

|    | Provision                                                 | Section in Franchise<br>Agreement | Summary                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|----|-----------------------------------------------------------|-----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| i. | Your obligations on termination/non-renewal               | Sections 14.1, 14.3, and 14.4     | Pay us all amounts owed, close the Store for business, discontinue the use of the Marks, de-identify as a franchisee, deliver all materials containing the Marks, cancel all fictitious names, transfer all telephone numbers and directory listings to us, discontinue use of all Confidential Information and return any materials loaned to you, provide evidence that you have complied with all of the above within 30 days of the expiration or termination and comply with post-term covenants.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| j. | Assignment of contract by us                              | Section 12.1                      | No restriction on our right to transfer or assign.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| k. | "Transfer" by you –<br>defined                            | Section 1(s)                      | "Transfer" means the voluntary or involuntary, direct or indirect transfer, assignment, sale, gift, pledge, mortgage, hypothecation or other disposition (including those occurring by operation of law and a series of transfers that in the aggregate constitute a Transfer) of any of your interest in the Franchise Agreement, your Store or its assets (other than in the ordinary course of business), your right to possession of the Premises, or a direct or indirect ownership interest in you.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 1. | Our approval of transfer by you                           | Sections 12.2, 12.3 and 12.4      | All transfers subject to our prior written approval. We will not unreasonably withhold approval if specified requirements are met.  Transfers to a wholly owned corporation or limited liability company do not require our consent.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| m. | Conditions for our approval of transfer                   | Section 12.3                      | Transferee is of good moral character, transferee has sufficient business experience and resources, transferee completes our training program, you have paid all amounts owed to us or our Affiliates and third-party creditors, you and the transferee enter into our form of assignment and assumption agreement, transferee enters into our then-current form of franchise agreement, you or the transferee pays a transfer fee, we have approved the material conditions of the transfer including price, if any part of the sales price was financed you and transferee agree to subordinate that interest to our and our Affiliates' interest, you agree to comply with the post-term covenants not to compete as set forth in the Franchise Agreement, landlord consents to the transfer (if applicable), you or the transferee agrees to refurbish the Store, you and the transferee, at your cost, use a licensed escrow professional or other qualified third party acceptable to us to conduct the closing, and you and the transferee have complied with any other conditions we may reasonably require. |
| n. | Our right of first<br>refusal to acquire your<br>business | Section 12.5                      | We can match any offer to purchase the interests proposed to be transferred, with the right to substitute cash for other forms of payment, equal credit to the offer, no less than 60 days to close the purchase, and the right to receive all customary representations and warranties.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 0. | Our option to<br>purchase your<br>business                | Sections 14.2 and 16              | We may, at our option, purchase your Store upon termination of Franchise Agreement unless we are in default. Under the security agreement contained in Section 16 of the Franchise Agreement we may exercise all rights of a secured creditor granted to us by law, in addition to our other rights under this Agreement if you default.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| p. | Your death or<br>disability                               | Section 12.6                      | If you are an individual, upon your death or permanent disability or, if you are an Entity, upon the death or permanent disability of an individual owner of a Controlling Interest in you, the executor, administrator, conservator or other personal representative of that person will transfer his interest in this Agreement or his Controlling Interest in you within six (6) months from the date of death or permanent disability, to a transferee approved by us.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |

|    | Durantotan                                                             | Section in Franchise                  | C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|----|------------------------------------------------------------------------|---------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|    | Provision                                                              | Agreement                             | Summary  Voy your Entity Owner(s) (and their direct family mambars) and your                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| q. | Non-competition covenants during the term of the franchise             | Section 11.1                          | You, your Entity Owner(s) (and their direct family members) and your managers will not: (1) have any direct or indirect interest in a Competitive Business (as defined below); (2) perform services as a director, officer, manager, employee, consultant, representative, agent or otherwise for a Competitive Business; (3) divert or attempt to divert business to a Competitive Business; or (4) use or duplicate the System or System Standards for any other endeavor. A "Competitive Business" means any business that operates, grants franchises or licenses to others to operate, acts as a landlord to, or otherwise provides products, services or assistance to a restaurant, retail outlet or other food service business (other than a Mrs. Fields Store operated pursuant to a franchise agreement with us or our affiliate) which offers or sells cookies, bakery items, snack foods or desserts as primary or featured items or for which the sale of cookies, bakery items, snack foods or desserts accounts (either individually or in the aggregate) accounts for at least 10% of its sales during any calendar month. |
| r. | Non-competition covenants after the franchise is terminated or expires | Section 11.2                          | For 2 years after termination of the Franchise Agreement, you will not have any direct or indirect interest as an owner, investor, partner, director, officer, employee, consultant, representative or agent or in any other capacity in any Competitive Business located or operating within (a) 10 miles of your Store, or (b) 10 miles of any Store, except Stores that you operate under agreements with us or our Affiliates. If we purchase your Store pursuant to our right in Section 14.5 of the Franchise Agreement, this non-competition covenant will extend to 3 years.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| S. | Modification of the agreement                                          | Sections 11.4, 18.1,<br>18.2 and 18.7 | Subject to automatic modification to conform to mandatory provisions of applicable law. Other modifications require mutual consent.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| t. | Integration/merger clause                                              | Sections 18.5(c)                      | Only the terms of Franchise Agreement are binding (subject to state law). Any representations or promises outside of the disclosure document and Franchise Agreement may not be enforceable. However, nothing in the Franchise Agreement is intended to disclaim the representations made in this disclosure document.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| u. | Dispute resolution by arbitration or mediation                         | Section 17.8                          | All controversies, disputes, or claims (notwithstanding temporary restraining orders or preliminary injunctive relief) between us or our Affiliates, on the one hand, and you and your Affiliates, on the other hand, must be submitted to the American Arbitration Association for binding arbitration in accordance with the then-current Commercial Arbitration Rules of the American Arbitration Association; and proceedings will be conducted at a suitable location chosen by the arbitrator in Wilmington, Delaware. These provisions may be subject to applicable state law.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| v. | Choice of forum                                                        | Section 17.5                          | Disputes must be conducted in Wilmington, Delaware (subject to applicable state law).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| w. | Choice of law                                                          | Section 17.4                          | Delaware law, unless superseded by applicable federal or state law.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |

#### AREA DEVELOPMENT AGREEMENT

This table lists certain important provisions in the Area Development Agreement and related agreements. You should read these provisions in the agreements attached to this disclosure document.

|    |                                                                              | Section in Area              |                                                                                                                                                                                                                                                                                                                                                   |
|----|------------------------------------------------------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|    | Provision                                                                    | <b>Development Agreement</b> | Summary                                                                                                                                                                                                                                                                                                                                           |
| a. | Length of the franchise term                                                 | Section 6                    | Varies                                                                                                                                                                                                                                                                                                                                            |
| b. | Renewal or extension of the term                                             | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| c. | Requirements for you to renew or extend                                      | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| d. | Termination by you                                                           | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| e. | Termination by us without cause                                              | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| f. | Termination by us with cause                                                 | Section 7                    | We can terminate only if you default under the Franchise Agreement or Area Development Agreement                                                                                                                                                                                                                                                  |
| g. | "Cause" defined - curable defaults                                           | Section 7(B)                 | 30 days to cure defaults under the Area Development Agreement                                                                                                                                                                                                                                                                                     |
| h. | "Cause" defined – non-<br>curable defaults                                   | Section 7(A)                 | Repeated noncompliance, bankruptcy, you attempt to subfranchise or sublicense in any manner                                                                                                                                                                                                                                                       |
| i. | Your obligations on termination/non-renewal                                  | Section 7(D)                 | All rights revert to us and right develop Stores within the Designated Area will cease                                                                                                                                                                                                                                                            |
| j. | Assignment of contract by us                                                 | Section 8(A)                 | No restriction on our right to transfer                                                                                                                                                                                                                                                                                                           |
| k. | "Transfer" by you – defined                                                  | Section 8(B)                 | May transfer with our approval                                                                                                                                                                                                                                                                                                                    |
| 1. | Our approval of transfer by you                                              | Section 8(B)                 | We have the right to approve all transfers                                                                                                                                                                                                                                                                                                        |
| m. | Conditions for our approval of transfer                                      | Section 8(B)                 | Compliance with terms of Area Development Agreement, our approval or proposed transferee, purchase price of undeveloped franchised does not exceed then-current development fees we charge franchisees under an Area Development Agreement; proposed transferee completes our training program, payment of transfer fee, executed general release |
| n. | Our right of first refusal to acquire your business                          | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| 0. | Our option to purchase your business                                         | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| p. | Your death or disability                                                     | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| q. | Non-competition covenants during the term of the franchise                   | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| r. | Non-competition covenants<br>after the franchise is<br>terminated or expires | N/A                          | Not applicable                                                                                                                                                                                                                                                                                                                                    |
| S. | Modification of the agreement                                                | Section 11                   | Requires writing by both parties                                                                                                                                                                                                                                                                                                                  |
| t. | Integration/merger clause                                                    | Section 11                   | Only terms of Area Development Agreement are binding, provided that this provision does not disclaim representations made in the FDD                                                                                                                                                                                                              |

|    | Provision                                      | Section in Area Development Agreement | Summary                     |
|----|------------------------------------------------|---------------------------------------|-----------------------------|
| u. | Dispute resolution by arbitration or mediation | Section 10                            | Same as Franchise Agreement |
| v. | Choice of forum                                | Section 10                            | Same as Franchise Agreement |
| w. | Choice of law                                  | Section 10                            | Same as Franchise Agreement |

#### ITEM 18 <u>PUBLIC FIGURES</u>

We do not use any public figure to promote our Franchise.

#### ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

#### The Data Set

The data set represents unaudited historical data that we received from 75 traditional Stores (including 69 Stores and 6 Kiosks) in which Mrs. Fields is the dominant brand (with over 55% of sales for such Stores and Kiosks derived from Mrs. Fields) for calendar year 2024 ("Calendar Year 2024").

As of the end of Calendar Year 2024, there were 113 Stores in operation, all of which were owned by third-party franchisees. For the data set used in this Item 19, we excluded from that group (a) 5 Stores which opened during Calendar Year 2024 (and were not open and operational for 12 full months as of the end of Calendar Year 2024) and (b) 33 non-traditional Stores (those that, for example, are only open part time, that operate seasonally, that are located in non-traditional locations such as gas stations, or are operated as an add-on to another primary brand). The non-traditional Stores were excluded for two reasons: (1) because of the wide variation in factors that made them non-traditional and (2) because the focus of our sales efforts is on selling franchises for traditional locations. As a result, we do not make financial performance representations with respect to non-traditional Stores.

#### **Data Presented**

To determine the average Gross Revenue for each group shown in the tables below, we took the group's total Gross Revenue, then divided that number by the number of Stores in the group. For example, to determine the average Gross Revenue for the 75 Stores that made up the entire data set, we added the Gross Revenue of all 75 Stores for Calendar Year 2024, then we divided that number by 75.

In the results below, we also report, for each group, the range and the median. The range reflects the low and high data point for each group. The median is the data point that is the center of all data points used for the group, except that, in a given group of Stores where there is an even number of Stores, the median is derived by adding the two numbers in the middle of the data set for the given group of Stores, then dividing that number by two.

"Gross Revenue" means the aggregate of all revenue, in whatever form (including the value of all barter, exchange, trade or other credit), attributed to the operation of your Store (whether or not in compliance with this Agreement) based on the gross (undiscounted) price of all products and services sold at or through your Store and all other revenue sources, but excluding all federal, state or municipal sales, use, or service taxes collected from customers and paid to the appropriate taxing authority. Gift certificate, gift card and similar program payments will be included in Gross Revenue as and in accordance with the policies set forth from time to time in the Operations Manual. Gross Revenue also includes all insurance proceeds you receive for loss of business due to a casualty or a similar event. This is the same definition of Gross Revenue on which you will be required to pay royalties and Brand Fund Contributions.

In Tables 1 and 4 below, we report the results for all Stores and Kiosks in the data set. In Tables 2 and 3, we present data for two subsets of the entire data set – data from only Stores in Table 2 and data from only Kiosks in Table 3.

In Tables 1 through 3, we report the results for each data set or subset by quartile. Each quartile represents 25% of the Stores or Kiosks that form the data set or subset. In Table 4, we present data for the entire data set but have reported the results based on the decade in which the Store or Kiosk originally opened.

#### The Results

Table 1: Gross Revenue for Calendar Year 2024 by Quartile for Stores and Kiosks

| Quartiles                | Average<br>Gross<br>Revenue | Number and Percentage of<br>Stores Attaining or Exceeding<br>Average Gross Revenue | Median Gross<br>Revenue | Lowest Gross<br>Revenue | Highest Gross<br>Revenue |
|--------------------------|-----------------------------|------------------------------------------------------------------------------------|-------------------------|-------------------------|--------------------------|
| Top Quartile             | \$646,761                   | 8 of 19 (42%)                                                                      | \$616,643               | \$479,647               | \$1,005,526              |
| 2 <sup>nd</sup> Quartile | \$392,851                   | 7 of 19 (37%)                                                                      | \$378,092               | \$331,730               | \$470,266                |
| 3 <sup>rd</sup> Quartile | \$282,672                   | 8 of 19 (42%)                                                                      | \$282,081               | \$239,375               | \$331,292                |
| Bottom Quartile          | \$183,936                   | 11 of 19 (58%)                                                                     | \$201,548               | \$112,705               | \$236,393                |
| Total                    | \$378,393                   | 34 of 75 (45%)                                                                     | \$331,730               | \$112,705               | \$1,005,526              |

Table 2: Gross Revenue for Calendar Year 2024 by Quartile for Inline Stores Only

| Quartiles                | Average<br>Gross<br>Revenue | Number and Percentage of<br>Stores Attaining or Exceeding<br>Average Gross Revenue | Median Gross<br>Revenue | Lowest Gross<br>Revenue | Highest Gross<br>Revenue |
|--------------------------|-----------------------------|------------------------------------------------------------------------------------|-------------------------|-------------------------|--------------------------|
| Top Quartile             | \$646,939                   | 7 of 17 (41%)                                                                      | \$616,643               | \$492,197               | \$1,005,526              |
| 2 <sup>nd</sup> Quartile | \$399,111                   | 8 of 18 (44%)                                                                      | \$383,258               | \$331,730               | \$479,647                |
| 3 <sup>rd</sup> Quartile | \$283,445                   | 7 of 17 (41%)                                                                      | \$282,081               | \$236,393               | \$331,292                |
| Bottom Quartile          | \$180,850                   | 9 of 17 (53%)                                                                      | \$199,064               | \$112,705               | \$235,888                |
| Total                    | \$377,898                   | 31 of 69 (45%)                                                                     | \$331,730               | \$112,705               | \$1.005,526              |

Table 3: Gross Revenue for Calendar Year 2024 by Quartile for Kiosks Only

| Quartiles                | Average<br>Gross<br>Revenue | Number and Percentage of<br>Stores Attaining or Exceeding<br>Average Gross Revenue | Median Gross<br>Revenue | Lowest Gross<br>Revenue | Highest Gross<br>Revenue |
|--------------------------|-----------------------------|------------------------------------------------------------------------------------|-------------------------|-------------------------|--------------------------|
| Top Quartile             | \$810,862                   | 1 of 1 (100%)                                                                      | \$810,862               | \$810,862               | \$810,862                |
| 2 <sup>nd</sup> Quartile | \$379,908                   | 1 of 2 (50%)                                                                       | \$379,908               | \$378,092               | \$381,725                |
| 3 <sup>rd</sup> Quartile | \$285,059                   | 1 of 2 (50%)                                                                       | \$285,059               | \$269,801               | \$300,317                |
| Bottom Quartile          | \$239,375                   | 1 of 1 (100%)                                                                      | \$239,375               | \$239,375               | \$239,375                |
| Total                    | \$396,695                   | 4 of 6 (67%)                                                                       | \$339,205               | \$239,375               | \$810,862                |

Table 4: Gross Revenue for Calendar Year 2024 by Decade in which the Store or Kiosk Originally Opened

| Fiscal Year in which<br>the Store Originally<br>Opened | Average<br>Gross<br>Revenue | Number and Percentage of<br>Stores Attaining or Exceeding<br>Average Gross Revenue | Median Gross<br>Revenue | Lowest Gross<br>Revenue | Highest Gross<br>Revenue |
|--------------------------------------------------------|-----------------------------|------------------------------------------------------------------------------------|-------------------------|-------------------------|--------------------------|
| 2015 to 2024                                           | \$394,131                   | 10 of 28 (36%)                                                                     | \$314,538               | \$161,548               | \$1,005,526              |
| 2005 to 2014                                           | \$389,271                   | 13 of 30 (43%)                                                                     | \$360,018               | \$54,637                | \$842,650                |
| 1995 to 2004                                           | \$319,025                   | 9 of 17 (53%)                                                                      | \$323,986               | \$112,705               | \$635,543                |
| 1981 to 1994                                           | \$282,932                   | 1 of 1 (100%)                                                                      | \$282,932               | \$282,932               | \$282,932                |
| Total                                                  | \$379,402                   | 33 of 75 (43%)                                                                     | \$331,730               | \$112,705               | \$1,005,526              |

#### Notes to Item 19

- 1. The numbers provided in this Item 19 do not reflect the expenses associated with operating a Store (and, thus, do not reflect gross or net profit).
- 2. Some outlets have earned this amount. Your individual results may differ. There is no assurance that you'll earn as much.
- 3. Because the data set is comprised entirely of franchised Stores, the data is based on the unaudited results that franchisees reported to us directly or through their Stores' point-of-sale system.
- 4. The data is historical data received from franchised Stores and not a projection of your potential future performance.
- 5. Written substantiation for the financial performance representations above will be made available to prospective franchisees upon reasonable request.

Other than the preceding financial performance representation, Mrs. Fields Franchising, LLC, does not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting our Interim Chief Executive Officer, James Carnrite, Mrs. Fields Franchising, LLC, 1717 S. 4800 W., Salt Lake City, Utah 84104, (801) 412-8890, the Federal Trade Commission, and the appropriate state regulatory agencies.

#### ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

#### **TABLE NUMBER 1**

# Systemwide Outlet Summary For Years 2022 to 2024<sup>(1)</sup>

| Store Type    | Year <sup>(1)</sup>   | Outlets at the Start of the Year | Outlets at the End<br>of the Year | Net Change |
|---------------|-----------------------|----------------------------------|-----------------------------------|------------|
|               | 2022                  | 128                              | 125                               | -3         |
| Franchised    | 2023                  | 125                              | 125                               | 0          |
| Franchised    | 7/1/2023 – 12/31/2023 | 125                              | 121                               | -4         |
|               | 2024                  | 121                              | 113                               | -8         |
|               | 2022                  | 0                                | 0                                 | 0          |
| Company-Owned | 2023                  | 0                                | 0                                 | 0          |
| Company-Owned | 7/1/2023 – 12/31/2023 | 0                                | 0                                 | 0          |
|               | 2024                  | 0                                | 0                                 | 0          |
|               | 2022                  | 128                              | 125                               | -3         |
| Total         | 2023                  | 125                              | 125                               | 0          |
| Total         | 7/1/2023 – 12/31/2023 | 125                              | 121                               | -4         |
|               | 2024                  | 121                              | 113                               | -8         |

(1) The numbers in these Item 20 tables reflect information as of July 1, 2023 and July 3, 2022 as our prior fiscal years ended as of those dates. Following the Transaction described in Item 1, our fiscal year end currently is December 31. As a result, we have added a row to disclose information for the balance of calendar year 2023.

TABLE NUMBER 2

Transfers of Outlets from Franchisee to New Owners (other than the Franchisor)
For Years 2022 to 2024

| State      | Year                  | Number of Transfers |
|------------|-----------------------|---------------------|
|            | 2022                  | 1                   |
| California | 2023                  | 1                   |
| Camornia   | 7/1/2023 - 12/31/2023 | 0                   |
|            | 2024                  | 0                   |
|            | 2022                  | 0                   |
| Illinois   | 2023                  | 1                   |
| Illinois   | 7/1/2023 - 12/31/2023 | 2                   |
|            | 2024                  | 2                   |
|            | 2022                  | 1                   |
| Kansas     | 2023                  | 0                   |
| Kalisas    | 7/1/2023 - 12/31/2023 | 0                   |
|            | 2024                  | 0                   |
|            | 2022                  | 1                   |
| Michigan   | 2023                  | 0                   |
| Wilcingan  | 7/1/2023 – 12/31/2023 | 0                   |
|            | 2024                  | 0                   |
|            | 2022                  | 0                   |
| New Jersey | 2023                  | 0                   |
| New Jersey | 7/1/2023 - 12/31/2023 | 0                   |
|            | 2024                  | 1                   |

| State        | Year                  | Number of Transfers |
|--------------|-----------------------|---------------------|
|              | 2022                  | 0                   |
| Donnaylyania | 2023                  | 1                   |
| Pennsylvania | 7/1/2023 - 12/31/2023 | 0                   |
|              | 2024                  | 0                   |
|              | 2022                  | 0                   |
| Elonido      | 2023                  | 0                   |
| Florida      | 7/1/2023 - 12/31/2023 | 1                   |
|              | 2024                  | 0                   |
|              | 2022                  | 0                   |
| Towas        | 2023                  | 0                   |
| Texas        | 7/1/2023 - 12/31/2023 | 1                   |
|              | 2024                  | 0                   |
|              | 2022                  | 3                   |
| TOTAL        | 2023                  | 3                   |
| IOIAL        | 7/1/2023 - 12/31/2023 | 4                   |
|              | 2024                  | 3                   |

<sup>(1)</sup> The numbers in these Item 20 tables reflect information as of July 1, 2023 and July 3, 2022 as our prior fiscal years ended as of those dates. Following the Transaction described in Item 1, our fiscal year end currently is December 31. As a result, we have added a row to disclose information for the balance of calendar year 2023.

#### **TABLE NUMBER 3**

#### Status of Franchised Outlets For Years 2022 to 2024

| State      | Year            | Outlets at<br>Start of Year | Outlets<br>Opened | Terminations | Non-<br>Renewals | Reacquired by<br>Franchisor | Ceased Operations -<br>Other Reasons | Outlets at End<br>of the Year |
|------------|-----------------|-----------------------------|-------------------|--------------|------------------|-----------------------------|--------------------------------------|-------------------------------|
|            | 2022            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2023            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Alaska     | 7/1/23-12/31/23 | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2024            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2022            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| A 1 - 1    | 2023            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Alabama    | 7/1/23-12/31/23 | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2024            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2022            | 0                           | 1                 | 0            | 0                | 0                           | 0                                    | 1                             |
| A          | 2023            | 1                           | 1                 | 0            | 0                | 0                           | 0                                    | 2                             |
| Arizona    | 7/1/23-12/31/23 | 2                           | 0                 | 0            | 0                | 0                           | 1                                    | 1                             |
|            | 2024            | 1                           | 0                 | 0            | 0                | 0                           | 1                                    | 1                             |
|            | 2022            | 37                          | 0                 | 0            | 0                | 0                           | 2                                    | 35                            |
| California | 2023            | 35                          | 0                 | 0            | 0                | 0                           | 0                                    | 35                            |
| Camornia   | 7/1/23-12/31/23 | 35                          | 0                 | 0            | 0                | 0                           | 2                                    | 33                            |
|            | 2024            | 33                          | 2                 | 0            | 0                | 0                           | 3                                    | 32                            |
|            | 2022            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Calamada   | 2023            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Colorado   | 7/1/23-12/31/23 | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2024            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2022            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Doloviore  | 2023            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
| Delaware   | 7/1/23-12/31/23 | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |
|            | 2024            | 1                           | 0                 | 0            | 0                | 0                           | 0                                    | 1                             |

| ~             |                 | Outlets at    | Outlets |              | Non-     | Reacquired by | <b>Ceased Operations -</b> |             |
|---------------|-----------------|---------------|---------|--------------|----------|---------------|----------------------------|-------------|
| State         | Year            | Start of Year | Opened  | Terminations | Renewals | Franchisor    | Other Reasons              | of the Year |
|               | 2022            | 7             | 2       | 0            | 0        | 0             | 0                          | 9           |
| Florida       | 2023            | 7             | 0       | 0            | 0        | 0             | 1                          | 8           |
|               | 7/1/23-12/31/23 | 8             | 0       | 0            | 0        | 0             | 0                          | 8           |
|               | 2024            | 8             | 0       | 0            | 0        | 0             | 1                          | 7           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Georgia       | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2022            | 0             | 0       | 0            | 0        | 0             | 0                          | 0           |
| Hawaii        | 2023            | 0             | 1       | 0            | 0        | 0             | 0                          | 1           |
| 110000        | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Idaho         | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Iduno         | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2022            | 21            | 0       | 0            | 0        | 0             | 2                          | 19          |
| Illinois      | 2023            | 19            | 1       | 0            | 0        | 0             | 1                          | 18          |
| Illinois      | 7/1/23-12/31/23 | 18            | 0       | 0            | 0        | 0             | 0                          | 18          |
|               | 2024            | 18            | 1       | 0            | 0        | 0             | 0                          | 19          |
|               | 2022            | 2             | 0       | 0            | 0        | 0             | 0                          | 2           |
| Indiana       | 2023            | 2             | 0       | 0            | 0        | 0             | 0                          | 2           |
| Iliulalia     | 7/1/23-12/31/23 | 2             | 0       | 0            | 0        | 0             | 0                          | 2           |
|               | 2024            | 2             | 0       | 0            | 0        | 0             | 2                          | 0           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Kansas        | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Kansas        | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2022            | 3             | 0       | 0            | 0        | 0             | 0                          | 3           |
| Louisiana     | 2023            | 3             | 0       | 0            | 0        | 0             | 0                          | 3           |
| Louisiana     | 7/1/23-12/31/23 | 3             | 0       | 0            | 0        | 0             | 1                          | 2           |
|               | 2024            | 2             | 0       | 0            | 0        | 0             | 0                          | 2           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| 3.6 :         | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Maine         | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 1                          | 0           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Maryland      | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 1                          | 0           |
|               | 2022            | 1             | 1       | 0            | 0        | 0             | 0                          | 2           |
|               | 2023            | 2             | 0       | 0            | 0        | 0             | 1                          | 1           |
| Massachusetts | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2022            | 7             | 0       | 0            | 0        | 0             | 1                          | 6           |
|               | 2023            | 6             | 0       | 0            | 0        | 0             | 1                          | 5           |
| Michigan      | 7/1/23-12/31/23 | 5             | 0       | 0            | 0        | 0             | 0                          | 5           |
|               | 2024            | 5             | 0       | 0            | 0        | 0             | 0                          | 5           |
|               | 2022            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2023            | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
| Missouri      | 7/1/23-12/31/23 | 1             | 0       | 0            | 0        | 0             | 0                          | 1           |
|               | 2024            | 1             | 0       | 0            | 0        | 0             | 1                          | 0           |
|               | 2027            | 1             | U       | U            | U        | U             | 1                          | L           |

| Ctata            | Vaca            | Outlets at    | Outlets | T               | Non-     | Reacquired by | _                        | Outlets at End |
|------------------|-----------------|---------------|---------|-----------------|----------|---------------|--------------------------|----------------|
| State            | Year 2022       | Start of Year | Opened  | Terminations () | Renewals | Franchisor    | Other Reasons<br>()      | of the Year    |
|                  | 2022            | 6             | 0       | 0               | 0        | 0             |                          | 6<br>7         |
| Nevada           |                 | 6             | 1       |                 |          |               | 0                        |                |
|                  | 7/1/23-12/31/23 | 7             | 1       | 0               | 0        | 0             | 2                        | 6              |
|                  | 2024            | 6             | 1       | 0               | 0        | 0             | 1                        | 5              |
|                  | 2022            | 9             | 0       | 0               | 0        | 0             | 0                        | 9              |
| New Jersey       | 2023            | 9             | 1       | 0               | 0        | 0             | 0                        | 10             |
|                  | 7/1/23-12/31/23 | 10            | 0       | 0               | 0        | 0             | 0                        | 10             |
|                  | 2024            | 10            | 0       | 0               | 0        | 0             | 0                        | 10             |
|                  | 2022            | 6             | 0       | 0               | 0        | 0             | 1                        | 5              |
| New York         | 2023            | 5             | 1       | 0               | 0        | 0             | 1                        | 6              |
| Tiew Tork        | 7/1/23-12/31/23 | 6             | 1       | 0               | 0        | 0             | 0                        | 7              |
|                  | 2024            | 7             | 1       | 0               | 0        | 0             | 0                        | 8              |
|                  | 2022            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
| North Carolina   | 2023            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
| TYOTHI Catollila | 7/1/23-12/31/23 | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2024            | 1             | 1       | 0               | 0        | 0             | 0                        | 2              |
|                  | 2022            | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
| Ohio             | 2023            | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
| Onio             | 7/1/23-12/31/23 | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
|                  | 2024            | 2             | 0       | 0               | 0        | 0             | 1                        | 1              |
|                  | 2022            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
| 0111             | 2023            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
| Oklahoma         | 7/1/23-12/31/23 | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2024            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2022            | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
|                  | 2023            | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
| Pennsylvania     | 7/1/23-12/31/23 | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
|                  | 2024            | 2             | 0       | 0               | 0        | 0             | 0                        | 2              |
|                  | 2022            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2023            | 1             | 0       | 0               | 0        | 0             | 1                        | 0              |
| Tennessee        | 7/1/23-12/31/23 | 0             | 0       | 0               | 0        | 0             | 0                        | 0              |
|                  | 2024            | 0             | 0       | 0               | 0        | 0             | 0                        | 0              |
|                  | 2022            | 10            | 1       | 0               | 0        | 0             | 2                        | 9              |
|                  | 2023            | 10            | 0       | 0               | 0        | 0             | 0                        | 9              |
| Texas            | 7/1/23-12/31/23 | 9             | 0       | 0               | 0        | 0             | 0                        | 9              |
|                  |                 |               |         |                 |          |               | 1                        |                |
|                  | 2024            | 9             | 0       | 0               | 0        | 0             | 0                        | 8              |
|                  | 2022            | 1             |         | 0               | 0        | 0             | 0                        | 1              |
| Utah             |                 | 1             | 0       |                 | 0        | 0             | 0                        | 1              |
|                  | 7/1/23-12/31/23 | 1             | 0       | 0               |          |               | 0                        | 1              |
|                  | 2024            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2022            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
| Wisconsin        | 2023            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 7/1/23-12/31/23 | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2024            | 1             | 0       | 0               | 0        | 0             | 0                        | 1              |
|                  | 2022            | 128           | 5       | 0               | 0        | 0             | 8                        | 125            |
| TOTAL            | 2023            | 125           | 6       | 0               | 0        | 0             | 6                        | 125            |
| 1011111          | 7/1/23-12/31/23 | 125           | 2       | 0               | 0        | 0             | 6                        | 121            |
|                  | 2024            | 121           | 5       | 0               | 0        | 0             | 13<br>23 and July 3 2022 | 113            |

<sup>(1)</sup> The numbers in these Item 20 tables reflect information as of July 1, 2023 and July 3, 2022 as our prior fiscal years ended as of those dates. Following the Transaction described in Item 1, our

fiscal year end currently is December 31. As a result, we have added a row to disclose information for the balance of calendar year 2023.

#### **TABLE NUMBER 4**

#### Status of Company-Owned Outlets For Years 2022 to 2024

|        |                 | Outlets at the Start | Outlets | Outlets Reacquired | Outlets | Outlets Sold to | Outlets at the  |
|--------|-----------------|----------------------|---------|--------------------|---------|-----------------|-----------------|
| State  | Year            | of the Year          | Opened  | from Franchisees   | Closed  | Franchisees     | End of the Year |
|        | 2022            | 0                    | 0       | 0                  | 0       | 0               | 0               |
| TD 4 1 | 2023            | 0                    | 0       | 0                  | 0       | 0               | 0               |
| Total  | 7/1/23-12/31/23 | 0                    | 0       | 0                  | 0       | 0               | 0               |
|        | 2024            | 0                    | 0       | 0                  | 0       | 0               | 0               |

<sup>(1)</sup> The numbers in these Item 20 tables reflect information as of July 1, 2023 and July 3, 2022 as our prior fiscal years ended as of those dates. Following the Transaction described in Item 1, our fiscal year end currently is December 31. As a result, we have added a row to disclose information for the balance of calendar year 2023.

#### **TABLE NUMBER 5**

#### Projected Openings As of December 31, 2024

|                | Franchise Agreements Signed | Projected New Franchised               | <b>Projected New Company-Owned</b> |
|----------------|-----------------------------|----------------------------------------|------------------------------------|
| State          | But Outlet Not Opened       | <b>Outlets in the Next Fiscal Year</b> | Outlets in the Next Fiscal Year    |
| California     | 0                           | 1                                      | 0                                  |
| Florida        | 0                           | 2                                      | 0                                  |
| Illinois       | 0                           | 1                                      | 0                                  |
| Nevada         | 0                           | 1                                      | 0                                  |
| North Carolina | 0                           | 1                                      | 0                                  |
| Texas          | 1                           | 2                                      | 0                                  |
| TOTAL          | 1                           | 8                                      | 0                                  |

Exhibit F to this disclosure document discloses all of our franchisees, addresses, and telephone numbers as of the end of our most recent fiscal year (ended December 31, 2024), as well as all outlets who ceased to operate during our last fiscal year, or franchisees who failed to communicate with us within 10 weeks of the date of this disclosure document. If you buy a franchise for a Store, your contact information may be disclosed to other buyers when you leave the franchise system.

During the last 3 fiscal years, franchisees have signed confidentiality clauses. In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experience with the Mrs. Fields franchise system. You may wish to speak with current and former franchisees, but be aware that not all such franchisees will be able to communicate with you.

We have not created, sponsored, or endorsed any trademark-specific franchisee organization associated with the System, and no such organization has asked to be included in this disclosure document.

#### ITEM 21 FINANCIAL STATEMENTS

Exhibit A contains the audited consolidated financial statements of Mrs. Fields Franchising Holdco, LLC (MFF Holdco) for the period from August 1, 2024 (date of formation) to December 31, 2024. Following the Transaction, we are now owned by a new parent entity and cannot provide the otherwise required three years of financial statements. MFF Holdco has absolutely and unconditionally guaranteed to assume our duties and obligations to you under the Franchise Agreement if we become unable to perform our duties and obligations. The Guarantee of Performance is attached to this Disclosure Document as Exhibit A.

#### ITEM 22 CONTRACTS

The following agreements proposed for use regarding the offering of a Store are attached to this disclosure document:

Exhibit B – Franchise Agreement

Exhibit C – Area Development Agreement

Exhibit D – Form Deposit Agreement

Exhibit I – Sample Release Agreement

Exhibit J – Disclosure Acknowledgment Agreement

#### ITEM 23 RECEIPTS

Attached as  $\underline{\text{Exhibit } K}$  to this disclosure document are detachable acknowledgments of receipt. Please sign and return to us our copy of the receipt (Copy for Franchisor), and sign and retain for your records your copy of the receipt (Copy for Prospective Franchisee).

## EXHIBIT A

## FINANCIAL STATEMENTS AND GUARANTEE OF PERFORMANCE

# **Consolidated Financial Statements**

# Mrs. Fields Franchising Holdco, LLC

For the Period from August 1, 2024 (date of formation) to December 31, 2024



# **CONTENTS**

|                                                      | Page |
|------------------------------------------------------|------|
| Independent Auditor's Report                         | 3    |
| Consolidated Financial Statements:                   |      |
| Consolidated Balance Sheet                           | 5    |
| Consolidated Statement of Comprehensive Income       | 6    |
| Consolidated Statement of Changes in Member's Equity | 7    |
| Consolidated Statement of Cash Flows                 | 8    |
| Notes to Consolidated Financial Statements           | 9    |



To the Member Mrs. Fields Franchising Holdco, LLC Salt Lake City, Utah

#### **Independent Auditor's Report**

#### **Opinion**

We have audited the consolidated financial statements of Mrs. Fields Franchising Holdco, LLC (the Company), which comprise the consolidated balance sheet as of December 31, 2024 and the related consolidated statements of comprehensive income, changes in member's equity and cash flows for the period from August 1, 2024 (date of formation) to December 31, 2024, and the related notes to the consolidated financial statements.

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2024 and the results of its operations and its cash flows for the period from August 1, 2024 (date of formation) to December 31, 2024 in accordance with accounting principles generally accepted in the United States of America.

#### **Basis for Opinion**

We conducted our audit in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the *Auditor's Responsibilities for the Audit of the Consolidated Financial Statements* section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

#### Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are issued or available to be issued.

#### Auditor's Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

GBQ Partners LLC

Cincinnati, Ohio April 28, 2025

Consolidated Balance Sheet December 31, 2024

| ASSETS                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Current Assets                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Cash and cash equivalents                    | \$ 526,107                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Accounts receivable                          | 383,652                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Prepaid expenses and other current assets    | 18,412                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Due from related party  Total current assets | 725,334<br>1,653,505                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Total cultent assets                         | 1,055,505                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Noncurrent Assets                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Deferred expenses                            | 20,630                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Intangible assets, net                       | 16,323,444                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Goodwill, net                                | 2,521,946                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| Property and equipment, net                  | 57,349                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Total noncurrent assets                      | 18,923,369                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| TOTAL ASSETS                                 | \$ 20,576,874                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| LIABILITIES AND MEMBER'S EQUITY              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Current Liabilities                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Accounts payable                             | \$ 187,694                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Accrued expenses                             | 73,654                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Accrued compensation                         | 90,209                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Gift card liability, net                     | 32,772                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Current portion of deferred revenue          | 41,349                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Total current liabilities                    | 425,678                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Noncurrent Liabilities                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Deferred revenue, net of current portion     | 209,540                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Total noncurrent liabilities                 | 209,540                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Total liabilities                            | 635,218                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| Member's Equity                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| Member's equity                              | 19,913,681                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| Accumulated other comprehensive income       | 27,975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| Total member's equity                        | 19,941,656                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                              | - / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / - / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / - / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / / |
| TOTAL LIABILITIES AND MEMBER'S EQUITY        | \$ 20,576,874                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |

Consolidated Statement of Comprehensive Income For the Period from August 1, 2024 (date of formation) to December 31, 2024

| Revenues  Royalty fees  Marketing and advertising fees  Franchise related fees  Other income  Total revenues                            | \$<br>1,008,872<br>256,241<br>32,239<br>311,219<br>1,608,571 |
|-----------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------|
| Operating Costs and Expenses General and administrative expense Marketing and advertising expense Depreciation and amortization expense | 978,620<br>250,141<br>163,033                                |
| Total Operating Costs and Expenses                                                                                                      | 1,391,794                                                    |
| Net Income                                                                                                                              | 216,777                                                      |
| Unrealized Gain on Foreign Currency Translation                                                                                         | 27,975                                                       |
| Comprehensive Income                                                                                                                    | \$<br>244,752                                                |

Consolidated Statement of Changes in Member's Equity For the Period from August 1, 2024 (date of formation) to December 31, 2024

|                                                                                                                                  | Member's<br>Equity         | Accumulated Other Comprehensive Income | Total<br>Member's<br>Equity     |
|----------------------------------------------------------------------------------------------------------------------------------|----------------------------|----------------------------------------|---------------------------------|
| Balance - August 1, 2024 (date of formation)                                                                                     | \$ -                       | \$ -                                   | \$ -                            |
| Noncash capital contribution (see Nature<br>and Scope of Business Note)<br>Net income<br>Unrealized gain on currency translation | 19,696,904<br>216,777<br>- | -<br>-<br>27,975                       | 19,696,904<br>216,777<br>27,975 |
| Balance - December 31, 2024                                                                                                      | \$ 19,913,681              | \$ 27,975                              | \$ 19,941,656                   |

Consolidated Statement of Cash Flows
For the Period from August 1, 2024 (date of formation)
to December 31, 2024

| Cash Flows from Operating Activities                           |                  |
|----------------------------------------------------------------|------------------|
| Net income                                                     | \$<br>216,777    |
| Adjustments to reconcile net income to net cash                |                  |
| flows used in operating activities:                            |                  |
| Depreciation and amortization expense                          | 163,033          |
| Accretion of deferred revenues                                 | (32,239)         |
| Changes in operating assets and liabilities:                   |                  |
| Accounts receivable                                            | 14,302           |
| Prepaid expenses and other current assets                      | 5,907            |
| Due from related party                                         | 144,443          |
| Deferred expenses                                              | 6,180            |
| Accounts payable                                               | (26,087)         |
| Accrued expenses                                               | 67,748           |
| Accrued compensation                                           | (48,778)         |
| Gift card liability                                            | (3,348)          |
| Net cash provided by operating activities                      | 507,938          |
| Cash Flows from Investing Activities                           |                  |
| Purchases of property and equipment                            | (9,806)          |
| Net cash used in investing activities                          | (9,806)          |
| There eash asea in investing derivities                        | (3,000)          |
| Effect of Foreign Exchange Rate Changes on Accounts Receivable | 27,975           |
| Not Increase in Cash and Cash Equivalents                      | 526,107          |
| Net Increase in Cash and Cash Equivalents                      | 320,107          |
| Cash and Cash Equivalents - Beginning of Period                | _                |
| Cash and Cash Equivalents - End of Period                      | \$<br>526,107    |
|                                                                |                  |
| Supplemental Disclosure of Noncash Financing Activities        |                  |
| Noncash capital contribution from Member                       | \$<br>19,696,904 |

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Nature and Scope of Business**

Mrs. Fields Franchising Holdco, LLC was formed and commenced operations on August 1, 2024 (date of formation) in the state of Delaware. Mrs. Fields Franchising Holdco, LLC and its wholly owned subsidiary Mrs. Fields Franchising, LLC (the "Company") develop and franchise retail stores that sell core products, including cookies and brownies, through the specialty branded concept Mrs. Fields. Mrs. Fields Franchising Holdco, LLC is a wholly owned subsidiary of Famous Brands, LLC ("Member"), which is a wholly owned subsidiary of Famous Brands International, LLC ("FBI"), which is a wholly owned subsidiary of Famous Brands International Parent, LLC ("FBIP").

On August 1, 2024, after the formation of the Company, FBIP contributed all assets, liabilities, and rights attributable to Mrs. Fields' franchising operations to FBI, who immediately contributed to Famous Brands, LLC, who immediately contributed to Mrs. Fields Franchising Holdco, LLC. These assets and liabilities included the related trade name, franchise agreements, and goodwill associated with the brand derived from FBIP acquiring these assets on October 19, 2023. This formation and contribution aimed to separate the operations of this brand from a related brand, which was previously consolidated into one entity, to better manage and grow the acquired businesses. The assets and liabilities contributed on August 1, 2024 were based on the historical value of the related party contributing and are as follows:

| \$<br>397,954    |
|------------------|
| 24,319           |
| 869,777          |
| 50,728           |
| 16,363,860       |
| 2,641,378        |
| 26,810           |
| (213,781)        |
| (5,906)          |
| (138,987)        |
| (36,120)         |
| (283,128)        |
|                  |
| \$<br>19,696,904 |
|                  |

The Company authorizes franchisees to use certain business formats, systems, methods, procedures, designs, layouts, specifications, and trade name in the United States of America and other countries. The Company licenses the use of its trade name, logos, and recipes to third parties for the distribution of Mrs. Fields' branded products through specialty stores and cafes. There were 191 and 189 franchised locations operating as of December 31, 2024 and the contribution date, August 1, 2024 respectively, inclusive of both domestic and international locations.

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies**

#### Basis of Presentation

The accompanying consolidated financial statements have been prepared on an accrual basis in accordance with accounting principles generally accepted in the United States of America ("GAAP").

The Company operates on a calendar year basis; however, the consolidated financial statements present the period from August 1, 2024 (the date of formation) to December 31, 2024.

#### Principles of Consolidation

The consolidated financial statements include the accounts of Mrs. Fields Franchising Holdco, LLC and its wholly owned subsidiary, Mrs. Fields Franchising, LLC. All inter-company accounts and transactions have been eliminated upon consolidation.

#### Variable Interest Entities

The Company has evaluated its business relationships with related party entities to determine whether any are considered variable interest entities in accordance with the guidance under ASC 810 - Consolidation. Based on this evaluation, the Company has determined that consolidation of these entities is not required for the period presented.

#### Use of Estimates

The preparation of consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses and related disclosures of contingent liabilities in the consolidated financial statements and accompanying notes. From time to time, management evaluates these estimates, including those that relate to the fair value of goodwill and intangible assets, long-lived asset impairments, gift card liabilities, contingencies, and the practical expedient estimate for revenue recognition. The Company bases its estimates on historical experience and on various assumptions that it believes are reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities. Actual results could differ from those estimates, making it possible that a change could occur in the near term.

#### Foreign Currency Translation

A related party administers the assets and liabilities of certain international franchise operations. They are included in due from related party on the consolidated balance sheet. The net amount receivable from the related party is translated into U.S. dollars using period-end exchange rates. Adjustments resulting from these financial statement translations are included within accumulated other comprehensive income within the accompanying consolidated financial statements. Related revenues and expenses are translated at average rates prevailing during the period annually.

See Independent Auditor's Report.

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies** (continued)

#### Cash and Cash Equivalents

The Company considers all cash and other liquid investments with maturities of three months or less to be cash equivalents.

At times, the Company's bank account balances may exceed federally insured limits. Management monitors the financial condition of the institutions with which it invests; nothing has been identified to suggest any of the institution's financial condition were at risk. The Company has not experienced any losses in such accounts.

#### **Accounts Receivable**

The Company's receivables are primarily due from domestic and international franchisees. The Company follows Accounting Standards Codification ("ASC") Topic 326, *Financial Instruments – Credit Losses* (ASC 326) which requires the Company to measure all expected credit losses for financial assets (or a group of financial assets) held at the reporting date based on historical experience, current conditions, and reasonable supportable forecasts. The carrying amount of receivables is reduced by a valuation allowance for expected credit losses, as necessary, that reflects management's best estimate of the amount that will not be collected. This estimation takes into consideration historical experience, current conditions and, as applicable, reasonable supportable forecasts. Actual results could vary from the estimate. Accounts are charged against the allowance when management deems them to be uncollectible. Based on its assessment, management determined that the risk of credit loss was not material; valuation allowance recorded as of December 31, 2024 is considered de minimis. The Company does not accrue finance or interest charges on outstanding balances.

#### Property and Equipment

Property and equipment are recorded at cost. Property and equipment include computers of \$29,307, software of \$38,906, and accumulated depreciation of \$10,864, and a net property and equipment balance of \$57,349. Computers and software are depreciated over a useful life of 3 years, and is computed using the straight-line method over the estimated useful lives of the assets. Depreciation expense was \$3,184 for the period ended December 31, 2024.

Expenditures that materially extend useful lives of property and equipment are capitalized. Routine maintenance, repairs and renewal costs are expensed as incurred. Gains or losses from the sale or retirement of property and equipment are recorded in current operations.

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies** (continued)

#### Goodwill

Goodwill represents the excess of the consideration transferred above the fair value of the assets acquired and liabilities assumed related to acquisitions. See *Nature and Scope of Business*.

Under Accounting Standards Update (ASU) 2014-02, *Intangibles – Goodwill and Other (Topic 350) Accounting for Goodwill*, the Company has elected to amortize goodwill over a period of 10 years. In addition, the Company has elected to test for goodwill impairment at the Company level when a triggering event occurs. An impairment loss is recognized to the extent the carrying value of the entity exceeds its fair value. During the period ended December 31, 2024, no triggering events were identified. The Company determined that no impairment was required for the period ended December 31, 2024.

#### **Intangible Assets**

The trade name is an indefinite lived asset recorded at fair value when originally acquired and reflects the distinctive signs, symbols, or words that identify and distinguish products or services of a company from those of others. The trade name represents valuable intellectual property rights that can contribute significantly to a company's brand recognition, reputation, and competitive advantage.

Acquired franchise agreements are amortized on a straight-line basis using 20 years or term remaining. Acquired franchise agreements relate to franchise rights acquired for future development and are recognized at fair value when originally acquired. Fair value reflects the present value of expected future cash flows associated with the franchise agreements, considering factors such as the brand strength, market conditions, and contractual terms.

#### Impairment or Disposal of Long-Lived Assets

Long-lived assets, such as property and equipment and intangible assets subject to amortization are reviewed for impairment when events or changes in circumstances indicate that the book value of an asset may not be fully recovered. If events or circumstances indicate that the carrying amount of a long-lived asset may be impaired, the Company will make an assessment of recoverability using an estimate of future undiscounted net cash flows of the related asset or group of assets over the remaining life in measuring whether the assets are recoverable. Impairment of long-lived assets is assessed at the lowest level for which there are identifiable cash flows that are independent of other groups of assets. Impairment to be recognized is measured by the amount by which the carrying amount of the assets exceeds the fair value of the assets. The impairment of long-lived assets requires judgments and estimates. If circumstances change, estimates could also change. The Company determined no impairments had occurred during the period ended December 31, 2024.

The Company's trade name is not subject to amortization but is tested for impairment annually or whenever events or changes in circumstances indicate that the asset might be impaired. The Company determined no impairments had occurred as of December 31, 2024.

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies** (continued)

#### Revenue Recognition

Revenue consists of franchise related fees including initial franchise fees and transfer fees, royalty fees, marketing and advertising fees, and other income. The Company recognizes revenue when it satisfies a performance obligation by transferring control over a product or service to a franchisee.

Franchise related fees, deferred revenue, and deferred expenses

The performance obligations under the franchise agreements consist of a franchise license, including a license to use the Mrs. Fields brand and Marketing and Advertising Fund ("MAF") management; pre-opening services, such as guidance in selecting sites, assistance in layout and signage for store training and inspections; and ongoing services, such as development of training materials and store location monitoring and inspections.

Deferred revenue includes the initial franchisee fee paid upon execution of the franchise agreement and the renewal fee paid at the expiration of the initial term of the franchise agreement. Fees paid are typically nonrefundable, and in exchange, the Company provides a variety of services outlined above.

Deferred revenue obtained prior to October 19, 2023, is accounted for under ASC 606 as a single performance obligation, which is satisfied over time by providing a right to use the intellectual property over the term of each franchise agreement. Initial and renewal franchise fees and area development fees are recognized as revenue over the term of the respective agreement unless the franchise agreement is terminated early, in which case the remaining initial or renewal franchise fee is fully recognized in the period of termination. Franchise fees paid under Master Franchise Agreements are amortized over the remaining life of the Master Franchise Agreement. The pro rata amount apportioned to each location is accounted for as an initial franchise fee. The accounting treatment of revenue recognition follows how the deferred revenue was accounted for under prior ownership.

The initial franchise fees are included in deferred revenues upon receipt as they are not considered distinct performance obligations under ASC 606. Revenue from the initial franchise fees is recognized at the beginning of the performance obligation and continues throughout the life of the franchise agreements. The Company typically grants franchise rights to franchisees for a term of 10 years. The Company considers the franchise location opening to be when the performance obligation commences.

For franchise agreements entered into after October 19, 2023 (see *Nature and Scope of Business*), the Company has elected to apply the practical expedient under (ASC) *Revenue from Contracts with Customers* ("Topic 606"), ASU 2021-02 *Franchisors – Revenue from Contracts with Customers* (*Subtopic 952-606*) and has divided its performance obligations into two components: upfront obligations (satisfied at a point in time) and obligations under the ongoing franchise agreement (satisfied over a period of time).

Notes to the Consolidated Financial Statements
December 31, 2024

**Summary of Significant Accounting Policies** (continued)

Revenue Recognition (continued)

Franchise related fees, deferred revenue, and deferred expenses (continued)

The Company has identified its pre-opening services as a separate and distinct element of its contract satisfied at a point in time because, upon completion, the franchisee has full knowledge of the Company's proprietary methods. Further, most of the Company's direct costs are associated with the recruiting and training of franchisees. The Company calculated upfront revenue in reference to the total transaction price over the term of the initial franchise agreement and an allocation to the specific performance obligations based on their relative stand-alone values. Based on this calculation, the Company recognizes upfront revenue of 80% of the franchise fee. The contract liability resulting from the income deferral will be amortized on a straight-line basis over the remaining term of the franchise agreement.

In the event a franchise agreement is terminated, any remaining deferred revenue is recognized in the period of termination.

The Company has agreements with certain employees to facilitate the sale of franchises. The terms of these agreements require that the Company pay commissions to these employees for each franchise sold. Under ASC 606, these commissions are recorded as deferred expenses and recognized beginning when the performance obligation commences through the life of the franchise agreement, which is the same term over which the revenue is recognized.

The Company has recorded both the current and long-term portions of deferred expenses and deferred revenue. The current portions consist of the Company's commission expense and revenue estimate, which is expected to be recognized within one year of the balance sheet date based on the stores that are expected to commence operations.

A summary of significant changes to the deferred revenue and deferred expenses balances are shown below:

|                                                                                          | Defe | rred Revenue | Deferr | red Expenses |
|------------------------------------------------------------------------------------------|------|--------------|--------|--------------|
| Balance, August 1, 2024 (date of formation)                                              | \$   | 283,128      | \$     | 26,810       |
| Revenue recognition from initial franchise fees<br>Expense recognition from direct costs |      | (32,239)     |        | -<br>(6,180) |
| Balance, December 31, 2024                                                               | \$   | 250,889      | \$     | 20,630       |

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies** (continued)

#### Revenue Recognition (continued)

Franchise related fees, deferred revenue, and deferred expenses (continued)

The following table illustrates the estimated revenue recognition from franchise stores as of December 31, 2024 to be recognized in the future related to performance obligations that are unsatisfied as of December 31, 2024:

| 2025       | \$ | 41,349  |
|------------|----|---------|
| 2026       | ·  | 37,801  |
| 2027       |    | 36,859  |
| 2028       |    | 36,443  |
| 2029       |    | 36,359  |
| Thereafter |    | 62,078  |
|            |    |         |
|            | \$ | 250,889 |
|            | \$ | 62,078  |

Future store openings are subject to adjusted opening schedules and have not been included in the above table as a result.

#### Royalty Fees

Royalty fees are calculated as a percentage of franchisee sales. Royalty fees represent salesbased payments that are related entirely to the performance obligation under the franchise agreement and are recognized as franchisee sales occur.

#### Marketing and Advertising Fund

MAF contributions are calculated as a percentage of franchisee sales. MAF contributions paid by franchisees are used for the creation and development of brand advertising, marketing and public relations, merchandising research and related programs, activities and materials. Under Topic 606, MAF contributions will be reflected as a separate line in the revenue section and the related expenses will be reflected as a separate line in the operating costs and expenses on the consolidated statement of comprehensive income. MAF contributions represent sales-based payments that are related entirely to the performance obligation under the franchise agreement and are recognized as franchisee sales occur.

#### Other Income

Other income includes gift card breakage and revenue received by the Company as a result of franchisee purchases of products and services from vendors that supply food and supplies to franchisees, and are recognized as revenue as the thresholds of product purchases by the Company's franchisees are met.

Notes to the Consolidated Financial Statements
December 31, 2024

#### **Summary of Significant Accounting Policies** (continued)

#### Gift Cards

The Company sells gift cards primarily through certain third-party distributors. These gift cards do not expire or incur a service fee on unused balances. The Company acts as a clearinghouse for its franchisees' gift card transactions. On a monthly basis, franchisees are billed or paid for the gift card sales over or under gift card redemptions at the respective franchisees' store for the month.

Sales of gift cards are initially recorded as a liability at their expected redemption value. The portion of gift cards sold to customers that are never redeemed is recorded as gift card breakage in accordance with Topic 606 or escheated to the state in accordance with laws and guidelines under which the Company is required to adhere to. Under Topic 606, gift card breakage revenue is recognized in proportion to the pattern of gift card redemptions exercised by customers, using an estimated breakage rate based on historical experience. During the period from August 1, 2024 (date of formation) to December 31, 2024, the Company recognized \$739 of breakage income within other income on the accompanying consolidated statement of comprehensive income.

#### Advertising Expenses

The Company expenses all advertising costs as they are incurred. Total advertising costs for the period ended December 31, 2024 were \$52,171, and are included in marketing and advertising expense and general and administrative expense in the accompanying consolidated statement of comprehensive income.

#### Income Taxes

The Company, with the consent of its Member, has elected to be formed as a limited liability company. The Company has also elected to be treated as a partnership for federal and state income tax purposes. In lieu of paying federal and most state taxes at the company level, the Member of a limited liability company is taxed on the Company's taxable income. No provision or liability for federal or state income taxes has been included in the consolidated financial statements.

As of December 31, 2024, the Company had no uncertain tax positions that qualify for either recognition or disclosure in the consolidated financial statements. Additionally, the Company did not incur any interest and penalties related to income taxes.

Notes to the Consolidated Financial Statements
December 31, 2024

Goodwill

Goodwill as of December 31, 2024 is summarized as follows:

|                                   | Weighted<br>Average Life<br>in Years |   |                        |
|-----------------------------------|--------------------------------------|---|------------------------|
| Goodwill Accumulated amortization | 10                                   |   | 2,866,373<br>(344,427) |
| Net carrying value                |                                      | 9 | \$ 2,521,946           |

Amortization expense for the period ended December 31, 2024 was \$119,432.

Amortization expense for the next five years and thereafter is as follows:

| 2025       | \$<br>286,637   |
|------------|-----------------|
| 2026       | 286,637         |
| 2027       | 286,637         |
| 2028       | 286,637         |
| 2029       | 286,637         |
| Thereafter | 1,088,761       |
|            |                 |
| Total      | \$<br>2,521,946 |
|            |                 |

#### **Intangible Assets**

Intangible assets as of December 31, 2024 are summarized as follows:

|                                    | Amortizable<br>Life (Years) | Gross<br>Carrying<br>Value | Accumulated<br>Amortization | Net Carrying<br>Value      |
|------------------------------------|-----------------------------|----------------------------|-----------------------------|----------------------------|
| Trade name<br>Franchise agreements | Indefinite<br>20            | \$ 14,500,000<br>1,940,000 | \$ -<br>116,556             | \$ 14,500,000<br>1,823,444 |
|                                    |                             | \$ 16,440,000              | \$ 116,556                  | \$ 16,323,444              |

Amortization expense for the period ended December 31, 2024 was \$40,417.

Notes to the Consolidated Financial Statements
December 31, 2024

**Intangible Assets** (continued)

Amortization expense for the next five years and thereafter is as follows:

| 2025<br>2026<br>2027<br>2028 | \$<br>97,000<br>97,000<br>97,000<br>97,000 |
|------------------------------|--------------------------------------------|
| 2029<br>Thereafter           | 97,000<br>1,338,444                        |
|                              | \$<br>1,823,444                            |
|                              | <br>                                       |

#### **Member's Equity**

The Member is the entity's sole member and manages the Company's business and affairs. Contributions and distributions shall be made at the Member's discretion and at such time as the Member determines. See *Nature and Scope of Business* note.

#### **Commitments and Contingencies**

From time to time the Company is subject to lawsuits and other charges from customers and employees, which are typical within the industry. In the opinion of management, any open matters will not have a material effect upon the financial position of the Company. The Company incurred expenses of \$1,228,761 related to shared expenses paid by the Member during the period ended December 31, 2024. These expenses are included as a component of general and administrative expense and marketing and advertising expense on the accompanying statement of operation.

#### **Related Party Activity**

On December 31, 2024, the Company was due \$725,334 from FBI related to funds collected on behalf of the Company and in custody of FBI. These funds are included in due from related party on the accompanying consolidated balance sheet and are expected to be repaid within one year.

#### **Subsequent Events**

Management has evaluated subsequent events through the date of the Independent Auditor's Report, the date on which the consolidated financial statements were available to be issued.

Subsequent to the period end, five franchise locations closed.

#### **GUARANTEE OF PERFORMANCE**

For value received, Mrs. Fields Franchising Holdco, LLC, a Delaware limited liability company (the "Guarantor") located at 1717 S. 4800 W., Salt Lake City, Utah 84104 absolutely and unconditionally guarantees to assume the duties and obligations of Mrs. Fields Franchising, LLC, a Delaware limited liability company, 1717 S. 4800 W., Salt Lake City, Utah 84104 (the "Franchisor") under its franchise registration in each state where the franchise is registered and under its Franchise Agreement identified in its 2024 Franchise Disclosure Document, as it may be amended, and as that Franchise Agreement may be entered into with franchisees and amended, modified or extended from time to time. This guarantee continues until all such obligations of the Franchisor under its franchise registrations and the Franchise Agreement are satisfied or until the liability of Franchisor to its franchisees under the Franchise Agreement has been completely discharged, whichever first occurs. The Guarantor is not discharged from liability if a claim by a franchisee against the Franchisor remains outstanding. Notice of acceptance is waived. The Guarantor does not waive receipt of notice of default on the part of the Franchisor. This guarantee is binding on the Guarantor and its successors and assigns.

The Guarantor signed this Guarantee of Performance at working, Florida, on the 3° day of poil, 2025.

Guarantor:

MRS. FIELDS FRANCHISING HOLDCO, LLC

By: James Carnrite
Its: Interim CEO

# EXHIBIT B

## **FRANCHISE AGREEMENT**

#### FRANCHISE AGREEMENT

#### **BETWEEN**

MRS. FIELDS FRANCHISING, LLC 1717 S. 4800 W. Salt Lake City, Utah 84104 (720) 599-3350

|    |           | AND        |            |
|----|-----------|------------|------------|
|    |           |            |            |
|    |           |            |            |
|    | Name      | (s) of Fra | nchisee    |
|    |           | Street     |            |
|    | City      | State      | ZIP Code   |
| )_ | Area Code |            | Telephone  |
|    | Franchise |            | formation: |
|    |           | Street     |            |
|    | City      | State      | ZIP Code   |
| )_ | Area Code |            | Telephone  |
|    | Date of F | ranchise A | Agreement  |
|    |           |            | 20         |

## FRANCHISE AGREEMENT

## TABLE OF CONTENTS

| 1.     | DEFI                      | NITIONS                                                      | 1  |  |  |  |
|--------|---------------------------|--------------------------------------------------------------|----|--|--|--|
| 2. GRA |                           | NT OF FRANCHISE                                              | 3  |  |  |  |
|        | 2.1.                      | Grant of Franchise                                           | 3  |  |  |  |
|        | 2.2.                      | No Exclusivity                                               | 3  |  |  |  |
|        | 2.3.                      | Approved Products                                            | 4  |  |  |  |
| 3.     | SUCC                      | ESSOR FRANCHISE                                              | 4  |  |  |  |
|        | 3.1.                      | Renewal                                                      | 4  |  |  |  |
| 4.     | SITE                      | SELECTION, LEASE OF PREMISES AND DEVELOPMENT OF YOUR STORE   | 5  |  |  |  |
|        | 4.1.                      | Site Selection                                               | 5  |  |  |  |
|        | 4.2.                      | Your Obligation to Secure Possession                         |    |  |  |  |
|        | 4.3.                      | Development of Your Store                                    |    |  |  |  |
|        | 4.4.                      | Fixtures, Furnishings, Equipment, Signs and Computer Systems |    |  |  |  |
|        | 4.5.                      | Opening Your Store                                           |    |  |  |  |
|        | 4.5.<br>4.6.              | Grand Opening Promotion                                      |    |  |  |  |
| 5.     | TRAI                      | TRAINING AND GUIDANCE                                        |    |  |  |  |
|        | 5.1.                      | Operations Manual                                            | 7  |  |  |  |
|        | 5.2.                      | Training                                                     |    |  |  |  |
|        | 5.3.                      | Other Guidance and Operating Assistance                      |    |  |  |  |
| 6.     | FEES                      |                                                              | 8  |  |  |  |
|        | 6.1.                      | Initial Franchise Fee                                        | 8  |  |  |  |
|        | 6.2.                      | Continuing Royalty Fees                                      |    |  |  |  |
|        | 6.3.                      | Technology Fee                                               |    |  |  |  |
|        | 6.4.                      | Method and Application of Payments                           |    |  |  |  |
|        | 6.5.                      | Late Fees; Interest on Late Payments                         |    |  |  |  |
|        | 6.6.                      | No Right of Offset                                           |    |  |  |  |
| 7.     | OBLI                      | GATIONS RELATING TO OPERATIONS                               | 9  |  |  |  |
|        | 7.1.                      | Compliance with Laws and Good Business Practices             | 9  |  |  |  |
|        | 7.2.                      | Compliance with System Standards                             |    |  |  |  |
|        | 7.3.                      | Restrictions on Operations                                   |    |  |  |  |
|        | 7.4.                      | Internet Use                                                 |    |  |  |  |
|        | 7. <del>-</del> .<br>7.5. | Our Right to Inspect Your Store                              |    |  |  |  |
|        | 7.5.<br>7.6.              | Surveys                                                      |    |  |  |  |
|        | 7.0.<br>7.7.              | Good Standing and Guaranties by Entity Owners                |    |  |  |  |
|        |                           | • •                                                          |    |  |  |  |
|        | 7.8.                      | Insurance                                                    |    |  |  |  |
|        | 7.9.                      | Pricing                                                      |    |  |  |  |
|        | 7.10.                     | Information Security                                         |    |  |  |  |
|        | 7.11.                     | Employees, Agents and Independent Contractors                | 12 |  |  |  |

| 8.  | REPO                     | REPORTS AND RECORD KEEPING                         |      |  |  |  |
|-----|--------------------------|----------------------------------------------------|------|--|--|--|
|     | 8.1.                     | Accounting, Reports and Financial Statements       | . 12 |  |  |  |
|     | 8.2.                     | Our Right to Audit                                 |      |  |  |  |
| 9.  | MAR                      | KETING AND PROMOTION                               | . 14 |  |  |  |
|     | 9.1.                     | Brand Fund                                         | . 14 |  |  |  |
|     | 9.2.                     | Your Advertising and Promotional Activities        | . 15 |  |  |  |
|     | 9.3.                     | Our Advertising Materials                          |      |  |  |  |
|     | 9.4.                     | Advertising Cooperatives                           |      |  |  |  |
|     | 9.5.                     | Participation in Certain Programs and Promotions   |      |  |  |  |
| 10. | USE (                    | USE OF THE MARKS AND CONFIDENTIAL INFORMATION      |      |  |  |  |
|     | 10.1.                    | Ownership and Goodwill of Marks                    | . 16 |  |  |  |
|     | 10.2.                    | Limitations on Your Use of Marks                   | . 16 |  |  |  |
|     | 10.3.                    | Discontinuance of Use of Marks                     | . 16 |  |  |  |
|     | 10.4.                    | Infringements and Claims                           | . 16 |  |  |  |
|     | 10.5.                    | Non-Disparagement                                  | . 16 |  |  |  |
|     | 10.6.                    | Concepts Developed by You                          | . 17 |  |  |  |
|     | 10.7.                    | Confidential Information                           | . 17 |  |  |  |
| 11. | COVENANTS NOT TO COMPETE |                                                    |      |  |  |  |
|     | 11.1.                    | In-Term Non-Compete and Non-Solicitation           | . 17 |  |  |  |
|     | 11.2.                    | Post-Term Non-Compete and Non-Solicitation         | . 18 |  |  |  |
|     | 11.3.                    | Validity and Application of Restrictions           | . 18 |  |  |  |
|     | 11.4.                    | Enforcement of Non-Competes                        | . 18 |  |  |  |
| 12. | TRAN                     | ISFERS                                             | . 18 |  |  |  |
|     | 12.1.                    | Transfers by Us                                    |      |  |  |  |
|     | 12.2.                    | Restrictions on Transfers by You                   |      |  |  |  |
|     | 12.3.                    | Conditions for Our Approval of a Proposed Transfer | . 18 |  |  |  |
|     | 12.4.                    | Transfer to a Wholly Owned Entity                  |      |  |  |  |
|     | 12.5.                    | Our Right of First Refusal                         | . 20 |  |  |  |
|     | 12.6.                    | Death or Permanent Disability                      | . 21 |  |  |  |
|     | 12.7.                    | Effect of Consent to Transfer                      |      |  |  |  |
|     | 12.8.                    | Preparation of a Financial Report by You           | . 21 |  |  |  |
| 13. | DEFA                     | ULT AND TERMINATION                                | . 21 |  |  |  |
|     | 13.1.                    | Your Defaults                                      |      |  |  |  |
|     | 13.2.                    | Other Rights and Remedies                          |      |  |  |  |
|     | 13.3.                    | Assumption of Management                           | . 23 |  |  |  |
| 14. | POST                     | -TERM OBLIGATIONS                                  | . 23 |  |  |  |
|     | 14.1.                    | Your Obligations                                   |      |  |  |  |
|     | 14.2.                    | Our Option to Purchase Your Store                  |      |  |  |  |
|     | 14.3.                    | Lost Revenue Damages                               |      |  |  |  |
|     | 14.4.                    | Continuing Obligations                             | . 26 |  |  |  |

| 15. | RELATIONSHIP OF THE PARTIES/INDEMNIFICATION |                                              |    |
|-----|---------------------------------------------|----------------------------------------------|----|
|     | 15.1.                                       | Independent Contractors                      | 26 |
|     | 15.2.                                       | Our Approval and Enforcement                 | 26 |
|     | 15.3.                                       | Taxes                                        |    |
|     | 15.4.                                       | Indemnification                              |    |
|     | 15.5.                                       | Waiver of Claims                             |    |
| 16. | SECU                                        | JRITY AGREEMENT                              | 27 |
|     | 16.1.                                       | Security Interest                            | 27 |
| 17. | DISPU                                       | UTE RESOLUTION                               | 28 |
|     |                                             |                                              |    |
|     | 17.1.                                       | Injunctive Relief                            | 28 |
|     | 17.2.                                       | Rights of Parties Are Cumulative             |    |
|     | 17.3.                                       | Costs and Attorneys' Fees                    | 28 |
|     | 17.4.                                       | Governing Law                                | 28 |
|     | 17.5.                                       | Consent to Jurisdiction                      | 28 |
|     | 17.6.                                       | Waiver of Punitive Damages and Jury Trial    | 28 |
|     | 17.7.                                       | Limitation of Claims; Waiver of Class Action |    |
|     | 17.8.                                       | Arbitration                                  | 29 |
| 18. | GENE                                        | ERAL PROVISIONS                              | 30 |
|     | 18.1.                                       | Severability                                 | 30 |
|     | 18.2.                                       | Rights Provided by Law                       | 30 |
|     | 18.3.                                       | Waivers by Either of Us                      | 30 |
|     | 18.4.                                       | Certain Acts Not to Constitute Waivers       | 31 |
|     | 18.5.                                       | Interpretation of Rights and Obligations     | 31 |
|     | 18.6.                                       | Notice of Our Potential Profit               | 31 |
|     | 18.7.                                       | Binding Effect                               | 32 |
|     | 18.8.                                       | Execution                                    | 32 |
|     | 18.9.                                       | Notices and Payments                         | 32 |

# **SCHEDULES**

Schedule 1 – Ownership Addendum

Schedule 2 – Guaranty and Assumption of Obligations

Schedule 3 – Store Premises; Start Date

Alternative Schedule 3 – Store Premises; Start Date

Schedule 4 – Lease Addendum

#### FRANCHISE AGREEMENT

# MRS. FIELDS®

| THIS FRANCHISE AGREEMENT (the "Agreement") is                                        | made as of the Effective Date between |  |  |
|--------------------------------------------------------------------------------------|---------------------------------------|--|--|
| Mrs. Fields Franchising, LLC, a Delaware limited liability compar                    | ny, whose address is 1717 S. 4800 W.  |  |  |
| Salt Lake City, Utah 84104 ("us"), and                                               | , a                                   |  |  |
| whose address is                                                                     | (" <b>you</b> "). The Effective       |  |  |
| Date is the date on which we sign this Agreement as indicated beneath our signature. |                                       |  |  |

#### **PREAMBLES**

We and our Affiliates have expended a considerable amount of time and effort in developing, refining and protecting the Marks, the distinctive business formats and System Standards under which Stores are developed and operate, and the Approved Products they offer and sell (including the recipes for and the methods of preparation of Approved Products to obtain high product quality). We and our Affiliates have franchised and licensed and, in the future, have the right to continue to franchise and license others to operate Stores.

You and, if applicable, your Owners have requested that we grant you a Franchise to develop, own and operate a Store as described in this Agreement. In support of your request, you and, if applicable, your Owners, have provided us with certain information and documentation. In reliance on that information and documentation and on your and, if applicable, your Owners' representations, we are willing to grant you a Franchise on the terms and conditions contained in this Agreement.

#### 1. **DEFINITIONS**

Certain terms may be defined where they appear within the text of the Agreement. The following terms have the meanings below:

- (a) "Affiliate," means any person or entity that directly or indirectly owns or controls, is directly or indirectly owned or controlled by, or is under common control with a person or entity, now or in the future.
- (b) "Approved Products," means products we approve or require from time to time for sale at or from Stores, including specialty snacks and other bakery items, desserts, and beverages such as cookies, brownies, cakes, muffins, bagels, croissants, cinnamon rolls, sticky buns, chocolate, dried fruit, nuts, candies, popcorns, and coffee.
- (c) "Competitive Business" means any business that operates, grants franchises or licenses to others to operate, acts as a landlord to, or otherwise provides products, services or assistance to a restaurant, retail outlet or other food service business (other than a Store operated pursuant to a franchise agreement with us or our Affiliate) which offers or sells cookies, bakery items, snack foods or desserts as primary or featured items or for which the sale of cookies, bakery items, snack foods or desserts accounts (either individually or in the aggregate) accounts for at least 10% of its sales during any calendar month.
- (d) "Confidential Information" means any information relating to the Approved Products or the development or operation of Stores, including site selection criteria; trade secrets; recipes and methods for the preparation of Approved Products; methods, techniques, formats, standards, specifications, systems, procedures, sales and marketing techniques and knowledge of and experience in the development and operation of Stores; marketing programs for Stores; knowledge of specifications for and suppliers of certain Approved Products, materials, supplies, equipment,

furnishings and fixtures; customer data; and knowledge of the operating results and financial performance of Stores, including Your Store.

- (e) "Controlling Interest" means an interest, the ownership of which empowers the holder to exercise a material influence over the management, policies or personnel of an Entity. Ownership of five percent (5%) or more of the equity or voting securities of a corporation, limited liability company or limited liability partnership or ownership of any general partnership interest in a general or limited partnership will be deemed conclusively to constitute a Controlling Interest in the corporation, limited liability company, or partnership, as the case may be.
- (f) "Entity" means a corporation, general partnership, joint venture, limited partnership, limited liability partnership, limited liability company, trust, estate or other business entity.
- (g) "Entity Owner" means, with respect to an Entity, any shareholder owning directly or beneficially five percent (5%) or more of any class of securities of the Entity; any general partner or co-venturer in the Entity; any partner in a limited liability partnership or member in a limited liability company owning directly or beneficially 5% or more of the ownership interests in the limited liability partnership or limited liability company; the trustees or administrators of any trust or estate; and any beneficiary of a trust or estate owning, directly or beneficially, five percent (5%) or more of the interests in the trust or estate. If any Entity Owner within the scope of this definition is itself an Entity (including an Entity Owner that is an Entity Owner because of this sentence), the term "Entity Owner" also includes Entity Owners (as defined in the preceding sentence) in the Entity. It is the intent of this definition to "trace back" and include within the definition of Entity Owner all natural persons owning the requisite interests to qualify as Entity Owners.
- (h) "**Franchise**" means the non-exclusive, non-transferable (except as provided herein, and non-sublicensable right and license, as described in this Agreement, to develop, own and operate a Store as described in this Agreement.
- (i) "Gross Revenue" means the aggregate of all revenue, in whatever form (including the value of all barter, exchange, trade or other credit), attributed to the operation of Your Store (whether or not in compliance with this Agreement) based on the gross price of all products and services sold at or through Your Store and all other revenue sources, including fees for any delivery, catering and other approved off-site activities, but excluding all approved customer discounts and all federal, state or municipal sales, use, or service taxes collected from customers and paid to the appropriate taxing authority. Gift certificate, gift card and similar program payments will be included in Gross Revenue consistent with the policies periodically stated in the Operations Manual. Gross Revenue also includes all insurance proceeds you receive for loss of business due to a casualty or similar event.
- (j) "Marks" means trademarks, trade names, service marks, logos and other commercial symbols that we authorize franchisees from time to time to use to identify their Stores and the Approved Products, including the trademarks and service marks MRS. FIELDS® and MRS. FIELDS COOKIES®, and the associated Trade Dress.
- (k) "Operations Manual" means the manual or series of manuals, memos, newsletters and other written communications that we provide or make available to you for your use solely in developing and operating Your Store. The Operations Manual contain both mandatory and suggested specifications, standards and operating procedures, including System Standards, regarding various aspects of developing and operating Stores and may be modified by us from time to time to reflect changes in the image, specifications, standards, procedures, Approved Products, System, and System Standards.

- (l) "**Premises**" means the physical location at which you are authorized to develop, own and operate Your Store under this Agreement.
- (m) "Restricted Person" means you; each of your Entity Owners, if you are an Entity; and the spouses, natural and adopted children, and siblings of any of you and your Entity Owners.
- (n) "Store" means a retail snack, dessert, and beverage outlet selling any Approved Products for on- or off-premises consumption and other products and services specified by us. Stores include carts and kiosks, and other satellite units we approve to sell the Approved Products.
- (o) "**System**" means our business formats, signs, equipment, methods, procedures, designs, layouts, standards and specifications, including the use of the Marks and the Trade Dress, which we have the right to modify in the future.
- (p) "**System Standards**" means the operating procedures, standards, requirements and specifications, whether contained in the Operations Manual or elsewhere, which we have the right to improve, further develop or modify from time to time and which are mandatory in nature so as to comprise the requirements to be followed by Stores, and the use of the Marks in connection therewith.
- (q) "**Term**" means the period beginning on the Effective Date and ending on the 10<sup>th</sup> anniversary of that date.
- (r) "**Trade Dress**" means the designs, color schemes, decor and images which we authorize and require our franchisees to use in connection with the operation of a Store, which we or our Affiliates have the right to revise and further develop from time to time.
- (s) "**Transfer**" means the voluntary or involuntary, direct or indirect transfer, assignment, sublicense, sale, gift, pledge, mortgage, hypothecation or other disposition (including those occurring by operation of law and a series of transfers that in the aggregate constitute a Transfer) of any of your interests in this Agreement, Your Store or its assets (other than in the ordinary course of business), your right to possession of the Premises, or a direct or indirect ownership interest in you.
- (t) "Your Store" means the Store that you are authorized to develop, own and operate pursuant to this Agreement.

# 2. **GRANT OF FRANCHISE**

- 2.1. Grant of Franchise. We grant you the Franchise for the Term and subject to the terms and conditions of this Agreement. The Franchise is specific to the Premises identified on Schedule 3; however, if at the time of signing this Agreement we have not accepted and you have not secured the Premises for Your Store, you will have six (6) months from the Effective Date to obtain our acceptance of and secure the Premises for Your Store. If you subsequently obtain our acceptance of and secure the Premises for Your Store within the 6-month period, we and you will sign Alternative Schedule 3 **identifying** the Premises for Your Store. If you fail to obtain our acceptance of and secure the Premises for Your Store within the 6-month period, however, we have the right to terminate this Agreement. For purposes of this Agreement, to "secure the Premises for Your Store" means that you have either (i) signed a Lease (defined below) that we have accepted as described in Section 4.2 below, if you are leasing or subleasing the Premises; or (ii) taken possession of the Premises, if you own the Premises. You hereby accept the Franchise and undertake the obligation to operate Your Store using the System in accordance with the System Standards.
- 2.2. <u>No Exclusivity</u>. You are not granted an exclusive area or protected territory around the Premises within which we or our Affiliates agree not to issue franchises or operate competing businesses. We and our Affiliates have the right, anywhere in the world, to issue franchises, grant licenses for products

or services, and conduct any other business or activities, with or without the Marks and in any type of distribution channel, regardless whether such activities are competitive with Your Store. We and our Affiliates will not be required to provide you with any compensation for, or any right to participate, in any such of activities.

2.3. <u>Approved Products</u>. In operating Your Store, you must offer all Approved Products and may offer for sale only those Approved Products that we approve from time to time for you to sell at the Premises. The Approved Products that you initially are authorized to offer at Your Store are explained in the Operations Manual. In the future, we have the right to change or add to the Approved Products that you are authorized to offer at the Premises and notify you of such changes or additions, as we determine, through references to the Operations Manual, bulletins and other written materials, electronic computer messages, telephonic conversations, and/or consultations at our offices or at Your Store.

### 3. SUCCESSOR FRANCHISE

- 3.1. <u>Renewal</u>. You have the one-time option to acquire a successor franchise for an additional 10-year term subject to the following:
  - (a) You must give us written notice of your intention to acquire a successor franchise between 90 and 180 days prior to expiration of the Term;
  - (b) You must sign our then-current form of Franchise Agreement, the terms and conditions of which may be materially different than those contained in this Agreement, including with respect to fees and charges and performance criteria. The successor franchise agreement will be amended to acknowledge that Your Store is then in operation, to provide for payment of the fee described in clause (h) below instead of an initial franchise fee, and to remove any further rights to secure another successor franchise agreement;
  - (c) You, your Affiliates, and your and their respective officers, directors, owners, and agents must execute a general release of any and all claims against us, our Affiliates and our and their respective officers, directors, owners, agents and employees;
  - (d) At our request, you must refurbish, remodel, redecorate, and renovate Your Store at the commencement of the renewal term to meet our then-current System Standards for Stores, including designs and service systems, computer and point-of-sale equipment, and Trade Dress;
  - (e) You must have complied with all of the terms and conditions of this Agreement or any other agreement between you and us during the Term;
  - (f) All amounts you owe to us, our Affiliates or your suppliers or creditors, whether pursuant to this Agreement or otherwise, must have been paid in full prior to acquiring the successor franchise, and must have been paid in a timely manner throughout the Term;
  - (g) You must have the right to maintain the Premises for at least the duration of the term of the successor franchise, and you must provide a copy of the Lease to us;
  - (h) You must pay us a fee in the amount of twenty percent (20%) of the then-current initial franchise fee; and
  - (i) You must follow our then-current renewal process, which may require you to deliver certain financial statements and other records and reports to us, attend additional training and cooperate in any audits and/or inspections we may conduct or require.

If we determine that you have met all of the conditions described above prior to the expiration of the Term, we will provide you with an execution copy of the form of Franchise Agreement to be entered into for the successor franchise term. If you do not execute and return the successor Franchise Agreement to us within 30 days of your receipt, then you will be deemed to have withdrawn your notice of renewal, and this Agreement will terminate at the end of the Term.

# 4. SITE SELECTION, LEASE OF PREMISES AND DEVELOPMENT OF YOUR STORE

- 4.1. <u>Site Selection.</u> You must locate and obtain our written acceptance of the Premises before you sign a lease or sublease for or begin construction of the Premises. You must use our designated real estate broker to evaluate potential sites, unless we otherwise agree in writing. Our acceptance of the Premises is for our own internal purposes and will, to a large extent, be based on information you provide us with respect to the demographic and commercial characteristics of the proposed Premises. Our acceptance does not constitute a representation or warranty of any kind as to the suitability of the Premises for Your Store or for any other purpose. You must make your own determination in that regard. Our criteria may change, the commercial characteristics of the Premises may change, and there are no guarantees with respect to the continued suitability of the Premises or its potential for success. Those factors are beyond our control, and we will not be responsible to you if the Premises do not meet your expectations. We may assist you in identifying and securing possession of the Premises, including by referring you to brokers and identifying one or more sites that we believe to be available and that might meet our criteria. Our assistance does not replace your own diligence and investigation, and you may not rely on any assistance we provide for any purpose whatsoever.
- 4.2. <u>Your Obligation to Secure Possession</u>. Once we have confirmed, in writing, our acceptance of your proposed Premises, it will be your responsibility to secure the right to occupy the Premises under an agreement that allows you to develop and operate Your Store in accordance with this Agreement for the duration of the Term (for ease of reference, that agreement will be referred to as the "**Lease**"). However, you agree not to execute the Lease until we have accepted its terms. In addition, you and the landlord of the Store Premises must sign a "Lease Addendum" in the form attached hereto as Schedule 4.

Our acceptance of a Lease does not constitute an express or implied warranty by us of the successful operation or profitability of a Store operated at the Premises or that the terms of the Lease are as favorable to you as they might be. Our acceptance means only that we believe the Lease meets criteria we have established for our own internal purposes. If, after we accept your proposed Lease, you sign it, you do so based on your own diligence and acceptance of the terms and conditions contained in the Lease. You must provide us with a copy of the fully executed Lease within 15 days after its execution.

Development of Your Store. You are responsible, at your expense, for doing all things necessary to construct and develop Your Store in accordance with our System Standards, the Lease, and applicable law. We will furnish you with prototypical plans and specifications for a Store, including requirements for exterior and interior materials and finishes, dimensions, design, image, interior layout, decor, fixtures, equipment, signs, furnishings and color scheme. You must comply with these plans and specifications, as adapted, with our prior written approval, to accommodate the particular shape and dimensions of the Premises and to ensure that the plans and specifications comply with applicable ordinances, building codes and permit requirements and with lease requirements and restrictions. You agree to submit construction plans and specifications to us for our approval before construction of Your Store is commenced, and you agree to submit all revised plans and specifications to us for our approval during the course of construction. In constructing and outfitting Your Store, you may use only those materials and items (including furniture, fixtures and equipment) we designate, and you may secure those items from and use only those providers (including an architect and general contractor) that we designate or approve from time to time. We may require you to use our designated architect to assist you with site design and architectural plans for the Premises. On completion of construction, you also agree to provide us with a set of "as built" plans and specifications. You acknowledge and agree that you assume all risk relating to the construction and development of Your Store, and our designation or approval of your architect, contractor,

construction plans and specifications does not constitute an express or implied representation or warranty of any kind as to the quality of such construction or development or the success of Your Store.

- 4.4. <u>Fixtures, Furnishings, Equipment, Signs and Computer Systems</u>. In developing and operating Your Store, you agree to do each of the following:
  - (a) Use only the fixtures, furnishings, equipment and signs that we require and have approved for Stores as meeting our System Standards;
  - (b) Place or display at the Premises (interior and exterior) only the signs, emblems, lettering, logos and display materials that we approve in writing; provided, however, that we have the right to install all required signs at the Premises at your sole expense; and
  - Use the computer equipment and operating software, including any point-of-sale (c) or electronic cash register ("Computer System") that we specify from time to time. You must use our designated credit card processor. We have the right to require you to obtain specified computer hardware and/or software and modify specifications for and components of the Computer System from time to time from designated suppliers which may include us and our affiliates. Our modification of specifications for the Computer System's components may require you to incur costs to purchase, lease and/or license new or modified computer hardware and/or software, obtain service and support for the Computer System during the Term and pay any related connection fees. We have the right to require you in the future to purchase additional or different components of the Computer System, including computer hardware and software and connection and other related services, from a supplier or suppliers we designate, which may include us or our Affiliates. You agree to incur the costs of obtaining the computer hardware and software comprising the Computer System (or additions or modifications). There are no contractual limitations to the frequency and cost of the obligation to upgrade and maintain the Computer System We have the right to independently access the information and data you collect and gather using any Computer System or other data collection equipment (such as an electronic cash register) we require for Your Store. For defective equipment, products, software or other items purchased by you, you must deal directly with that manufacturer. We are not responsible for claims or damages incurred by you as a result of you using our Computer System. If you do not comply with the required Computer System that we specify, you will be assessed a Computer System Compliance Fee in the amount of the greater of \$500 per month, or the then current monthly rate of the software license. In addition, your failure to comply with the required Computer System that we specify shall constitute a default under Section 13.1(m) of this Agreement. Any deviation from our required Computer System is subject to our prior approval.

You agree that all fixtures, furnishings, equipment, signs and computer systems used in connection with the operation of Your Store will be free and clear of all liens, claims and encumbrances, except for liens, claims or encumbrances asserted by us and except for third-party purchase money security interests.

If you acquire an existing Store with an unapproved cash register system or that does not otherwise comply with our required Computer System and Brand Standards, we will require you or the selling franchisee to purchase and install the approved POS system as well as other approved Computer System components in the Store prior to transfer as a condition of our consent to the transfer. If you remodel an existing Store with an older cash register or POS system, we will require you to purchase and install our then-current POS system and approved Computer System components.

4.5. Opening Your Store. You will not open Your Store for business until we approve it for opening. We will not approve it for opening until all required pre-opening training has been completed to our satisfaction, all amounts then owed to us and our Affiliates have been paid, you have provided us with a copy of (and we have accepted) your fully executed Lease, you have provided us with a certificate of insurance issued by your insurance provider confirming that you have procured and fully paid for the insurance required under this Agreement.

Subject to your compliance with the conditions set forth in this Section 4.5, you agree to open Your Store for business no later than the "**Start Date**," which is the date that is 150 days after either: (i) the Effective Date, if at the time of signing this Agreement we have identified the Premises on Schedule 3 attached to this Agreement; or (ii) the date you and we sign Alternative Schedule 3 identifying the Premises for Your Store. Your Start Date is listed on Schedule 3 or Alternative Schedule 3, as applicable.

If you do not open your store by the Start Date, we have the right to (a) extend the Start Date, (b) charge you a \$1,000 per month Delinquent Opening fee, or (c) terminate your Agreement.

4.6. <u>Grand Opening Promotion</u>. You agree to conduct a grand opening advertising and promotion program for Your Store in a manner that we approve, commencing within 30 days either before or after the opening of Your Store, as we may reasonably designate. The grand opening program shall conform to our requirements and shall utilize the media and advertising formats we designate. You must submit a grand opening plan containing details about your planned grand opening promotion, and obtain our approval of the plan, at least sixty (60) days before your Start Date. You must spend for the grand opening program an amount we determine, which will be a minimum of \$10,000.

#### 5. TRAINING AND GUIDANCE

- 5.1. Operations Manual. To facilitate the training and guidance we provide and to assist you in developing and operating Your Store, we will, during the Term and provided you remain in compliance with this Agreement, provide you with access to a copy of our Operations Manual. You may not at any time copy any part of the Operations Manual, either physically or electronically. If there is a dispute relating to the contents of the Operations Manual, the master copy we maintain at our corporate headquarters, will control. The Operations Manual and any passwords that we provide to access it electronically form part of our Confidential Information and, as such, are subject to the same restrictions as other Confidential Information regarding its use, disclosure and return.
- 5.2. <u>Training</u>. We will provide training to you as described in this Section. You agree to comply with all of our training requirements as described below. We will charge an additional fee for attendance or participation in our training programs only as described below, but in every instance, you must arrange for and pay all of your and your other representatives' other costs of attending or participating in the initial training program and all other training programs we offer. We will determine the format, content, method of delivery and location of each training program. We currently offer much of the initial training program through our online Learning Management System ("LMS").
  - (a) <u>Initial Training</u>. We will provide you with access to our initial training program on the operation of Stores. Before opening Your Store for operation, you, at least one of your Entity Owners (if you are an Entity), and your initial store manager must successfully complete, to our satisfaction, our initial training program. You may not allow Your Store to be managed by a person who has not completed, to our satisfaction, our initial training program or who we have not otherwise certified to do so. We will not charge an additional fee for up to two (2) people to attend or participate in the initial training program.
  - (b) <u>Replacement Managers</u>. If you replace your initial store manager, the replacement manager will also be required to complete, to our satisfaction, all phases of our initial training program. We may charge a fee for the training for subsequent managers, which you will be required to pay at least ten (10) days prior to the beginning of training.
  - (c) <u>Refresher and Additional Training</u>. We have the right to require you and/or your previously trained and experienced representatives to attend periodic refresher training programs and training programs or courses regarding specific products or processes. We may charge fees for these additional or refresher training courses. Currently \$500 per day per person plus travel expenses (up to \$750 per day per person).

- (d) <u>National Conventions and Regional Meetings</u>. You or at least one of your Entity Owners (if you are an Entity) and, when we request, Your Store manager and/or an approved trainer if you are a multi-unit franchisee must attend all national conventions and regional meetings that we designate as mandatory. National conventions and regional meetings will take place at the time and locations we designate, and we may charge you a reasonable fee for each of your attendees (up to \$1,000 per attendee). We may also charge you a reasonable fee if you or any of your required representatives fails to attend any national convention or regional meeting we designate as mandatory (up to \$1,000 each person).
- (e) <u>Employee Training</u>. With the exception of the initial training program, we may require that you provide initial and/or ongoing training for your other employees using our online library of training modules on our LMS. The online training program includes tests that each trainee must pass before he or she will be deemed to have successfully completed the training program.
- 5.3. Other Guidance and Operating Assistance. Although we do not have an obligation to do so, we may advise you from time to time regarding various aspects of the operation of Your Store, including with respect to System Standards, operating procedures, purchasing fixtures, furnishings, equipment, signs, Approved Products, materials and supplies, and advertising and promotional programs. We will determine and control the form, format, method of delivery, and timing of any guidance and assistance we furnish or make available to you. You agree that we will not be liable to you or any other person, and you waive all claims for liability or damages of any type (whether direct, indirect, incidental, consequential, or exemplary), on account of any guidance or operating assistance offered by us in accordance with this Section 5.3, except to the extent caused by our gross negligence or intentional misconduct.

#### 6. **FEES**

- 6.1. <u>Initial Franchise Fee.</u> You must pay us, on your execution of this Agreement, a nonrecurring, non-refundable initial franchise fee in the amount of \$35,000. If applicable, we will apply the deposit of \$10,000 previously received from you to the initial franchise fee and the balance of \$25,000 is payable when you sign this Agreement. The initial franchise fee will be deemed fully earned by us upon receipt of the balance due and is not refundable.
- 6.2. <u>Continuing Royalty Fees</u>. You also agree, for the entire Term, to pay us a continuing royalty fee ("**Royalty**") equal to six percent (6%) of Your Store's Gross Revenue. Unless we provide otherwise in the Operations Manual, each Royalty payment will be due on or before the close of business on Wednesday of each week based on the Your Store's Gross Revenue for the preceding week as defined by us from time to time.
- 6.3. <u>Technology Fee.</u> We reserve the right to charge you a technology fee that we may use, in our discretion, to research, develop, maintain, update or upgrade technology for the System (the "**Technology Fee**"). If implemented, the Technology Fee will be due at the same time and in the same manner as the Royalty, unless we specify otherwise in writing. We may adjust the Technology Fee once per calendar year upon sixty (60) days' notice to you; provided, however, that we will not increase the potential maximum fee (currently the greater of \$100 per month or 1% of Gross Revenue) more than 10% per calendar year.
- 6.4. Method and Application of Payments. We will determine from time to time the method by which you must pay Royalty and all other fees and amounts due us or our Affiliates under this Agreement or relating to Your Store. We currently require payment of all such amounts by pre-authorized electronic bank transfer, and you agree to sign all documents necessary to allow us to initiate debit entries and/or credit correction entries to your designated bank account for that purpose. You agree to maintain, at all times, a balance in your designated bank account sufficient to allow the appropriate amount to be debited from your account. If you do not timely provide us with a report of your Gross Revenue to allow us the opportunity to calculate the amount of fees you owe, we have the right to estimate in good faith Your Store's

Gross Revenue for the missing period and debit your account in an amount equal to the fees that would be due based on such estimation. In making our good faith estimate, we may consider the last report of Gross Revenues that we received from you, any seasonal sales trends, and any system-wide averages and other pertinent information available to us. You are responsible for any penalties, fines or other similar expenses associated with the pre-authorized bank transfers described in this Section. Regardless of any designation by you, we have the right to apply any payments by you to any of your past-due indebtedness to us or our Affiliates.

- 6.5. <u>Late Fees; Interest on Late Payments</u>. To compensate us for the administrative and management costs resulting from your late payment or late report, we may charge a \$100 late fee for each delinquent payment, due when the delinquent payment is due, and for each delinquent report, due when the delinquent report is due. We will continue to charge a late fee for each period that the fee or report remains delinquent. These fees are neither interest nor a penalty. All amounts you owe to us or our Affiliates will bear interest from their due date until paid at a rate equal to the lesser of the highest rate allowed under applicable law or 1.5% per month, payable when the corresponding delinquent payment is made. You agree that this Section 6.5 does not constitute our or our Affiliates' agreement to accept payments after they are due or a commitment by us or our Affiliates to extend credit to you or otherwise to finance the operation of Your Store.
- 6.6. No Right of Offset. You have no right of "offset" and will not withhold payment, for any reason, of any Royalties, Brand Fund Contributions or any other payment due to us under this Agreement or any other agreement.

### 7. OBLIGATIONS RELATING TO OPERATIONS

7.1. Compliance with Laws and Good Business Practices. You must secure and maintain in force throughout the Term all required licenses, permits and certificates relating to the operation of Your Store and operate Your Store in full compliance with all applicable laws, ordinances and regulations, including PCI compliance standards. You agree to comply and assist us in our compliance efforts, as applicable, with any and all laws, regulations, Executive Orders or otherwise relating to anti-terrorist activities, including the U.S. Patriot Act, Executive Order 13224, and related U.S. Treasury or other regulations. In connection with such compliance efforts, you agree not to enter into any prohibited transactions and to properly perform any currency reporting and other activities relating to Your Store as may be required by us or by law. You and your Entity Owners confirm that you are not listed in or otherwise in violation of the Annex to Executive Order 13224 (or any other present and future laws, policies, lists, and other requirements of any governmental authority addressing or in any way relating to terrorist acts and acts of war) and agree not to hire any person so listed or have any dealing with a person so listed (the Annex is currently available at http://www.treasury.gov). You are solely responsible for ascertaining what actions must be taken by you to comply with all such laws, orders or regulations. Notwithstanding the foregoing, in the event any guidance or recommendation issued by any federal, state, or local authority directly or indirectly affects the operation of Your Store, you will not close Your Store unless you obtain our prior written consent. In all dealings with your customers, suppliers, us and the public, you must adhere to the highest standards of honesty, integrity, fair dealing and ethical conduct. You agree to refrain from any business or advertising practice which might injure our business or the goodwill associated with the Marks or other Stores. You agree to comply with our website privacy policy, as it may be amended periodically; you further agree to comply with any requests to return or delete consumer personal information, whether requested by us or directly by the consumer, as required by applicable data sharing privacy laws. You must notify us in writing within three business days of: (1) the commencement of any action, suit or proceeding relating to Your Store; (2) the issuance of any order, writ, injunction, award or decree of any court, agency or other governmental instrumentality relating to Your Store; (3) any notice of violation of any law, ordinance or regulation relating to Your Store; (4) receipt of any notice of complaint from the Better Business Bureau, any local, state or federal consumer affairs department or division, or any other government or independent third party involving a complaint from a client or potential client relating to Your Store; and (5) written complaints from any customer. You must immediately provide to us copies of any documentation you receive of events in (1) through (5) above and resolve the matter in a prompt and reasonable manner in accordance with good business practices.

- 7.2. Compliance with System Standards. The operation of Your Store in accordance with the System Standards is essential to preserve the goodwill of the Marks and all Stores within the System. Therefore, you agree that you will maintain and operate Your Store in accordance with each of the System Standards. We have the right to modify the System Standards from time to time, and you acknowledge that the modifications may obligate you to invest additional capital in Your Store and to incur higher operating costs. If, at any time in our reasonable judgment, Your Store does not meet our then-current System Standards, we will notify you, specifying in reasonable detail the actions to be taken by you to comply with System Standards. If you fail or refuse to initiate promptly and timely complete the necessary actions as set forth in the notice, in addition to any of our other rights to enforce this Agreement, we will have the right (but not the obligation) to enter Your Store premises and complete the necessary actions described in the notice. You will reimburse us for the entire cost thereof upon demand. You agree to cooperate fully with us in connection with any of our actions under this Section.
- 7.3. Restrictions on Operations. You may not operate Your Store at any site other than the Premises without our prior written consent. In addition, you may only offer and sell finished Approved Products that have been approved for sale over the counter to retail customers from Your Store, and may not sell Approved Products or any materials, supplies, or inventory bearing the Marks at any other location or through any alternative channel of distribution without our prior written consent. "Alternative channels of distribution" include, but are not limited to, the operation of a food cart or kiosk, sales through the Internet (or any other form of electronic commerce), vending machine sales, and mail order and telephone sales. Notwithstanding the above restrictions, you may: (i) offer and sell Approved Products as part of offsite catering events and company account programs, provided you or a third-party local delivery company deliver (and do not engage a major air/ground shipping carrier to deliver) Approved Products that meet System Standards for freshness the same day they are made and the sales are not part of a mail order program; (ii) engage in delivery of Approved Products from your Store according to the standards and specifications we periodically specify; (iii) offer samples of Approved Products at or directly in front of Your Store or other locations near Your Store as approved by your landlord; or (iv) upon our prior written approval, offer and sell Approved Products from a table, kiosk or cart at satellite locations that we approve. You may not sell to anyone any materials, supplies, or inventory used in the preparation of any Approved Products. Further, you may not sell any Approved Products to any person or entity purchasing the Approved Products for resale.
- 7.4. Internet Use. You may not advertise Your Store or the Approved Products over the internet (or any other form of electronic commerce) or establish a related site on the world wide web without our prior written consent. Your general conduct on the internet (or any other form of electronic commerce) and specifically your use of the Marks is subject to the provisions of this Agreement and our Brand Standards. Without limiting the foregoing, you agree to follow our policies and procedures as they may be communicated to you periodically in the Operations Manual or otherwise regarding the use of social media and similar methods of communication. You must obtain our written consent prior to your use of the Marks or any content using the Marks on social media, in any form available now or in the future, including but not limited to Facebook, Instagram and Twitter, and your failure to obtain such consent shall constitute a breach of this Agreement. We also have the right to monitor, edit and/or delete any social media content you may post that does not meet our Brand Standards. Further, you acknowledge that we have the right to require you to have access to the internet from the Premises and to submit all sales activity through the Computer system through a pre-approved router and internet connection. We also have the right to require you to establish and maintain a valid email address and authorize us to communicate with you by this method at such address.
- 7.5. Our Right to Inspect Your Store. We and our designated agents have the right to, at any reasonable time and without prior notice to you, to inspect the Premises, to observe, photograph and record Your Store's operations for such consecutive or intermittent periods as we deem necessary, to remove

samples of any Approved Products, materials or supplies for testing and analysis, to interview Your Store's personnel and customers, to inspect your Computer System (including hardware, software, security, configurations, connectivity, and data access), and to access, inspect and copy any books, records and documents relating to the operation of Your Store. You agree to cooperate fully with us in connection with any of our inspections, observations, photographing, videotaping, product removal and interviews.

7.6. <u>Surveys</u>. You will present to your customers such evaluation forms as we periodically require and will participate in and request your customers to participate in any surveys performed by or on our behalf.

# 7.7. Good Standing and Guaranties by Entity Owners. If you are an Entity:

- (a) you represent and warrant to us that you are and will, throughout the Term, remain in good standing under the laws of the state of your incorporation or formation and in all other states in which you are required to qualify; and
- (b) each of your Entity Owners will execute our the-current form of Guaranty pursuant to which they will personally assume and guaranty your performance of each and every obligation under this Agreement.
- 7.8. Insurance. Throughout the Term, you must maintain in force at your sole expense comprehensive insurance policies that comply with our specifications as to amount and type of coverage, which we may specify from time to time in writing. You must purchase and maintain for each Store you operate: (1) comprehensive general liability insurance with minimum limits of \$1,000,000 per occurrence and \$2,000,000 annual aggregate; (2) workers' compensation, employer's liability and other insurance to meet the greater of all applicable statutory requirements or the then-current minimum levels of coverage we require; (3) commercial property insurance including, at a minimum, fire, vandalism, theft, burglary, and extended coverage with limits of at least 100% replacement value of the Store premises, fixtures, equipment and inventory; (4) business interruption/time element coverage in the amounts we require either as a component of or an endorsement to a commercial property insurance policy; (5) automobile liability insurance, including personal injury, wrongful death and property damage, with limits of at least \$1,000,000 per occurrence; (6) data privacy insurance; (7) employment practices liability insurance; and (8) an umbrella liability coverage of \$2,000,000 each occurrence and \$4,000,000 in the aggregate. Your general liability policy must name us and our affiliates and any other person that we designate as an additional insured and must meet any other requirements that we designate. If you own more than one Store, (1) we may require you to obtain an umbrella liability policy; and (2) we may allow you to obtain single policy for such Stores provided that we periodically will determine the levels of insurance coverage that you must obtain and other requirements that you must satisfy for each Store covered by such policy. We place no restriction on what insurance carrier you use to obtain the coverage outlined above. We may periodically change the amounts of coverage required under these insurance policies, require a minimum rated insurance carrier, or require different or additional insurance coverages (including reasonable excess liability insurance) at any time to reflect inflation, identification of new risks, changes in law or standards of liability, higher damage awards or other relevant changes in circumstances.

You release and relieve us and our Affiliates, and all of our and their officers, directors, shareholders, employees, agents, successors, assigns, contractors, and invitees and waive your entire right of recovery against us and our Affiliates and all of our officers, directors, shareholders, employees, agents, successors, assigns, contractors, and invitees for loss or damage arising out of or incident to the perils required to be insured against under this Section, which perils occur in, on or about the Premises or relate to the operation of Your Store, whether due to the negligence of us or our Affiliates or you or any of our or your related parties.

The requirements we impose regarding insurance are for our own purposes. You must assess whether our requirements are sufficient for your own needs or to comply with applicable law. Your

compliance with the insurance requirements herein, and the availability of insurance coverage to defend and indemnify us, will not relieve you of your obligations under the defense, indemnification, and hold harmless provisions of this Agreement, which are separate and independent. If any of your insurers denies or limits coverage to us for any claim falling within the scope of the applicable policy or your obligations herein, then you will provide that defense and indemnity directly. Your requirements regarding insurance are not affected by any insurance we maintain ourselves.

- 7.9. Pricing. Unless prohibited by applicable law, we may periodically set a maximum or minimum price that you may charge for products and services offered by Your Store. If we impose such a maximum or minimum price for any product or service, you may charge any price for the product or service up to and including our designated maximum price or down to and including our designated minimum price. The designated maximum and minimum prices for the same product or service may, at our option, be the same. For any product or service for which we do not impose a maximum or minimum price, we may require you to comply with an advertising policy adopted by us which will prohibit you from advertising any price for a product or service that is different than our suggested retail price. Although you must comply with any advertising policy we adopt, you will not be prohibited from selling any product or service at a price above or below the suggested retail price unless we impose a maximum price or minimum price for such product or service.
- 7.10. <u>Information Security</u>. You must implement all administrative, physical and technical safeguards necessary to protect any information that can be used to identify an individual, including names, addresses, telephone numbers, e-mail addresses, employee identification numbers, signatures, passwords, financial information, credit card information, biometric or health data, government-issued identification numbers and credit report information ("**Personal Information**") in accordance with applicable law and industry best practices. It is entirely your responsibility (even if we provide you any assistance or guidance in that regard) to confirm that the safeguards you use to protect Personal Information comply with all applicable laws and industry best practices related to the collection, access use, storage, disposal and disclosure of Personal Information. If you become aware of a suspected or actual breach of security or unauthorized access involving Personal Information, you will notify us immediately and specify the extent to which Personal Information was compromised or disclosed.
- 7.11. Employees, Agents and Independent Contractors. You acknowledge and agree that you are solely responsible for all decisions relating to employees, agents, and independent contractors that you may hire to assist in the operation of Your Store. You agree that any employee, agent or independent contractor that you hire will be your employee, agent or independent contractor, and not our employee, agent or independent contractor. You also agree that you are exclusively responsible for the terms and conditions of employment of your employees, including recruiting, hiring, firing, training, compensation, work hours and schedules, work assignments, safety and security, discipline, and supervision. You agree to manage the employment functions of Your Store in compliance with federal, state, and local employment laws.

# 8. REPORTS AND RECORD KEEPING

- 8.1. Accounting, Reports and Financial Statements. You agree to establish and maintain a bookkeeping, accounting, record keeping and data processing system conforming to the requirements and formats that we prescribe. You agree to furnish to us reports relating to Your Store by the delivery method and in such form and content as we prescribe from time to time. These reports include, but are not limited to, the following:
  - (a) <u>Gross Revenue Reports</u>. On or before noon on Wednesday of each week, a report of Your Store's Gross Revenue for the previous week;

- (b) <u>Monthly Financial Reports</u>. Within 25 days after the end of each calendar month, a profit and loss statement for Your Store for the previous calendar month and a year-to-date statement of financial condition as of the end of the previous calendar month;
- (c) <u>Semi-Annual Reports</u>. By January 31 and July 31 each year, a balance sheet for Your Store as of December 31 and June 30, respectively;
- (d) <u>Tax Returns</u>. Within ten (10) days after the returns are filed, exact copies of federal and state income, sales and any other tax returns and the other forms, records, books and other information as we have the right to periodically require;
- (e) <u>Loan Status</u>. By January 15, April 15, July 15 and October 15 of each calendar year, reports on the status (including the outstanding balance, then-current payment amounts, and whether such loan is in good standing) of any loans outstanding as of the previous calendar quarter for which Your Store or any of Your Store's equipment is collateral. You must also deliver to us, within five (5) days after your receipt, copies of any default notices you receive from any of such lenders. You agree that we or our affiliates may contact your banks, other lenders, and vendors to obtain information regarding the status of loans of the type described herein and your accounts (including payment histories and any defaults), and you hereby authorize your bank, other lenders, and vendors to provide such information to us and our affiliates; and
- (f) <u>Local Store Marketing</u>. By January 31 of each calendar year, proof of your Local Store Marketing (as defined in Section 9.2), in the format required by us.

Further, at our request, you will provide current financial information for each of your Entity Owners who personally guaranty your obligations as necessary to demonstrate the ability of such Entity Owners to satisfy their financial obligations under their individual guarantees.

You agree to verify and sign each report and financial statement in the manner we prescribe. We may disclose data derived from these reports. Moreover, we may, as often as we deem appropriate (including on a daily basis), access the point-of-sale system and retrieve all information relating to the operation of Your Store. You agree to preserve and maintain all records in a secure location at Your Store for at least three (3) years (including sales checks, purchase orders, invoices, payroll records, customer lists, check stubs, sales tax records and returns, cash receipts and disbursement journals, and general ledgers). We also have the right to require you to have audited or reviewed financial statements or other financial reports that we reasonably designate prepared and submitted to us on an annual basis.

You must obtain, at your expense, from our designated vendor the technology platform that will collect monthly financial statements for the Store. You must submit monthly, quarterly and annual financial statements to us through the technology platform using a standardized chart of accounts. If you fail to provide the required financial statements, we may require you to use one of our preferred bookkeepers at your expense to prepare the required financial statements.

8.2. Our Right to Audit. At any time during business hours and without prior notice to you, we and our representatives have the right to inspect and audit the business records, bookkeeping and accounting records, sales and income tax records and returns and other records of Your Store as well as your books and records. You agree to fully cooperate with our representatives to conduct any inspection or audit. If an inspection or audit discloses an understatement of Your Store's Gross Revenue, you will pay to us, within 15 days after receipt of the inspection or audit report, the continuing fees due on the amount of the understatement, plus interest (at the rate and on the terms provided in Section 6.5) from the date originally due until the date of payment. Further, if inspection or audit is made necessary by your failure to furnish reports, supporting records or other information as required by this Agreement, or to furnish the reports, records or information on a timely basis, or if an understatement of Gross Revenue for the period of any audit is determined by the audit or inspection to be greater than two percent (2%), then within 15 days after

receipt of the inspection or audit report, you will reimburse us for the cost of the audit or inspection, including the charges of attorneys and any independent accountants and the travel expenses, room and board and compensation of our employees. If you fail to cooperate with our audit, or are unwilling or unable to provide us with sufficient records, including the records and reports that you are required to maintain under this Agreement, to complete the audit to our reasonable satisfaction, we may establish a reasonable estimation of your Gross Revenue based on the data available to us (which may include records regarding product purchases, percentage rent reports or other information obtained from third parties) and collect from you any estimated amount that we deem was underreported or underpaid pursuant to this Agreement. These remedies are in addition to our other remedies and rights under this Agreement or applicable law, and our right to audit will continue for two (2) years following expiration or termination of this Agreement.

# 9. MARKETING AND PROMOTION

- 9.1. <u>Brand Fund</u>. We operate a national brand fund (the "**Brand Fund**") to generally advertise and promote the Stores and the Marks in the System. You will pay us a monthly national Brand Fund contribution of three percent (3%) of your Store's gross Revenues (the "**Brand Fund Contribution**").
  - Administration of the Brand Fund. We will, without any fiduciary obligation to (a) you, direct or cause our designee to direct the Brand Fund. We will determine the creative concepts, materials and endorsements used and the geographic, market and media placement and allocation. The expenses that may be funded by the Brand Fund Contributions we collect include preparing and producing video, audio, and written materials and electronic media; developing, implementing, and maintaining a website for the System and Stores and related strategies; administering regional and multi-regional marketing and advertising programs, including purchasing trade journal, direct mail, and other media advertising and using advertising, promotion, and marketing agencies and other advisors to provide assistance; administering online advertising and marketing campaigns (including search engine, social media, email, and display ad campaigns); developing and maintaining application software designed to run on computers and similar devices, including tablets, smartphones and other mobile devices, as well as any evolutions or "next generations" of any such devices; implementing a loyalty program or other marketing programs designed to encourage the patronage of Stores; and reimbursing us or our designees for the reasonable salaries and benefits of personnel who manage and administer the Brand Fund, including administrative costs, travel and other expenses reasonably related to administering or directing the Brand Fund and its programs. We have the right, at our option, to use Brand Fund Contributions to prepare, furnish and/or offer for sale to you advertising, marketing and promotional formats and materials for your use in your own local marketing efforts.
  - (b) Accounting for the Brand Fund Contributions. The Brand Fund Contributions we collect will be accounted for separately from our other funds, and we may, but will not be required to, segregate them from our other operating funds. If you make a written request within 90 days following the end of our fiscal year, we will make available to you no later than 120 days after the end of our fiscal year, an annual statement of Brand Fund Contributions collected and costs incurred for the Brand Fund during the prior fiscal year. We are not required to have this statement independently audited.
  - (c) <u>Benefits to Individual Stores</u>. The Brand Fund is intended for general promotion. We do not guaranty any particular result or that our expenditure of Brand Fund Contributions or the conduct of the Brand Fund will directly, indirectly, or proportionally benefit Your Store or have any relationship to the amount of Brand Fund Contributions you pay.
  - (d) <u>Your Participation in the Brand Fund</u>. You agree to have your Store participate, in accordance with the rules and requirements we establish for the Brand Fund.

9.2. Your Advertising and Promotional Activities. In addition to paying Brand Fund Contributions and participating in the Brand Fund, you must spend at least \$2,000 per month for marketing purposes in the geographic area of your Store, beginning three (3) months after opening your Store and continuing for nine (9) months after opening ("Local Store Marketing"). You must use the Local Store Marketing on approved local marketing and promotion of your Store. You must provide us with an annual statement evidencing your Local Store Marketing expenditures. If you do not spend the minimum required amount on Local Store Marketing, you must deposit with us the difference between what you did spend and the minimum required amount for deposit in the Brand Fund. We will deposit such amounts into the Brand Fund to be spent on local marketing and promotional activities in, near or attempting to target the geographic area of your Store, including marketing and promotional activities on the Internet. You may use in your Local Store Marketing, including any social media activities, only those materials that we provide or make available to you or for which we have given our prior written approval. You must submit your proposed materials for our review at least 15 days prior to your proposed use. After the first year of operations, you must spend each month the greater of (a) 2% of Gross Revenue, or (b) \$1,000. Your Local Store Marketing must include the creation of an annual marketing plan, which we must approve.

We also may require you to participate in a social medial technology platform ("Social Media Technology Platform") and allocate a portion of your Local Store Marketing for the Social Media Technology Platform.

- Our Advertising Materials. From time to time, we may provide you with copies of 9.3. advertising, marketing and promotional formats and materials for use in Your Store, which we have prepared using Brand Fund Contributions we have collected from Stores. While we expect to provide some of these materials without charge, we reserve the right to charge you for these materials or require that you pay certain shipping and related expenses associated with delivering them to you. In addition, we have the right to develop and market special mandatory promotional items for Stores and require you to maintain a representative inventory of these promotional items to meet public demand. In such case, we will make these items available to you at our direct cost plus a reasonable mark-up and any shipping, handling and storage charges. If you do not place minimum orders of products and other items necessary for a mandatory promotion or product roll-out by a certain date, we have the right to send, or direct suppliers to send, an automatic shipment of a specified minimum quantity of such products and items to you, and you must accept and pay for them upon receipt. We also have the right to conduct coupon or discount promotions. In such case, we have the right to require you to accept coupons or discounts that are issued or communicated by us or our Affiliates and presented at Your Store by your customers. You may receive certain compensation for these coupons when you tender them to us in accordance with our System Standards.
- 9.4. Advertising Cooperatives. We have the right to designate any geographical area for purposes of establishing a regional advertising and promotional cooperative ("Cooperative"), and to determine whether a Cooperative is applicable to Your Store. If a Cooperative exists or is established for the area in which Your Store is located, you must join and participate in the activities of the Cooperative. Each Cooperative will be organized and governed in a form and manner, and will commence and cease operations on the dates, we determine. Each Cooperative will be organized for the exclusive purpose of administering regional advertising programs and standardizing advertising materials for use by the members in local advertising. No promotional or advertising plans or materials may be used by a Cooperative or furnished to its members without our prior approval. We will determine the amount and timing of the required contributions to the Cooperative and all activities and other decisions of the Cooperative with input from the Stores that are members of the Cooperative. Contributions will be payable to the Cooperative or us or our designee if we are administering the accounting and vendor payments of the Cooperative.
- 9.5. <u>Participation in Certain Programs and Promotions</u>. You must use your best efforts to promote and advertise your Store and must participate in all advertising and promotional programs we establish in the manner we direct. You must, at your expense, participate in, and honor all provisions of

any gift card and/or loyalty program that we have established or may establish and as we may modify, as further described in the Operations Manual.

#### 10. USE OF THE MARKS AND CONFIDENTIAL INFORMATION

- 10.1. Ownership and Goodwill of Marks. You acknowledge that we or our Affiliates are the exclusive owners of the Marks and that your right to use the Marks is derived solely from this Agreement and is limited to the conduct of business in compliance with this Agreement and all applicable System Standards, specifications and operating procedures that we require. Any unauthorized use of the Marks by you will constitute a breach of this Agreement and an infringement of our rights in the Marks. You agree that your use of the Marks and any goodwill established by that use will be for our and our Affiliates' exclusive benefit. This Agreement does not confer any past, present or future goodwill or other interests in the Marks upon you, other than the right to operate Your Store in compliance with this Agreement.
- Store you are operating under this Agreement as the sole identification of Your Store. You will not use any Mark as part of any corporate or trade name or with any prefix, suffix or other modifying words, terms, designs or symbols (other than logos licensed to you under this Agreement), or in any modified form, nor may you use any Mark in connection with the performance or sale of any unauthorized services or products or in any other manner not expressly authorized in writing by us. You agree to display the Marks prominently at Your Store and on materials used to promote Your Store, in each case, only in the manner we direct or approve in writing. You agree to refrain from any business or marketing practice which may be injurious to Your Store, our business and the goodwill associated with the Marks and other Stores. You agree to give such notices of trade and service mark registrations as we specify and to obtain such fictitious or assumed name registrations as may be required under applicable law. You may not use any Mark as part of an electronic mail address or on any sites on the internet or world wide web or as part of any internet domain name or address.
- 10.3. <u>Discontinuance of Use of Marks</u>. We have the right to require you to modify or discontinue use of any Marks or use one or more additional or substitute trade or service marks if we determine it is advisable to do so at any time. You agree to comply with our directions in that regard. We will not be required to reimburse you for your expenses in modifying or discontinuing the use of a Mark and substituting a different Mark or for any loss of goodwill associated with any modified or discontinued Mark or for any expenditures made by you to promote a modified or substitute Mark.
- 10.4. <u>Infringements and Claims</u>. You agree to immediately notify us of any apparent infringement of or challenge to your use of any Mark or claim by any person of any rights in any Mark, and you will not communicate with any person other than us or our counsel in connection with the infringement, challenge or claim. We and our Affiliates will have the right to take the action we deem appropriate and control exclusively any litigation, U.S. Patent and Trademark Office proceeding or any other administrative or court proceeding arising out of any such infringement, challenge or claim or otherwise relating to any Mark. You agree to execute any instruments and documents, render such assistance and do those things as, in the opinion of our legal counsel, may be necessary or advisable to protect and maintain our interests in any litigation or U.S. Patent and Trademark Office or other proceeding or otherwise to protect and maintain our interests in the Marks. If you have timely notified us of the claim or proceeding and have otherwise complied with this Agreement, we will indemnify you against and reimburse you for all damages for which you are held liable in any proceeding arising out of your authorized use of any Mark in compliance with this Agreement. You will allow us and our Affiliates to control the defense of any proceeding arising out of your authorized use of any Mark.
- 10.5. <u>Non-Disparagement</u>. You agree not to (and to use your best efforts to cause your current and former owners, officers, directors, principals, agents, partners, employees, representatives, attorneys, spouses, affiliates, successors and assigns not to) disparage or otherwise speak or write negatively, directly or indirectly, of us, our Affiliates, any of our or our Affiliates' directors, officers, employees, representatives

or affiliates, current and former franchisees or developers of us or our Affiliates, the Marks, the System, any Store (including Your Store), any business using the Marks, any other brand or service-marked or trademarked concept of us or our Affiliates, or which would subject the Mrs. Fields brand or such other brands to ridicule, scandal, reproach, scorn, or indignity, or which would negatively impact the goodwill of us, the Mrs. Fields brand or such other brands.

- Concepts Developed by You. You may not test, offer, or sell any new products without our prior written consent. As between us and you, we are the sole owner of all right, title, and interest in and to the System and any Confidential Information. All improvements, developments, derivative works, enhancements, or modifications to the System and any Confidential Information (collectively, "Innovations") made or created by you, your employees or your contractors, whether developed separately or in conjunction with us, shall be owned solely by us. You represent, warrant, and covenant that your employees and contractors are bound by written agreements assigning all rights in and to any Innovations developed or created by them to you. To the extent that you, your employees or your contractors are deemed to have any interest in such Innovations, you hereby agree to assign, and do assign, all right, title and interest in and to such Innovations to us. To that end, you shall execute, verify, and deliver such documents (including assignments) and perform such other acts (including appearances as a witness) as we may reasonably request for use in applying for, obtaining, perfecting, evidencing, sustaining, and enforcing such ownership rights in and to the Innovations, and the assignment thereof. Your obligation to assist us with respect to such ownership rights shall continue beyond the expiration or termination of this Agreement. If we are unable for any reason, after reasonable effort, to secure your signature on any document needed in connection with the actions specified in this Section, you hereby irrevocably designate and appoint us and our duly authorized officers and agents as your agent and attorney in fact, which appointment is coupled with an interest and is irrevocable, to act for and on your behalf to execute, verify, and file any such documents and to do all other lawfully permitted acts to further the purposes of this Section with the same legal force and effect as if executed by you. The obligations of this Section shall survive any expiration or termination of the Agreement.
- 10.7. <u>Confidential Information</u>. During the Term, we may disclose certain Confidential Information to you. You are not acquiring any interest in Confidential Information, other than the right to utilize the portions of it we disclose to you in the operation of Your Store during the Term. Your use or duplication of any Confidential Information in any other business will constitute an unfair method of competition and a violation of this Agreement. The Confidential Information is proprietary, includes our trade secrets and is disclosed to you solely on the condition that you agree not to use Confidential Information in any other business or capacity, to maintain the absolute confidentiality of Confidential Information during and after the Term, not to make unauthorized copies of, disclose or distribute any portion of Confidential Information, and to adopt and implement all reasonable procedures that we prescribe to prevent unauthorized use or disclosure of Confidential Information, including restrictions on disclosure of Confidential Information to your employees and compliance with the requirement that certain key employees execute confidentiality agreements as a condition of employment.

# 11. COVENANTS NOT TO COMPETE

- 11.1. <u>In-Term Non-Compete and Non-Solicitation</u>. We would be unable to protect the Confidential Information against unauthorized use or disclosure and would be unable to encourage a free exchange of ideas and information among Stores if you or the manager of Your Store were permitted to hold interests in or perform services for a Competitive Business. We have granted the Franchise to you in consideration of and reliance upon your agreement to deal exclusively with us. Therefore, during the Term, no Restricted Person and no manager of Your Store will:
  - (a) Have any direct, indirect or beneficial interest in a Competitive Business, except other Stores operated by you under franchise agreements with us or any of our Affiliates;

- (b) Perform services as a director, officer, manager, employee, consultant, representative, agent, landlord or otherwise for a Competitive Business, except other Stores operated by you under franchise agreements with us or any of our Affiliates;
- (c) Divert or attempt to divert any actual or potential business or customer of Your Store to a Competitive Business; or
- (d) Directly or indirectly appropriate, use or duplicate the System or System Standards, or any portion thereof, for use in any other business or endeavor.
- 11.2. <u>Post-Term Non-Compete and Non-Solicitation</u>. You agree that all Restricted Persons will also be bound by the restrictions described in Section 11.1 above for a period of two (2) years following the expiration or termination of this Agreement for any reason other than as a result of our default (three (3) years if we purchase Your Store as provided in Section 14.2), except that the restrictions described in clauses (a) through (c) in Section 11.1 will apply only with respect to Competitive Businesses that are located or operating within ten (10) miles of Your Store and within ten (10) miles of any other Store.
- 11.3. <u>Validity and Application of Restrictions</u>. You represent to us that you and the other Restricted Persons have skills and abilities of a general nature and have other opportunities for exploiting those skills. Consequently, enforcement of the covenants made in this Section 11 will not deprive you or them of their personal goodwill or ability to earn a living. Nothing in this Section 11 is intended to or will prohibit the ownership by any Restricted Person of two percent (2%) or less of the shares of a class of securities of a Competitive Business that are listed on a public stock exchange or traded on the over-the-counter market.
- 11.4. <u>Enforcement of Non-Competes</u>. If any covenant in this Agreement which restricts competitive activity is deemed by a court to be unenforceable by virtue of its scope, but would be enforceable by reducing any part or all of the covenant, you and we agree that the covenant will be enforced to the fullest extent permissible under the laws and public policies applied in the jurisdiction in which enforcement is sought.

# 12. TRANSFERS

- 12.1. <u>Transfers by Us.</u> This Agreement is fully transferable by us and will inure to the benefit of any transferee or other legal successor to our interest in this Agreement. We may delegate the performance of any portion or all of our obligations under this Agreement to third-party designees, whether these designees are our agents or independent contractors with whom we have contracted to perform these obligations.
- 12.2. Restrictions on Transfers by You. Your rights and duties created by this Agreement are personal to you, and we have granted this Agreement to you in reliance upon our perceptions of the individual or collective character, skill, aptitude, attitude, business ability and financial capacity of you and, if you are not an individual, your Entity Owners. Accordingly, you may not delegate any of your obligations under this Agreement, and you may not engage in or permit any Transfer without our prior written approval. Any Transfer without our approval will constitute a breach of this Agreement and will be void and of no effect.
- 12.3. <u>Conditions for Our Approval of a Proposed Transfer</u>. You must give us 60 days' prior written notice of any proposed Transfer. Your notice must include a description of the proposed Transfer and such information as we require about the persons involved in the proposed Transfer. If you are in full compliance with this Agreement, we will not unreasonably withhold our approval of a Transfer that meets all of the following requirements:

- (a) <u>Character</u>. The proposed transferee and, if the proposed Transfer involves an Entity, the Entity Owners, must, in our determination, be of good moral character and otherwise meet our then applicable standards for owners of Stores;
- (b) <u>Business Experience and Financial Capacity</u>. The transferee and, if the transferee is an Entity, its Entity Owners must, in our determination, have sufficient business experience, aptitude and financial resources given the nature of the proposed Transfer;
- (c) <u>Training</u>. The proposed transferee and/or its senior management personnel have completed to our satisfaction our then-current training program for transferees after signing the franchise documents set forth herein, but prior to assuming operations of the Store;
- (d) <u>Satisfaction of Obligations</u>. You have paid all amounts owed for purchases by you from us and our Affiliates and all other amounts owed to us or our Affiliates and third-party creditors;
- (e) Execution of Assignment and Assumption Agreement. You and your transferring Entity Owners, if you are an Entity, the transferee and its Entity Owners, if the transferee is an Entity, and us have entered into our then-current form of assignment and assumption agreement, pursuant to which: (i) the transferee has agreed to be bound by and has expressly assumed all of the terms and conditions of this Agreement for the remainder of its term; (ii) the transferee's Entity Owners, if any, have executed our then-current form of guaranty; and (iii) you and your Entity Owners, if any, have agreed to release us and our Affiliates and our and their respective officers, directors, employees and agents from any and all claims;
- (f) <u>Execution of New Agreement at Our Option</u>. In addition to entering into our thencurrent assignment and assumption agreement, at our option, the transferee has executed our thencurrent form of Franchise Agreement for a term equal to the remainder of the current Term, and if the transferee is an Entity, each Entity Owner of the transferee has executed our then-current form of guaranty;
- (g) <u>Payment of Transfer Fees</u>. You or the transferee pays us a transfer fee equal to fifty percent (50%) of then-current initial franchise fee for existing franchisees, a portion of which may be a non-refundable deposit that you or the transferee must pay to us when you submit your request for our consent to the Transfer. However, we will not charge a transfer fee if the Transfer is among your existing Entity Owners and the names and identity of all Entity Owners remain the same following the Transfer;
- (h) <u>Approval of Terms of Transfer</u>. We have approved the material terms and conditions of the Transfer, including the price and terms of payment. However, our approval of a Transfer does not ensure the transferee's success nor should the transferee rely upon our approval of the Transfer in determining whether to acquire Your Store;
- (i) <u>Subordination</u>. If you (or your Entity Owners) finance any part of the sale price of the transferred interest, you and the Entity Owners have agreed that all obligations of the transferee under any promissory notes, agreements or security interests reserved by you (or your Entity Owners) will be subordinate to the transferee's obligations to us and our Affiliates;
- (j) <u>Non-Competition and Other Post-Term Obligations</u>. Each Restricted Person who, as a result of the Transfer, will no longer have an interest in you or Your Store has executed a non-competition agreement acknowledging the application of the post-term covenants described in Section 11.2 and otherwise agrees to all other post-term obligations as though the Agreement had expired on the date of the Transfer;
- (k) <u>Landlord Consent</u>. If required, the landlord of the Premises consents to the assignment or sublease of the Premises to the transferee;

- (l) <u>Refurbishment</u>. You or the transferee has agreed to any refurbishment of the Store required by us to bring the Store in compliance with the then-current System Standards and Trade Dress:
- (m) <u>Licensed Escrow Professional</u>. You and the transferee, at your cost, use a licensed escrow professional or other qualified third party acceptable to us to conduct the closing of the Transfer. We have the right to require that all documents and fees payable to us shall be deposited into escrow prior to the time that your transferee attends our training program, together with escrow instructions in form and content satisfactory to us providing for a final closing of the proposed Transfer after the transferee successfully completes all required training; and
- (n) <u>Other Conditions</u>. You and your transferring Entity Owners, if you are an Entity, have complied with any other conditions that we reasonably require from time to time as part of our transfer policies.

In connection with any assignment permitted under this Section 12.3, you will provide us with all documents to be executed by you and the proposed transferee at least 30 days prior to execution, and you will provide us with a copy of all fully executed documents and agreements that were signed in connection with the Transfer.

- 12.4. Transfer to a Wholly Owned Entity. If you are in full compliance with this Agreement, you will have the right to transfer your rights in this Agreement to an Entity which will conduct no business other than the business contemplated by this Agreement, which you actually manage and in which you maintain management control and own and control 100% of the equity and voting power of all issued and outstanding capital stock or other ownership interests. If a transfer is made under this Section 12.4, you will remain personally liable under this Agreement as if the transfer to the Entity had not occurred, and we reserve the right to require you to execute our then-current form of personal guaranty. The articles of formation and other governing documents of the Entity will recite that the issuance and assignment of any interest in the corporation or limited liability company is restricted by the terms of this Section 12, and all issued and outstanding stock certificates and other documents representing ownership interests in you will bear a legend reciting or referring to these restrictions.
- Our Right of First Refusal. If you or one or more of your Entity Owners desires to make a Transfer, you or the Entity Owner will obtain a bona fide, executed written offer and an earnest money deposit (in the amount of five percent (5%) or more of the offering price) from a responsible and fully disclosed purchaser and will immediately submit to us a true and complete copy of such offer, which will include details of the payment terms of the proposed sale, the sources and terms of any financing for the proposed purchase price, a list of the owners of record and beneficial owners of any offeror that is an Entity, and a list of the individuals ultimately owning or controlling the offeror. If the offeror or an owner of the offeror is a publicly held Entity, you will also submit to us copies of the most current annual and quarterly reports of the publicly held Entity. To be a valid, bona fide offer, the proposed purchase price will be denominated in a dollar amount. The offer must apply only to an interest in you or this Agreement and may not include an offer to purchase any other property or rights of you or your Entity Owners. However, if the offeror proposes to buy any other property or rights from you or your Entity Owners under a separate, contemporaneous offer, the price and terms of purchase offered to you or your Entity Owners for the interest in you or this Agreement will reflect the bona fide price offered for that interest and will not reflect any value for any other property or rights. We will have the right, exercisable by written notice delivered to you or your Entity Owners within 30 days from the date of delivery of an exact copy of the offer to us, to purchase the interests proposed to be Transferred for the price and on the terms and conditions contained in the offer except that (a) we have the right to substitute cash for any form of payment proposed in the offer, (b) our credit will be deemed equal to the credit of any proposed purchaser, (c) we will have not less than 60 days to close the purchase, and (d) we will have the right to receive from you all customary representations and warranties given by the seller of the assets of a business or equity interest in an Entity, as applicable, including representations and warranties as to ownership, condition of and title to assets,

absence of liens and encumbrances relating to the ownership interest and assets, and validity of contracts and liabilities affecting the assets being purchased, contingent or otherwise. If we do not exercise our right of first refusal, you (or your Entity Owners) may complete the proposed Transfer as and on the terms outlined in the offer, subject to our approval of the proposed Transfer and transferee as provided in Sections 12.2 and 12.3. However, if the sale to the purchaser is not completed within 120 days after delivery of the offer to us, or if there is a material change in the terms of the sale, our right of first refusal will be extended for 30 days after the expiration of the 120-day period or after the material change in the terms of the sale.

- disability or, if you are an Entity, upon the death or permanent disability of an individual owner of a Controlling Interest in you, the executor, administrator, conservator or other personal representative of that person will transfer his interest in this Agreement or his Controlling Interest in you within a reasonable time, not to exceed six (6) months from the date of death or permanent disability, to a third party approved by us. A transfer under this Section 12.6, including transfer by devise or inheritance, will be subject to all of the terms and conditions for Transfers contained in Sections 12.2 and 12.3, and unless transferred by gift, devise or inheritance, subject to the terms of Section 12.5. Failure to dispose of such interest within the specified period of time will constitute a breach of this Agreement. For purposes of this Agreement, the term "permanent disability" means a mental or physical disability, impairment or condition that is reasonably expected to prevent or actually does prevent you or an owner of a Controlling Interest in you from supervising the operation of Your Store for a period of six (6) months from the onset of such disability, impairment or condition.
- 12.7. <u>Effect of Consent to Transfer</u>. Our consent to a Transfer will not constitute a waiver of any claims we may have against the transferor nor be deemed a waiver of our right to demand full compliance by the transferee with the terms or conditions of this Agreement.
- 12.8. Preparation of a Financial Report by You. We have the right to require you to prepare and furnish to a prospective transferee and/or us such financial reports and other data relating to Your Store and its operations as we deem necessary or appropriate for the prospective transferee and/or us to evaluate the Store and the proposed transfer. You agree that we have the right to confer with prospective transferees and furnish them with information concerning Your Store and the proposed transfer without being held liable to you, except for intentional misstatements made to any such transferee. Any such information furnished by us to prospective transferees is for the sole purpose of permitting the transferees to evaluate Your Store and the proposed transfer and shall not be construed in any manner or form whatsoever as financial performance representations, or representations or claims of success or failure.

#### 13. **DEFAULT AND TERMINATION**

- 13.1. <u>Your Defaults</u>. You will be in default under the terms of this Agreement, and we may terminate this Agreement, effective immediately on written notice, if any of the following occur:
  - (a) <u>Insolvency</u>. You file a petition in bankruptcy or for reorganization or for an arrangement pursuant to any federal or state bankruptcy law or any similar federal or state law, or are adjudicated a bankrupt or make an assignment for the benefit of creditors or admit in writing your inability to pay your debts generally as they become due, or if a petition or answer proposing the adjudication of you as bankrupt or your reorganization pursuant to any federal or state bankruptcy law or any similar federal or state law is filed in any court and you consent to or acquiesce in the filing thereof or such petition or answer is not discharged or denied within 60 days after the occurrence of any of the foregoing, or if a receiver, trustee or liquidator of you or of all or substantially all of your assets or your interest in this Agreement is appointed in any proceeding brought by you, or if any such receiver, trustee or liquidator is appointed in any proceeding brought against you and is not discharged within 60 days after the occurrence thereof, or if you consent to or acquiesce in such appointment (any such event described in this Section 13.1(a) being referred to as an "**Insolvency Event**");

- (b) <u>Unauthorized Transfer</u>. A Transfer occurs in violation of the provisions of Section 12;
- (c) <u>Failure to Complete Training</u>. You or your initial store manager fails to complete all phases of the initial training program to our satisfaction;
- (d) <u>Misstatements and Other Adverse Developments</u>. You (or, if you are an Entity, any Entity Owner) have made any material misrepresentation or omission in your application for the Franchise, are convicted by a trial court of or plead no contest to a felony or to any other crime or offense that, in our determination, may adversely affect the goodwill associated with the Marks, or if you engage in any conduct which, in our determination, may adversely affect the reputation of any Store or the goodwill associated with the Marks;
- (e) <u>Unauthorized Use of Marks or Confidential Information</u>. You or an Entity Owner make any unauthorized use of the Marks or any unauthorized use or disclosure of Confidential Information:
- (f) <u>Abandonment</u>. You abandon or fail actively to operate Your Store for three (3) consecutive days unless Your Store has been closed for a purpose approved in advance by us in writing or because of fire, flood or other casualty or government order;
- (g) <u>Breach of Lease; Loss of Right of Possession</u>. You are in breach of any of your obligations under your Lease or you lose the right to possession of the Premises;
- (h) Failure to Comply with Certain System Standards and Health Requirements. You fail or refuse to comply with System Standards relating to the cleanliness or sanitation of Your Store, or you violate any health, safety or sanitation law, ordinance or regulation and, in any event, do not correct the failure or violation within 48 hours of our written notice or notice from a governmental agency regarding a violation;
- (i) <u>Understatements of Gross Revenue</u>. You understate Your Store's Gross Revenue in any report or financial statement by an amount greater than two percent (2%);
- (j) <u>Failure to Make Payments</u>. You or any of your Affiliates fail, after 10 days' prior written notice, to make payments, when due, of any amounts due to us or our Affiliates under this Agreement or any other agreement with us or our Affiliates, or fail to make payments, when due, of any amounts due to vendors, distributors, suppliers or landlords of the Store that relate to the Store's operation;
- (k) <u>Failure to Pay Taxes</u>. You fail to pay any federal or state income, sales or other taxes due with respect to Your Store's operations unless you are in good faith contesting your liability for the taxes;
- (1) <u>Failure of Inspection</u>. You fail to achieve a passing score reasonably established by us on two (2) consecutive announced or unannounced store inspections conducted by us or our agents;
- (m) Other Breaches. You fail to comply with any other provision of this Agreement or any System Standard and do not correct the failure within 30 days following our written notice of the failure:
- (n) <u>Repeated Breaches.</u> You fail on two (2) or more separate occasions within any period of 12 consecutive months or on three (3) occasions during the Term: (i) to submit when due reports or other data, information or supporting records; (ii) to pay when due the continuing fees or

other payments due to us or our Affiliates; or (iii) to otherwise comply with this Agreement, whether or not the failures to comply are corrected after notice thereof is delivered to you;

- (o) <u>Financing Defaults</u>. You default with respect to any of your obligations to us or any other lender under any financing provided to you in connection with this Franchise Agreement or a purchase of Your Store's assets;
- (p) <u>Default of any Other Agreement</u>. You default in the performance or observance of any of your obligations under any other agreement with us or our Affiliates and do not correct the default within the cure periods, if any, provided under such agreements;
- (q) <u>Failure to Secure Store Premises</u>. You fail to obtain our approval of and secure the Premises for Your Store within the six (6) month period described in Section 2.1 of this Agreement, if at the time of signing this Agreement we have not identified the Premises on Schedule 3 attached to this Agreement;
- (r) <u>Failure to Open Store by Start Date</u>. You fail to open the Store by your Start Date, as described in Section 4.5 of this Agreement; or
- (s) <u>Possession or Use of Unauthorized Products</u>. You possess or use unauthorized products at the Premises of Your Store, as specified periodically by us in the Operations Manual or otherwise.
- 13.2. Other Rights and Remedies. If you cure any default after the applicable cure period has expired, we still have the right to terminate this Agreement. In any event, our right to terminate this Agreement is in addition to whatever other rights and remedies are available to us. Without limiting the foregoing, we reserve the right to interrupt your product shipments or ordering privileges instead of or in addition to exercising our right to terminate this Agreement, or to require you to sign our then-current form of Franchise Agreement if we choose to rescind our termination of this Agreement. We also reserve the right (but not the obligation) to take actions necessary to cure the default at your sole expense and to establish reasonable conditions that you must satisfy to cure any default, such as requiring you to pay for a year of monthly Store inspections if you commit a default relating to Section 13.1(1).
- 13.3. Assumption of Management. If you abandon the operation of Your Store or if we are deciding whether to exercise our right to purchase Your Store as described in Section 14.2 below, we have the right, at our option, to enter the Premises and assume the management of Your Store for any period of time we deem appropriate. If we assume management of Your Store, we will appoint a manager who will maintain Store operations. All funds from the operation of Your Store during the period of management by our appointed manager will be kept in a separate fund, and all expenses of Your Store, including compensation, other costs, and travel and living expenses of our appointed manager, will be charged to such fund. As compensation for such management services, we will charge such fund ten percent (10%) of the Gross Revenue of Your Store during the period of our management. Operation of Your Store during any such period will be on your behalf, provided that we will have a duty only to utilize our good faith effort and will not be liable to you for any debts or obligations incurred by Your Store or to any of your creditors for any merchandise, materials, supplies or services purchased by Your Store during any period in which Your Store is managed by our appointed manager. You will maintain in force for Your Store all insurance policies required by this Agreement. Our right to assume management of Your Store pursuant to this Section is in addition to and does not affect our right to terminate this Agreement under Section 13.1.

#### 14. **POST-TERM OBLIGATIONS**

14.1. <u>Your Obligations</u>. You agree that, on expiration or termination of this Agreement, you and, as applicable, the Restricted Persons, will do the following:

(a) <u>Payment of Amounts Owed to Us and Others</u>. Pay us within 15 days after the date of termination or expiration of this Agreement, or such later date as the amounts due to us are determined, all fees and other amounts you owe us or our Affiliates, including all interest due thereon.

#### (b) Discontinue Use of Marks.

- (i) Immediately close Your Store for business to customers and cease to sell, directly or indirectly, any products or services of any kind and in any manner from the Premises and/or using the Marks, unless we direct you otherwise in connection with our exercise of our option to purchase pursuant to Section 14.2;
- (ii) Not directly or indirectly at any time or in any manner (except with respect to other Stores owned and operated by you) identify yourself or any business as a current or former Store, or as a franchisee, licensee or dealer of us or our Affiliates, use any Mark, any colorable imitation of a Mark or other indicia of a Store in any manner or for any purpose or utilize for any purpose any trade name, trade or service mark or other commercial symbol that suggests or indicates a connection or association with us or our Affiliates:
- (iii) Deliver to us all signs, sign-faces, sign-cabinets, marketing materials, forms, invoices and other materials containing any Mark or otherwise identifying or relating to a Store and allow us, without liability, to remove all such items from Your Store;
- (iv) Take such action as may be required to cancel all fictitious or assumed name or equivalent registrations relating to your use of any Mark;
- (v) If we do not purchase Your Store as provided in Section 14.2, make the changes to the exterior and interior appearance of Your Store to distinguish the Trade Dress as are reasonably required by us;
- (vi) Deliver all materials and supplies identified by the Marks in full cases or packages to us for credit and dispose of all other materials and supplies identified by the Marks within 30 days after the effective date of termination of this Agreement;
- (vii) Notify the telephone company and all telephone and online directory publishers of the termination of your right to use any telephone and facsimile numbers and any regular, classified or other telephone and online directory listings associated with any Mark and to authorize transfer of those rights to us, or at our direction, our designee. You agree that, as between you and us, we have the right to and interest in all telephone and facsimile numbers and directory listings associated with any Mark. You authorize us and appoint us and any of our officers as your attorney in fact, to direct the telephone company and all telephone and online directory publishers to transfer any telephone and facsimile numbers and directory listings relating to Your Store to us, or our designee, should you fail or refuse to do so, and the telephone company and all telephone and online directory publishers may accept such direction or this Agreement as conclusive of our exclusive rights in the telephone and facsimile numbers and directory listings and our authority to direct their transfer; and
- (viii) Furnish us, within 30 days after the effective date of termination, with evidence satisfactory to us of your compliance with the obligations in this Section 14.1(b). If you fail to fulfill any of the obligations contained in this Section 14.1(b), we have the right, at our option, to perform such obligations at your expense.
- (c) <u>Discontinuance of Use of Confidential Information</u>. Immediately cease to use any Confidential Information disclosed to you pursuant to this Agreement in any business or otherwise

and you will return to us all copies of the Operations Manual and any other confidential materials which we have loaned to you.

(d) <u>Compliance with Post-Term Covenants</u>. Immediately comply with the post-term covenants described in Section 11.2 above.

# 14.2. Our Option to Purchase Your Store.

- (a) Option to Purchase. On expiration or termination of this Agreement other than as a result of our default, we will have the right, at our option, exercisable by giving written notice thereof within 60 days from the date of such termination or expiration, to acquire from you, free and clear of all liens or encumbrances, Your Store, your right to possess the Premises, any or all of the Approved Products, materials, and supplies that are in good and saleable condition and not obsolete or discontinued (the "Inventory"), and any or all of the equipment, furnishings, signs, and the other tangible assets of Your Store (collectively, with the Inventory, the "Assets"). We have the right to assign this option to purchase and our rights under this Section. We will be entitled to all customary warranties and representations in connection with our purchase, including representations and warranties as to ownership, condition of and title to the Assets, no liens and encumbrances on the Assets, and validity of contracts and agreements and liabilities benefiting us or affecting the Assets, contingent or otherwise.
- (b) <u>Purchase Price</u>. The purchase price for the Assets will be the net realizable value of the tangible assets in accordance with the liquidation basis of accounting (not the value of Your Store as a going concern) ("**Liquidation Value**"). If you dispute the calculation of the Liquidation Value, the purchase price will be determined by one independent accredited appraiser designated by us who will calculate the purchase price applying the criteria specified above. We agree to select the appraiser within 15 days after we receive the financial and other information necessary to calculate the purchase price (if you and we have not agreed on the purchase price before then). You and we will share equally the appraiser's fees and expenses. The appraiser must complete its calculation within 30 days after its appointment. The purchase price will be the appraiser's determination of the Liquidation Value, applying the appropriate mechanism as described above. We will have the right to set off against and reduce the purchase price by any and all amounts owed by you to us or our Affiliates.
- (c) Payment of Purchase Price. Closing of the purchase will take place on a date we select that is within 90 days after (1) your receipt of our notice of exercise of this option to purchase Your Store, or (2) the purchase price is determined as described in paragraph (b) above. The purchase price will be paid in cash at the closing of the purchase, at which time you will deliver instruments transferring to us good and merchantable title to the Assets purchased, free and clear of all liens and encumbrances and with all sales and other transfer taxes paid by you, and with all licenses or permits of Your Store which may be assigned or transferred. If the closing of the purchase does not occur within the specified time period because you fail to act diligently in connection with the purchase, the purchase price will be reduced by ten percent (10%). The purchase price will be further reduced by ten percent (10%) per month for each subsequent month you fail to act diligently to consummate the purchase. Prior to closing, you and we will comply with the applicable Bulk Sales provisions of the Uniform Commercial Code as enacted in the state where Your Store is located.
- (d) <u>Lease of Premises</u>. In connection with the purchase, you will also deliver to us an assignment of the Lease (or, if assignment is prohibited, a sublease for the full remaining term and on the same terms and conditions as your Lease). If you own the Premises of Your Store, you agree to lease the Premises to us pursuant to the terms of our standard lease.
- (e) <u>Interim Management</u>. If we exercise the option to purchase Your Store, pending the closing of such purchase, we have the right to appoint a manager to maintain the operation of

Your Store or, at our option, require you to close Your Store during such time period without removing any assets. If we appoint a manager to maintain the operation of Your Store pending closing of such purchase, we will have the right to manage Your Store under the same terms and conditions as described in Section 13.3.

- (f) <u>Termination of Franchise Agreement</u>. Upon the closing of the purchase of the Assets and satisfaction by you of all of your obligations under this Agreement accruing through the closing, this Agreement will terminate.
- Lost Revenue Damages. If we terminate this Agreement because of your breach or if you terminate this Agreement without cause, you and we agree that it would be difficult, if not impossible, to determine the amount of damages that we would suffer due to the loss or interruption of the revenue stream we otherwise would have derived from your continued payment of Royalty and Brand Fund Contributions through the remainder of the Term. Therefore, you and we agree that a reasonable estimate of such damages, less any cost savings we might have experienced (the "Lost Revenue Damages"), is an amount equal to the net present value of the Royalty and Brand Fund Contributions that would have become due had this Agreement not been terminated, from the date of termination to the earlier of: (a) five (5) years following the date of termination, or (b) the scheduled expiration of the Term (the "Measurement Period"). Lost Revenue Damages will be calculated as follows: (1) the number of calendar months in the Measurement Period, multiplied by (2) the aggregate of the Royalty and Brand Fund Contribution percentages, multiplied by (3) the average monthly Gross Revenue of Your Store during the 12 full calendar months immediately preceding the termination date; provided, that if as of the termination date, Your Store has not been operating for at least 12 months, the average monthly Gross Revenue of all Stores operating under the same Marks during the entirety of our fiscal year immediately preceding the termination date. You agree to pay us Lost Revenue Damages, as calculated in accordance with this Section, within 15 days after this Agreement expires or is terminated, or on any later date that we determine. You and we agree that the calculation described in this Section is a calculation only of the Lost Revenue Damages and that nothing herein shall preclude or limit us from proving and recovering any other damages caused by your breach of the Agreement.
- 14.4. <u>Continuing Obligations</u>. All obligations which expressly or by their nature survive the termination or expiration of this Agreement will continue in full force and effect subsequent to and notwithstanding termination or expiration and until they are satisfied in full or by their nature expire.

# 15. RELATIONSHIP OF THE PARTIES/INDEMNIFICATION

- 15.1. <u>Independent Contractors</u>. This Agreement does not create a fiduciary relationship between us and you. We and you are independent contractors and nothing in this Agreement is intended to make either of us a general or special agent, joint venturer, partner or employee of the other for any purpose. You will conspicuously identify yourself in all dealings as the owner of Your Store under a franchise granted by us and will place such other notices of independent ownership on the forms, business cards, stationery, marketing and other materials as we have the right to require from time to time. You agree not to use the Marks or otherwise engage in any conduct which might, in any way, indicate (a) to your employees or anyone else that you are our agent or (b) that might otherwise result in our direct or indirect liability for any of your indebtedness or obligations. You acknowledge that we are in the business of franchising and licensing systems and trademarks for the delivery of goods and services, and we are not in the same business as you are.
- 15.2. Our Approval and Enforcement. You have the sole right and responsibility for the manner and means by which the day-to-day operation of Your Store is determined and conducted and for achieving your business objectives. Subject to any approval, inspection and enforcement rights reserved to us, this right and responsibility includes the employment, supervision, setting the conditions of employment and discharge for your employees at Your Store, daily maintenance, safety concerns, and the achievement of conformity with the System Standards. Our retention and exercise of the right to approve certain matters,

to inspect Your Store and its operation and to enforce our rights, exists only to the extent we believe necessary to protect our and our Affiliates' interest in the System and the Marks. Neither the retention nor the exercise is for the purpose of establishing any control, or the duty to take control, over those matters which are clearly reserved to you, nor shall they be construed to do so.

- 15.3. <u>Taxes</u>. We will have no liability for any sales, use, service, occupation, excise, gross receipts, income, property or other taxes, whether levied upon you or your assets or upon us, arising in connection with your sales or the business conducted by you pursuant to this Agreement, except for taxes that we are required by law to collect from you with respect to purchases from us and except for our own income taxes. Payment of all such taxes will be your responsibility.
- Indemnification. You agree to indemnify, defend, and hold harmless us, our Affiliates, 15.4. and our and their respective owners, directors, managers, officers, employees, agents, successors, and assignees (the "Indemnified Parties") against, and to reimburse the Indemnified Parties for, all claims, obligations, and damages directly or indirectly arising out of the operation of Your Store, the business you conduct under this Agreement, product liabilities claims or defective manufacturing of Approved Products by you, or your breach of this Agreement, including those alleged to be caused by the Indemnified Party's negligence, unless (and then only to the extent that) the claims, obligations, or damages are determined to be caused solely by the Indemnified Party's intentional misconduct in a final, unappealable ruling issued by a court with competent jurisdiction. For purposes of this indemnification, "claims" include all obligations, damages (actual, consequential, or otherwise), and costs that any Indemnified Party reasonably incurs in defending any claim against it, including reasonable accountants', arbitrators', attorneys', and expert witness fees, costs of investigation and proof of facts, court costs, travel and living expenses, and other expenses of litigation or alternative dispute resolution, regardless of whether litigation or alternative dispute resolution is commenced. Each Indemnified Party may defend any claim against it at your expense and agree to settlements or take any other remedial, corrective, or other actions. This indemnity will continue in full force and effect subsequent to and notwithstanding this Agreement's expiration or termination. An Indemnified Party need not seek recovery from any insurer or other third party, or otherwise mitigate its losses and expenses, in order to maintain and recover fully a claim against you under this subparagraph. You agree that a failure to pursue a recovery or mitigate a loss will not reduce or alter the amounts that an Indemnified Party may recover from you under this Section.
- 15.5. <u>Waiver of Claims</u>. You waive all claims against us for damages to property or injuries to persons arising out of the operation of Your Store.

#### 16. **SECURITY AGREEMENT**

Agreement, including payments owed to us or our Affiliates for your purchases, you hereby collaterally assign to us the Lease and grant us a security interest in all of the assets used in or associated with the operation of Your Store, including inventory, accounts, supplies, contracts, cash derived from the operation of Your Store and sale of other assets, and proceeds and products of all those assets. You agree to execute such other documents as we may reasonably request in order to further document, perfect and record our security interest. If you default in any of your obligations under this Agreement, we may exercise all rights of a secured creditor granted to us by law, in addition to our other rights under this Agreement and at law. If a third-party lender requires that we subordinate our security interest in the assets of Your Store as a condition to lending you working capital for the construction or operation of Your Store, we will agree to subordinate pursuant to terms and conditions determined by us. This Agreement will be deemed to be a Security Agreement and Financing Statement and may be filed for record as such in the records of any county and state that we deem appropriate to protect our interests.

# 17. **DISPUTE RESOLUTION**

- 17.1. <u>Injunctive Relief.</u> Nothing in this Agreement, including the provisions of Section 17.8, bars our right to obtain specific performance of the provisions of this Agreement and injunctive relief against any threatened or actual conduct that will cause us, the Marks, or the System loss or damage, under customary equity rules, including applicable rules for obtaining restraining orders and temporary or preliminary injunctions. You agree that we may seek such relief from any court of competent jurisdiction in addition to such further or other relief as may be available to us at law or in equity. You agree that we will not be required to post a bond to obtain injunctive relief and that your only remedy if an injunction is entered against you will be the dissolution of that injunction, if warranted, upon due hearing (all claims for damages by injunction being expressly waived hereby).
- 17.2. <u>Rights of Parties Are Cumulative</u>. Our and your rights under this Agreement are cumulative and the exercise or enforcement of any right or remedy under this Agreement will not preclude the exercise or enforcement by a party of any other right or remedy under this Agreement which it is entitled by law or this Agreement to exercise or enforce.
- 17.3. Costs and Attorneys' Fees. The prevailing party in any arbitration or litigation shall be entitled to recover from the other party all damages, costs and expenses, including arbitration and court costs and reasonable attorneys' fees, incurred by the prevailing party in connection with such arbitration or litigation. If we incur expenses in connection with your failure to pay when due amounts owing to us, to submit when due any reports, information or supporting records or otherwise to comply with this Agreement, including court costs, legal and accounting fees, you will reimburse us for any such costs and expenses which we incur.
- 17.4. Governing Law. All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. Sections 1 et seq.). Except to the extent governed by the Federal Arbitration Act, the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.), or other federal law, this Agreement, the Franchise, and all claims arising from the relationship between us and you will be governed by the laws of the state of Delaware, the state in which we were formed, without regard to its conflict of laws rules, except that any state law regulating the offer or sale of franchises or governing the relationship of a franchisor and its franchisee will not apply unless its jurisdictional requirements are met independently without reference to this section.
- 17.5. <u>Consent to Jurisdiction</u>. Subject to the obligation to arbitrate under Section 17.8 below, you and the Restricted Persons agree that all actions arising under this Agreement or otherwise as a result of the relationship between you and us must be commenced in Wilmington, Delaware, and you (and each Restricted Person) irrevocably submit to the jurisdiction of that court and waive any objection you (or the owner) might have to either the jurisdiction of or venue in that court.
- 17.6. <u>Waiver of Punitive Damages and Jury Trial</u>. Except for your obligation to indemnify us for third-party claims under Section 15.4, we and you (and the Restricted Persons) waive to the fullest extent permitted by law any right to or claim for any punitive or exemplary damages against the other and agree that, in the event of a dispute between us and you, the party making a claim will be limited to equitable relief and to recovery of any actual damages it sustains. WE AND YOU IRREVOCABLY WAIVE TRIAL BY JURY IN ANY ACTION OR PROCEEDING, BROUGHT BY EITHER OF US.
- 17.7. <u>Limitation of Claims</u>; <u>Waiver of Class Action</u>. Except for claims arising from your non-payment or underpayment of amounts you owe us, any and all claims arising out of or relating to this Agreement or our relationship with you will be barred unless a judicial or arbitration proceeding is commenced in accordance with this Agreement within one (1) year from the date on which the party asserting the claim knew or should have known of the facts giving rise to the claims. We and you agree that any proceeding will be conducted on an individual basis and that any proceeding between us and any of our Affiliates, or our and their respective owners, officers, directors, agents, and employees, on the one

hand, and you (or your owners, guarantors, affiliates, and employees), on the other hand, may not be: (i) conducted on a class-wide basis, (ii) commenced, conducted or consolidated with any other proceeding, (iii) joined with any claim of an unaffiliated third-party, or (iv) brought on your behalf by any association or agent. No previous course of dealing shall be admissible to explain, modify, or contradict the terms of this Agreement. No implied covenant of good faith and fair dealing shall be used to alter the express terms of this Agreement.

Arbitration. We and you agree that all controversies, disputes, or claims between us or our 17.8. Affiliates, and our and their respective owners, officers, directors, agents, and employees, on the one hand, and you and your Affiliates, and your and their owners, officers, directors, agents and employees, on the other hand, arising out of or related to: (1) this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates); (2) our relationship with you; (3) the scope or validity of this Agreement or any other agreement between you (or any of your owners) and us (or any of our Affiliates) or any provision of any of such agreements (including the validity and scope of the arbitration provision under this Section 17.8, which we and you acknowledge is to be determined by an arbitrator, not a court); or (4) any System Standard, must be submitted for binding arbitration, on demand of either party, to the American Arbitration Association. The arbitration proceedings will be conducted by one arbitrator and, except as this Section otherwise provides, according to the American Arbitration Association's then-current Commercial Arbitration Rules. All proceedings will be conducted at a suitable location chosen by the arbitrator that is within Wilmington, Delaware. All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). The interim and final awards of the arbitrator shall be final and binding upon each party, and judgment upon the arbitrator's awards may be entered in any court of competent jurisdiction.

The arbitrator has the right to award or include in his or her awards any relief which he or she deems proper, including, without limitation, money damages, pre- and post-award interest, interim costs and attorneys' fees, specific performance, and injunctive relief, provided that the arbitrator may not declare any of the trademarks owned by us or our affiliates generic or otherwise invalid, or award any punitive or exemplary damages against any party to the arbitration proceeding (we and you hereby waiving to the fullest extent permitted by law any such right to or claim for any punitive or exemplary damages against any party to the arbitration proceeding). Further, at the conclusion of the arbitration, the arbitrator shall award to the prevailing party its attorneys' fees and costs.

We and you agree to be bound by the provisions of any applicable contractual or statutory limitations provision, whichever expires earlier. We and you further agree that, in any arbitration proceeding, each party must submit or file any claim which would constitute a compulsory counterclaim (as defined by Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding. Any claim which is not submitted or filed as required will be forever barred. The arbitrator may not consider any settlement discussions or offers that might have been made by either you or us.

We and you agree that arbitration will be conducted on an individual basis and that an arbitration proceeding between us and any of our Affiliates, or our and their respective owners, officers, directors, agents, and employees, on the one hand, and you (or your owners, guarantors, affiliates, and employees), on the other hand, may not be: (i) conducted on a class-wide basis, (ii) commenced, conducted or consolidated with any other arbitration proceeding, (iii) joined with any separate claim of an unaffiliated third-party, or (iv) brought on your behalf by any association or agent. Notwithstanding the foregoing, if any court or arbitrator determines that all or any part of the preceding sentence is unenforceable with respect to a dispute, controversy or claim that otherwise would be subject to arbitration under this Section, then all parties agree that this arbitration clause shall not apply to that dispute, controversy or claim and that such dispute, controversy or claim shall be resolved in a judicial proceeding in accordance with the dispute resolution provisions of this Agreement.

We and you agree that, in any arbitration arising as described in this Section, the arbitrator shall have full authority to manage any necessary exchange of information among the parties with a view to

achieving an efficient and economical resolution of the dispute. The parties may only serve reasonable requests for documents, which must be limited to documents upon which a party intends to rely or documents that are directly relevant and material to a significant disputed issue in the case or to the case's outcome. The document requests shall be restricted in terms of time frame, subject matter and persons or entities to which the requests pertain and shall not include broad phraseology such as "all documents directly or indirectly related to." You and we further agree that no interrogatories or requests to admit shall be propounded, unless the parties later mutually agree to their use.

In any arbitration each side may take no more than three depositions, unless the parties mutually agree to additional depositions. Each side's depositions are to consume no more than a total of 15 hours, and each deposition shall be limited to 5 hours, unless the parties mutually agree to additional time.

The provisions of this Section are intended to benefit and bind certain third-party non-signatories.

The provisions of this Section will continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

Any provisions of this Agreement that pertain to judicial proceedings shall be subject to the agreement to arbitrate contained in this Section.

#### 18. **GENERAL PROVISIONS**

- 18.1. Severability. Each article, section, paragraph, term and provision of this Agreement will be considered severable and if, for any reason, any provision of this Agreement is held to be invalid, contrary to or in conflict with any applicable present or future law or regulation in a final, unappealable ruling issued by any court, agency or tribunal with competent jurisdiction in a proceeding to which we are a party, that ruling will not impair the operation of, or have any other effect upon, such other portions of this Agreement as may remain otherwise intelligible, and such other portions will continue to be given full force and effect and bind the parties, although any portion held to be invalid will be deemed not to be a part of this Agreement from the date the time for appeal expires, if you are a party thereto, otherwise upon your receipt of a notice of non-enforcement thereof from us.
- 18.2. Rights Provided by Law. If any applicable and binding law or rule of any jurisdiction requires a greater prior notice of the termination or non-renewal of this Agreement than is required under this Agreement, or the taking of some other action not required under this Agreement, or if, under any applicable and binding law or rule of any jurisdiction, any provision of this Agreement is invalid or unenforceable, the prior notice and/or other action required by such law or rule will be substituted for the comparable provisions of this Agreement, and we will have the right to modify the invalid or unenforceable provision to the extent required to be valid and enforceable. You agree to be bound by any promise or covenant imposing the maximum duty permitted by law which is subsumed within the terms of any provision of this Agreement, as though it were separately articulated in and made a part of this Agreement, that may result from striking from any of the provisions of this Agreement any portion or portions which a court may hold to be unenforceable in a final decision to which we are a party, or from reducing the scope of any promise or covenant to the extent required to comply with such a court order. Such modifications to this Agreement will be effective only in such jurisdiction, unless we elect to give them greater applicability, and will be enforced as originally made and entered into in all other jurisdictions.
- 18.3. Waivers by Either of Us. Either we or you may by written instrument unilaterally waive or reduce any obligation of or restriction upon the other under this Agreement, effective upon delivery of written notice of waiver to the other or such other effective date stated in the notice of waiver. Any waiver granted by us will be without prejudice to any other rights we may have, will be subject to our continuing review and may be revoked by us at any time and for any reason, effective upon delivery to you of 10 days' prior written notice.

- 18.4. Certain Acts Not to Constitute Waivers. Neither we nor you will be deemed to have waived or impaired any right, power or option reserved by this Agreement (including the right to demand exact compliance with every term, condition and covenant in this Agreement or to declare any breach to be a default and to terminate this Agreement prior to the expiration of its term) by virtue of: (i) any custom or practice of the parties at variance with the terms of this Agreement; (ii) any failure, refusal or neglect of us or you to exercise any right under this Agreement or to insist upon exact compliance by the other with its obligations under this Agreement, including any waiver, forbearance, delay, failure or omission by us to exercise any right, power or option, whether of the same, similar or different nature, with respect to other Stores or franchise agreements; or (iii) our acceptance of any payments due from you after any breach of this Agreement.
- 18.5. <u>Interpretation of Rights and Obligations</u>. The following provisions will apply to and govern the interpretation of this Agreement, the parties' rights under this Agreement, and the relationship between the parties:
  - (a) <u>Our Rights</u>. Whenever this Agreement provides that we have a certain right, that right is absolute and the parties intend that our exercise of that right will not be subject to any limitation or review. We have the right to operate, administrate, develop, and change the System in any manner that is not specifically precluded by the provisions of this Agreement.
  - (b) Our Reasonable Business Judgment. Whenever we reserve or are deemed to have reserved discretion in a particular area or where we agree or are deemed to be required to exercise our rights reasonably or in good faith, we will satisfy our obligations whenever we exercise reasonable business judgment in making our decision or exercising our rights. A decision or action by us will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available, if our decision or action is intended, in whole or significant part, to promote or benefit the System generally even if the decision or action also promotes a financial or other individual interest of us. Neither you nor any third party (including a trier of fact), shall substitute its judgment for our reasonable business judgment.
  - Construction. The preambles and exhibits are a part of this Agreement, which together with this Agreement constitute our and your entire agreement, and there are no other oral or written understandings or agreements between us and you, or oral or written representations by us, relating to the subject matter of this Agreement, the franchise relationship, or Your Store (any understandings or agreements reached, or any representations made, before this Agreement are superseded by this Agreement). Nothing in this or in any related agreement, however, is intended to disclaim the representations we made in the Franchise Disclosure Document that we furnish to you. Any policies that we adopt and implement from time to time to guide us in our decision-making are subject to change, are not a part of this Agreement, and are not binding on us. Except as expressly stated, nothing in this Agreement is intended or deemed to confer any rights or remedies upon any person or legal entity not a party to this Agreement. If two or more persons are at any time the owners of the Franchise and Your Store, whether as partners or joint venturers, their obligations and liabilities to us will be joint and several. Unless otherwise specified, all references to a number of days shall mean calendar days and not business days. The term "Your Store" includes all of the assets of the Store you operate under this Agreement, including its revenue and the Lease. Unless indicated otherwise, the term "including" means "including, without limitation."
- 18.6. Notice of Our Potential Profit. We and/or our Affiliates have the right from time to time to make available to you goods, products and/or services for use in Your Store on the sale of which we and/or our Affiliates may make a profit. We and/or our Affiliates have the right from time to time to receive consideration from suppliers, distributors and/or manufacturers related (directly or indirectly) to sales of goods, products or services to you, the promotion of goods, products or services by the System or in consideration of services rendered or rights licensed to such persons. We and/or our Affiliates shall be entitled to use or not use any such profits and/or consideration for any purpose.

- 18.7. <u>Binding Effect</u>. Subject to the restrictions on Transfers contained in this Agreement, this Agreement is binding upon the parties hereto and their respective executors, administrators, heirs, assigns and successors in interest and will not be modified except by written agreement signed by both you and us.
- 18.8. <u>Execution</u>. This Agreement may be executed in counterparts which, when taken together, will constitute a single instrument. Signatures transmitted by fax or scanned and emailed will be given the full force and effect of originals.
- 18.9. Notices and Payments. All written notices and reports permitted or required to be delivered by the provisions of this Agreement will be deemed so delivered at the time delivered by hand; one (1) business day after being placed in the hands of a commercial courier service for next business day delivery; or three (3) business days after placement in the United States Mail by registered or certified mail, return receipt requested, postage prepaid, and will be addressed to the parties at the addresses set forth on the first page of this Agreement or to such other address as a party may specify in a written notice to the other party; provided, however, that any notice we provide to you may be addressed to your attention at Your Store. Any required payment or report not actually received by us during regular business hours on the date due (or postmarked by postal authorities at least two (2) days prior thereto) will be deemed delinquent.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Agreement on the dates shown below and made effective as of the Effective Date.

| MRS. FIELDS FRANCHISING, LLC  | FRANCHISEE:            |  |
|-------------------------------|------------------------|--|
| Ву:                           | By:                    |  |
| Name:                         | Name:                  |  |
| Title:                        | Title (if applicable): |  |
| Date*:                        | Date:                  |  |
| (*This is the Effective Date) |                        |  |

# **SCHEDULE 1**

# OWNERSHIP ADDENDUM TO MRS. FIELDS FRANCHISING, LLC FRANCHISE AGREEMENT

| 1.                           | Form of Owner.     |                         |                                                                                                                                          |  |  |  |  |
|------------------------------|--------------------|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                              | (a)                | Individual Proprietors  | <b>hip</b> . Your owner(s) (is) (are) as follows:                                                                                        |  |  |  |  |
|                              | (h)                | Comparation Limited     | Liability Company on Portnership. Voy word                                                                                               |  |  |  |  |
|                              | (b) incorporated o | r formed on             | , under the laws of the State of                                                                                                         |  |  |  |  |
|                              | You have not       | conducted business unde | r any name other than your corporate, limited liability icated in the following:                                                         |  |  |  |  |
| 2.                           | in the Franchis    |                         | me of each person who is one of your owners (as defined or of one of your owners, and fully describes the nature of pages if necessary). |  |  |  |  |
|                              | Owner's Name       |                         | Type / Percentage of Interest                                                                                                            |  |  |  |  |
|                              |                    |                         | %                                                                                                                                        |  |  |  |  |
|                              |                    |                         | %                                                                                                                                        |  |  |  |  |
|                              |                    |                         | %                                                                                                                                        |  |  |  |  |
| MRS. FIELDS FRANCHISING, LLC |                    | NCHISING, LLC           | FRANCHISEE:                                                                                                                              |  |  |  |  |
| Ву: _                        |                    |                         |                                                                                                                                          |  |  |  |  |
| Name                         | :                  |                         | Name:                                                                                                                                    |  |  |  |  |
| Title:                       |                    |                         | Title (if applicable):                                                                                                                   |  |  |  |  |
| Date:                        |                    |                         | Date:                                                                                                                                    |  |  |  |  |

# SCHEDULE 2 GUARANTY AND ASSUMPTION OF OBLIGATIONS

\_\_\_\_\_, 20\_\_\_\_\_, by each of the undersigned.

THIS GUARANTY AND ASSUMPTION OF OBLIGATIONS is given this \_\_\_\_\_ day of

| (as |
|-----|
| RS  |
| (a) |
| lec |
| e") |
| he  |
| ery |
| ng  |
| on  |
|     |
|     |

Each of the undersigned consents and agrees that: (1) his or her direct and immediate liability under this Guaranty will be joint and several, both with Franchisee and among other guarantors; (2) he or she will render any payment or performance required under the Agreement upon demand if Franchisee fails or refuses punctually to do so; (3) this liability will not be contingent or conditioned upon our pursuit of any remedies against Franchisee or any other person; (4) this liability will not be diminished, relieved, or otherwise affected by any extension of time, credit, or other indulgence which we may from time to time grant to Franchisee or to any other person, including, without limitation, the acceptance of any partial payment or performance or the compromise or release of any claims, none of which will in any way modify or amend this Guaranty, which will be continuing and irrevocable during the term of the Agreement; and (5) at our request, the undersigned shall present updated financial information to us as reasonably necessary to demonstrate his or her ability to satisfy the financial obligations of Franchisee under the Agreement.

Each of the undersigned waives: (i) all rights to payments and claims for reimbursement or subrogation which any of the undersigned may have against Franchisee arising as a result of the undersigned's execution of and performance under this Guaranty; and (ii) acceptance and notice of acceptance by us of his or her undertakings under this Guaranty, notice of demand for payment of any indebtedness or non-performance of any obligations hereby guaranteed, protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed, and any other notices to which he or she may be entitled.

If the undersigned guarantor is not an individual, it agrees that it will make no distributions to any persons at any time Franchisee is subject to an uncured notice of default.

Each of the undersigned represents and warrants that, if no signature appears below for such undersigned's spouse, such undersigned is either not married or, if married, is a resident of a state which does not require the consent of both spouses to encumber the assets of a marital estate.

The provisions contained in Section 17 (*Dispute Resolution*) of the Agreement, including Section 17.3 (*Costs and Attorneys' Fees*), Section 17.5 (*Consent to Jurisdiction*), and Section 17.8 (*Arbitration*) of the Agreement are incorporated into this Guaranty by reference and shall govern this Guaranty and any disputes between the undersigned and us. If we are required to engage legal counsel in connection with any failure by the undersigned to comply with this Guaranty, the undersigned shall reimburse us for any of the above-listed costs and expenses we incur.

By signing below, the undersigned spouse of the Guarantor indicated below, acknowledges and consents to the guaranty given herein by his/her spouse. Such consent also serves to bind the assets of the marital estate to Guarantor's performance of this Guaranty.

**IN WITNESS WHEREOF**, each of the undersigned has affixed his or her signature on the same day and year as this Guaranty and Assumption of Obligations was executed.

| GUARANTOR(S) | SPOUSE(S) |
|--------------|-----------|
|              |           |
| Name:        | Name:     |
| Sign:        | Sign:     |
| Address:     | Address:  |
|              |           |
|              |           |
|              |           |
| Name:        | Name:     |
| Sign:        | Sign:     |
| Address:     | Address:  |
|              |           |
|              |           |
|              |           |

#### **SCHEDULE 3**

## TO MRS. FIELDS FRANCHISING, LLC FRANCHISE AGREEMENT STORE PREMISES; START DATE

If at the time of signing this Agreement you have obtained our approval of and secured the Premises for Your Store, the Premises for Your Store and your applicable Start Date are identified below. However, if at the time of signing this Agreement you have not obtained our approval of and secured the Premises for Your Store, we and you will sign an Alternative Schedule 3 identifying the Premises of Your Store and your applicable Start Date, assuming you obtain our approval of and secure the Premises for Your Store within the applicable 6-month period described in Section 2.1 of this Agreement.

| 1.           | <u>Premises</u> . You and we agree that | at Your Store will be located at and only at the following  |
|--------------|-----------------------------------------|-------------------------------------------------------------|
| Premises: _  |                                         |                                                             |
|              |                                         |                                                             |
| 2            | Card Date Version laws are all          | at the College in the Great Date the city of the Atlanta    |
| 2.           |                                         | at the following is the Start Date described in Section 4.5 |
| of the Pranc | mse Agreement.                          |                                                             |
| MRS. FIEL    | LDS FRANCHISING, LLC                    | FRANCHISEE:                                                 |
| D.,,         |                                         | D <sub>vv</sub>                                             |
| Dy<br>Name:  |                                         | By:<br>Name:                                                |
| Title:       |                                         | Title (if applicable):                                      |
| Date:        |                                         | Date:                                                       |

#### **ALTERNATIVE SCHEDULE 3**

## TO MRS. FIELDS FRANCHISING, LLC FRANCHISE AGREEMENT STORE PREMISES; START DATE

| This Schedule, with a date of                        | , 20, is attached to and is an integral part            |
|------------------------------------------------------|---------------------------------------------------------|
| of the Mrs. Fields Franchising, LLC Franchise        | Agreement between you and us with a date of             |
| , 20 (the "Franchise Agree                           | •                                                       |
|                                                      |                                                         |
| 1. Premises. You and we agree that Y                 | our Store will be located at and only at the following  |
| Premises:                                            | •                                                       |
|                                                      |                                                         |
|                                                      |                                                         |
|                                                      |                                                         |
| 2. Start Date. You and we agree that the             | ne following is the Start Date described in Section 4.5 |
| of the Franchise Agreement:                          | <u> </u>                                                |
|                                                      |                                                         |
| 3. Defined Terms. All capitalized terms.             | ms contained in this Exhibit and not defined in this    |
| Exhibit will have the same meaning as provided in th |                                                         |
|                                                      | C                                                       |
| IN WITNESS WHEREOF, the parties have e               | executed and delivered this Exhibit on the day and year |
| first written above.                                 | •                                                       |
|                                                      |                                                         |
| MRS. FIELDS FRANCHISING, LLC                         | FRANCHISEE:                                             |
|                                                      |                                                         |
|                                                      |                                                         |
| By:                                                  | By:                                                     |
| Name:                                                | Name:                                                   |
| Title:                                               | Title (if applicable):                                  |
| Date:                                                | Date:                                                   |
| MRS. FIELDS FRANCHISING, LLC  By: Name: Title: Date: | By: Name: Title (if applicable):                        |

#### SCHEDULE 4

## TO MRS. FIELDS FRANCHISING, LLC FRANCHISE AGREEMENT LEASE ADDENDUM

|                                              | s entered into as of the date of the, Landlord and                                                                  | •                                                 |
|----------------------------------------------|---------------------------------------------------------------------------------------------------------------------|---------------------------------------------------|
| located at will use to operate a Mrs. Fields | parties to that certain Lease of even da<br>s® store under a Franchise Agreeme<br>Landlord and Tenant desire to ame | , which Tenant ent between Tenant and Mrs. Fields |

In consideration of the foregoing and the promises contained in the Lease, the parties agree as follows:

- 1. <u>Permitted Use</u>. Landlord and Tenant agree that so long as the Franchise Agreement remains in effect, Tenant may use the Lease premises only for a Mrs. Fields® store and Tenant may offer for sale and sell at the premises only those specialty snacks and other bakery items and related products, together with those services, which Franchisor approves.
- 2. <u>Notice of Default</u>. Landlord will provide Franchisor, by certified US mail or a recognized overnight delivery service at the address provided in Section 9 below, a minimum 30 day notice of any default under the Lease before Landlord initiates any action to terminate the Lease or exercise any remedy for such default.
- 3. <u>Cure</u>. Either Tenant or Franchisor may cure defaults under the Lease and Landlord will accept performance of obligations due under the Lease, as specified in the Lease, by either Franchisor or Tenant. Franchisor will not, however, be under any obligation to cure any default and nothing herein will require Franchisor at any time to comply with or take any action under the provisions of the Lease.
- 4. <u>Rights of Franchisor After Cure</u>. If Franchisor commences cure of any default under the Lease within the 30 day notice period described in Section 2 above, and if Franchisor thereafter diligently completes cure, Franchisor may, but will not be obligated to, give notice to Landlord and become Tenant under the Lease, in which event Landlord will not be entitled to terminate the Lease.
- 5. <u>Assignment and Renewal</u>. Landlord consents to an assignment or transfer of Tenant's rights under the Lease to Franchisor at any time during the term of the Lease; provided that such assignment or transfer is subject to Franchisor's written agreement to accept such assignment or transfer. Landlord will give Franchisor notice of expiration of the term of the Lease at least three months in advance thereof and grant Franchisor the right, but not the obligation, to exercise any then-existing renewal rights under the Lease.
- 6. <u>Permitted Assignments by the Franchisor</u>. If Franchisor becomes the Tenant under the rights described in Section 5 above, the following language will amend and be superior to any assignment and subletting language stated in the Lease:

Tenant will not be required to obtain Landlord's consent to assign this Lease to: (a) the parent or majority-owned subsidiary or affiliate of Tenant; (b) a person or entity having substantially the same net worth as the Tenant at the time of Transfer (collectively, a "Permitted Assignment") or (c) any entity resulting from a (i) merger, (ii) consolidation, (iii) initial public offering, or (iv) sale of the Tenant's assets. If there is a Permitted Assignment as described herein,

Tenant's obligations under the Lease will terminate as of the effective date of the Permitted Assignment.

- 7. Right of Entry and Subordination. Landlord will give Franchisor access to the Store at reasonable times on not less than 24 hours' notice (or such shorter notice as may be reasonable when circumstances dictate) either to inspect the Store for compliance with Franchisor's requirements, to remove from the Store any items bearing Franchisor's marks or logos or to take other action permissible under the Agreements between Tenant and Franchisor. Landlord specifically subordinates any lien it may have in such items to Franchisor's rights as licensor of the marks or logos displayed on items.
- 8. <u>Vacating Premises</u>. Upon vacating the Lease premises, or termination of the Franchise Agreement or Lease (whichever occurs first), Tenant must remove all signs and materials bearing any of the marks or logos.
  - 9. <u>Notices</u>. Any notices to Franchisor hereunder will be sent to:

Mrs. Fields Franchising, LLC 1717 S. 4800 W. Salt Lake City, Utah 84104

- 10. <u>Benefit</u>. Landlord and Tenant acknowledge that they enter into this Agreement for the express benefit of Franchisor and that Franchisor is an intended beneficiary hereof.
- 11. <u>Supremacy</u>. This Addendum shall control and supersede any inconsistent provision of the Lease.

The parties have signed this Agreement the day and year first above written.

| LANDLORD: | TENANT: |  |
|-----------|---------|--|
|           |         |  |
| By:       | By:     |  |
| Title:    | Title:  |  |

#### **EXHIBIT C**

#### AREA DEVELOPMENT AGREEMENT

## MRS. FIELDS® AREA DEVELOPMENT AGREEMENT

| THIS   | AREA DEVELOPMENT          | AGREEMENT ("Agi      | reement") made and | l entered into a | as of the | day        |
|--------|---------------------------|----------------------|--------------------|------------------|-----------|------------|
| of     |                           | _ ("Effective Date") | among Mrs. Fields  | Franchising,     | LLC, a    | Delaware   |
| limite | d liability company ("we" | or "us"), and        |                    |                  |           | <u> </u> • |
|        |                           |                      |                    |                  |           |            |

#### INTRODUCTION:

- A. We are the owner and developer of a system for establishing, operating, and promoting specialty snacks, bakery items, and other related products and services which are prepared in accordance with special methods and recipes ("Stores"). The Stores are operated under our valuable trade names, service marks and trademarks, including the service marks MRS. FIELDS® and MRS. FIELDS COOKIES® (the "Marks"), and pursuant to our distinctive plan for the establishment, operation and promotion of Stores and related system for doing business ("System").
- B. You desire to use our System in connection with the development of a certain number of Stores in a specific geographical area as stated herein. We are willing to grant you the right to establish and operate such Stores under the provisions contained below.

#### **AGREEMENTS**:

In consideration of the mutual covenants and agreements contained herein, the parties agree as follows:

- 1. Grant of Development Rights, Reservation of Rights.
- A. <u>Your Rights</u>. Subject to the provisions stated below, we hereby grant to you the right to establish and operate for your own account, but not to subfranchise, sublicense or resell, Mrs. Fields® franchised businesses (the "Franchised Businesses"), each pursuant to the thencurrent individual Mrs. Fields® franchise agreement ("Franchise Agreement"). Your rights to develop and establish Franchised Businesses under this Agreement will be limited to the geographic area described on <u>Exhibit A</u> attached hereto (the "Designated Area"). So long as you comply with the terms of this Agreement and comply with the terms of the individual Franchise Agreement for each Franchised Business developed under this Agreement, we will not, during the term of this Agreement, establish for our own account or franchise others to operate any Store under the Marks within the Designated Area other than to you pursuant to this Agreement.
- B. <u>Rights Reserved</u>. Section 1(A) represents your sole rights to the Designated Area. We (and our affiliates) expressly reserve all rights not granted to you by this Agreement, including the right to:
- (1) directly operate, or to grant other persons the right to operate Stores at locations outside the Designated Area;
- (2) promote, sell and distribute products and services authorized for sale at Stores under trademarks and service marks other than the Marks through similar or dissimilar channels of distribution inside or outside the Designated Area;
- (3) promote, sell and distribute products and services authorized for sale at Stores under the Marks through dissimilar channels of distribution, including by electronic means

such as the Internet, by mail order, catalog, retail store display, through the wholesale sale of products to unrelated retail outlets or distributors, or to grocery stores, convenience stores, or locations other than through Mrs. Fields Stores; and

- (4) to promote the System and Stores generally, including on the Internet (or any other existing or future form of one or more electronic commerce), and to create, operate, maintain and modify or discontinue the use of websites using the Marks.
- 2. <u>Development Fee.</u> For the rights described in Section 1 above, you will pay us: (i) an Initial Franchise Fee as provided in the Initial Franchise Agreement for the Franchised Business being developed pursuant to the Initial Franchise Agreement; and (ii) a non-refundable development fee of \_\_\_\_\_\_\_ Dollars (\$\_\_\_\_\_\_\_\_) for each of the remaining Franchised Businesses to be developed pursuant to this Agreement (the "Development Fee"). The fees paid to us under this Agreement are fully earned by us upon receipt and non-refundable. The Development Fee for each Franchised Business is credited against the applicable Initial Franchise Fee. The total Development Fee is described on the Development Schedule.
- 3. <u>Conditions to Develop Additional Franchised Businesses</u>. We will be obligated to enter into a Franchise Agreement for the development of a Franchised Business under this Agreement only if, at the time you intend to enter into a Franchise Agreement for such Store: (1) all amounts due and owing by you to us or our affiliates under or relating to the Initial Franchise Agreement or any Franchise Agreement are paid in full and you otherwise are in good standing under the Initial Franchise Agreement and any other Franchise Agreement; and (2) you are not in default for any reason stated in Section 7 below for which you have received written notice.
- 4. <u>Development Procedure</u>. Each Mrs. Fields® Franchised Business to be developed pursuant to this Agreement will be governed by the terms of the Franchise Agreement that we and you will sign for such Franchised Business, including all provisions applicable to site approval, lease approval, and development of the premises. You must sign a Franchise Agreement for each Store and pay the remaining portion of the Initial Franchise Fee at least four (4) months before the Store is scheduled to open or before you sign the lease for the Store, whichever occurs first. You will not develop any Mrs. Fields® Franchised Business at any site which we have not approved in writing and for which there is no Franchise Agreement between the parties.
- 5. <u>Development Schedule</u>. Your rights under this Agreement are conditioned upon your active development of the Designated Area. You agree to sign Franchise Agreements, and open for business and thereafter maintain in operation within the Designated Area not less than the number of Franchised Businesses within the time frames described in <u>Exhibit A</u> attached hereto (the "**Development Schedule**").
- 6. <u>Term.</u> Subject to Section 7 below, the term of this Agreement will commence on the Effective Date and expire earlier of: (a) the date the last Franchised Business required to be developed under this Agreement is open; or (b) the date the last Franchised Business required to be developed under this Agreement must be open under the Development Schedule. This Agreement is not renewable.

#### 7. Default and Termination.

A. You will be in default, and we may, at our option, terminate this Agreement if: (1) you fail to meet the Development Schedule stated herein, (2) you violate any other material provision of this Agreement, (3) you violate any material provision of the Initial Franchise Agreement or any Franchise Agreement, (4) you are declared bankrupt or becomes insolvent, or (5)

you attempt to subfranchise or sublicense in any manner all or part of your rights under this Agreement.

- B. Except as described below, you will have thirty (30) days, or such longer period as applicable law may require, after you receive from us a written Notice of Termination within which to remedy any default hereunder, and to provide evidence thereof to us. If you fail to correct the default within that time (or such longer period of time as applicable law may require), this Agreement will terminate without further notice to you effective immediately upon the expiration of the thirty (30) day period (or such longer period as applicable law may require).
- C. We may, at our option, terminate this Agreement immediately upon delivery of written notice to you if: (1) you fail to comply with one or more material requirements of this Agreement three (3) or more times during the term of this Agreement; (2) the nature of your breach is not curable; (3) any default under Sections (4) or (5) in Section 7(A) above; or (4) any default under the Initial Franchise Agreement or any Franchised Business Franchise Agreement for which we are not required to provide you any right to cure.
- D. During the period from the date we send a notice of default until you cure all violations and defaults specified therein or this Agreement is terminated, we will not be obligated to enter into any Franchised Business Franchise Agreement with you or otherwise perform any obligations pursuant to this Agreement. Upon termination or expiration of this Agreement, all rights licensed herein will automatically revert to us and your exclusive right to develop Mrs. Fields® Franchised Businesses within the Designated Area will cease. Termination or expiration of this Agreement will not affect your rights under any individual Franchised Business Franchise Agreements in effect at that time.
- E. Termination or expiration of this Agreement will represent a termination or expiration of the rights and license granted herein to you.

#### 8. Assignment.

- A. <u>By Us.</u> We may assign this Agreement and such assignment will benefit our successors and assigns. Any such assignment or transfer will require the assignee to fulfill our obligations under this Agreement.
- B. <u>By You</u>. You understand and acknowledge that the rights and duties created by this Agreement are personal to you and that we have granted this Agreement in reliance upon your individual or collective character, skill, aptitude, attitude, business ability, and financial capacity (or your owners). Neither this Agreement nor any part or all of your ownership may be voluntarily, involuntarily, directly or indirectly, assigned, sold, or otherwise transferred by you or your owners without our prior written consent. We may impose conditions to any proposed transfer or assignment, including the following:
- (1) You are in compliance with the terms of this Agreement, including the Development Schedule, each Franchise Agreement to which you are a party and all other agreements between the parties;
- (2) The proposed transferee has been approved by us as meeting our thencurrent standards and policies for multiple unit franchisees and, at our option, agrees to acquire all of your interest in each Franchised Business developed under this Agreement;

|                                                                                                                                                                                                                   | ice for the undeveloped franchises subject to this ent development fees we charge franchisees under an                                                                                                                                                                                                                                    |  |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| (4) The proposed tran                                                                                                                                                                                             | sferee has completed our training program;                                                                                                                                                                                                                                                                                                |  |
| (5) You pay us a trans<br>(\$); and                                                                                                                                                                               | sfer fee of Thousand Dollars                                                                                                                                                                                                                                                                                                              |  |
|                                                                                                                                                                                                                   | vner, if applicable) sign a general release, in form and claims against us and our affiliates, officers, directors, limited or prohibited by applicable law.                                                                                                                                                                              |  |
| consent as described in this Section 8(B), and may<br>Any assignment or transfer without our prior writte                                                                                                         | details related to, the conditions for transfer and our do so in our Operations Manual or otherwise in writing. In consent constitutes a breach and conveys no rights to hise Agreements may be transferred only pursuant to the                                                                                                          |  |
| rights to use the Marks is derived solely from the F<br>You agree that all use of the Marks and any goody<br>after termination or expiration of this Agreeme<br>Businesses you operate under individual Franchise | ou have no interest in or right to use the Marks. Any Franchise Agreements entered into between you and us. will established exclusively benefits us. You agree that nt, you will not, except with respect to Franchised Agreements, directly or indirectly, identify yourself or of, or otherwise associated with, Mrs. Fields or use in |  |
|                                                                                                                                                                                                                   | d any dispute arising hereunder, will be governed by ranchise Agreement including arbitration, governing law                                                                                                                                                                                                                              |  |
| relative to its subject and cannot be waived, altered writing by the parties. Notwithstanding the foregothe express representations we made in the franchi                                                        | d Exhibit A represent the entire Agreement of the parties I, or rescinded in whole or in part except by an expressing, nothing in this Agreement is intended to disclaim ise disclosure document that we furnished to you. The e invalidity or unenforceability of any of them will not                                                   |  |
| We and you have signed this Agreement as of the Effective Date.                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                           |  |
| "WE"                                                                                                                                                                                                              | "YOU"                                                                                                                                                                                                                                                                                                                                     |  |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                      | (Print Corporate Name)                                                                                                                                                                                                                                                                                                                    |  |
| By                                                                                                                                                                                                                | By                                                                                                                                                                                                                                                                                                                                        |  |

## EXHIBIT A TO DEVELOPMENT AGREEMENT

|                                                     | ched to and is an integral part of t                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | the Mrs. Fields® Area Development etween we and you.                                                                                             |
|-----------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------|
|                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and obligations granted to you under the the following described area (or in the map                                                             |
| \$ The Develop Franchise Agreement.  3. Development | ment Fee is in addition to the land of the | In the Franchised Businesses is Initial Franchise Fee due under the Initial Ity sign Franchise Agreements, pay permanent locations in compliance |
| FRANCHISED<br>BUSINESS NUMBER:                      | DATE BY WHICH<br>FRANCHISED BUSINESS<br>MUST OPEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | CUMULATIVE NUMBER OF<br>FRANCHISED BUSINESSES TO<br>BE OPEN                                                                                      |
|                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1                                                                                                                                                |
|                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 2                                                                                                                                                |
|                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 3                                                                                                                                                |
|                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 4                                                                                                                                                |
|                                                     | _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 5                                                                                                                                                |
| APPROVED:                                           | "YOU"                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                  |
| MRS. FIELDS FRANCHIS                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | rporate Name)                                                                                                                                    |
| By                                                  | By                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                  |
| Its                                                 | Its                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                  |

#### EXHIBIT D

#### FORM DEPOSIT AGREEMENT

#### STANDARD FORM DEPOSIT AGREEMENT

| Date:             | , 20                          |                                 |                        |
|-------------------|-------------------------------|---------------------------------|------------------------|
| Dear              | :                             |                                 |                        |
| This will acknow  | vledge receipt by Mrs. Field  | s Franchising, LLC ("Franchis   | sor") of the amount of |
| \$10,000 (the "I  | Deposit"), from               |                                 | ("Applicant"), in      |
| connection with   | the Applicant's intention to  | enter into a Franchise Agree    | ment with Franchisor   |
| under which Ap    | plicant would open and ope    | erate a Mrs. Fields® franchise  | ed store (the "Store") |
| within            | the                           |                                 | territory:             |
|                   |                               | ("Terri                         | tory"). The Deposit is |
| being paid in con | nsideration of Franchisor tak | king the Territory off the mark | et for a period of one |
| ~ .               |                               | ate") of this Agreement, and pr | <u>-</u>               |
| ` / •             | `                             | n approved location for the Sto | 0 11                   |

#### Refund of Deposit

Applicant's request to become a franchisee is subject to Franchisor's discretionary approval until such time as Franchisor approves Applicant as a franchisee and each of the parties signs a Franchise Agreement. Applicant may withdraw its request to be considered as a franchisee at any time, provided Applicant notifies Franchisor in writing that it wishes to withdraw its request to be considered as a franchisee prior to signing a Franchise Agreement with Franchisor. Franchisor will not refund any portion of the Deposit, which Franchisor will retain for its expenses in considering Applicant's application, providing real estate site selection services support and in consideration of lost opportunity if: (i) Applicant notifies Franchisor of its intent to withdraw its request prior to signing a Franchise Agreement; (ii) Franchisor locates a site within the Territory that meets Franchisor's approval, and Applicant refuses to accept such site; or (iii) the term of this Agreement expires. In such event, Franchisor also may terminate this Agreement upon ten (10) days' written notice to Applicant and place the Territory back on the market to offer to other applicants.

Franchisor has the sole and absolute right to deny Applicant's request to become a franchisee at any time before Franchisor signs the Franchise Agreement. If Franchisor disapproves Applicant's request to become a franchisee, Franchisor will refund the Deposit of \$10,000.

If Applicant is approved as a franchisee and has not previously requested a withdrawal of its request for approval, Applicant will sign the Franchise Agreement that Franchisor submits to Applicant.

Franchisor and Applicant agree that if Franchisor and Applicant sign a Franchise Agreement granting Applicant the right to establish a Mrs. Fields Store within the Territory, Applicant will receive a credit in the amount of the Deposit against any Initial Franchise Fee due under such Franchise Agreement.

| Please return the counter-signed original of this letter to Franchisor, along with the Deposit, to indicate your agreement with the terms of this Deposit Agreement. |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Sincerely,                                                                                                                                                           |
| Name: Title:                                                                                                                                                         |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                         |
| Agreed to this day of, 20                                                                                                                                            |
| APPLICANT:                                                                                                                                                           |
| By:                                                                                                                                                                  |
| Name:                                                                                                                                                                |
| Title:                                                                                                                                                               |

#### EXHIBIT E

#### TABLE OF CONTENTS OF OPERATIONS MANUAL





| MANUAL SECTION                    |    |  |
|-----------------------------------|----|--|
| Introduction                      |    |  |
| History                           | 1  |  |
| Vision, Mission, Values           | 2  |  |
| Brand Blueprint                   | 4  |  |
| Company Principles                | 5  |  |
| Beverages                         |    |  |
| Coca-Cola®                        | 1  |  |
| Icee®                             | 7  |  |
| Breezer Frozen Beverages®         | 9  |  |
| Cookies 'n' Cream Chiller         | 14 |  |
| Breezer Quick Reference           | 16 |  |
| Lemonade                          | 17 |  |
| Iced Coffees                      | 20 |  |
| Dillano's Coffee                  | 26 |  |
| Keurig Coffee                     | 27 |  |
| Brand Resources                   |    |  |
| Franchise Support Center Contacts | 1  |  |
| Approved Vendors                  | 2  |  |
| Websites                          | 3  |  |
| Navigator                         | 4  |  |
| Distribution Centers TCBY         | 6  |  |
| Distribution Centers Mrs. Fields  | 7  |  |
| Media Contact                     | 8  |  |
| Breakfast Items                   |    |  |
| Bagels                            | 1  |  |
| Muffins                           | 2  |  |
| Streusel                          | 5  |  |
| Pastries - Croissants             | 6  |  |
| Brownies                          |    |  |
| Fudge Brownies in Bakable Tray    | 1  |  |
| Fudge Brownie Trouble Shooting    | 5  |  |
| Preparing Crumb Crust             | 6  |  |





| MANUAL SECTION                                    |    |
|---------------------------------------------------|----|
| Peanut Butter Dreambars                           | 8  |
| Rocky Mountain Moguls                             | 10 |
|                                                   | 13 |
| Cutting Brownies                                  |    |
| Frosted Brownies                                  | 14 |
| Fudge Brownie Trouble Shooting Guide              | 16 |
| Cookies & Nibblers                                |    |
| Cookie & Nibbler Standards                        | 1  |
| Baking Classic Cookie Standards                   | 4  |
| Baking Nibbler Bite Sized Cookies                 | 6  |
| Frosted Sugar Butter Cookies                      | 8  |
| Milk Chocolate Chip with M&M's                    | 9  |
| Peanut Butter Milk Chocolate Chip                 | 11 |
| Sugar Butter with M&M's                           | 13 |
| Sugar Butter with Cinnamon Sugar or Sanding Sugar | 15 |
| Milk Chocolate Chip with Walnuts                  | 17 |
| Specialty - Cookie Cups                           | 19 |
| Specialty - Cookie Cups - Mini                    | 20 |
| Specialty - Cookie Sandwiches                     | 21 |
| Specialty - Hand Dipped Cookies                   | 22 |
| Mega Cookie                                       | 24 |
| Cookie Size Guide                                 | 27 |
| Nibbler Size Guide                                | 29 |
| Cookie Trouble Shooting Guide                     | 30 |
| Core Line Up                                      |    |
| Required Products Core                            | 1  |
| Optional Products                                 | 2  |
| Criteria to Expand                                |    |
| Criteria to Expand                                | 1  |
| Guidelines                                        | 2  |
| Performance Rating System                         | 3  |
| Rating Definitions                                | 3  |
|                                                   |    |



| MANUAL SECTION                                  |    |
|-------------------------------------------------|----|
| "C" Phase                                       | 4  |
| "C" Process                                     | 5  |
| Notice of Intent to Default & Notice of Default | 6  |
| Termination Notices                             | 6  |
| <b>Customer Service</b>                         |    |
| Benchmarks of Customer Service                  | 1  |
| The Greet, Treat, Complete Model                | 2  |
| Telephone Orders                                | 3  |
| Identifying Customer Types                      | 4  |
| Service Recovery                                | 5  |
| Customer Requests and Complaints                | 6  |
| Sampling                                        | 6  |
| Strategies for Increasing Your Average Sale     | 7  |
| Increasing Sales                                | 8  |
| Increasing Sales with Sampling                  | 9  |
| Typical Challenges of Restaurant Service        | 12 |
| Handling Customer Complaints from Navigator     | 12 |
| Equipment                                       |    |
| Freezer Standards                               | 1  |
| Refrigerator                                    | 6  |
| Freezer and Refrigerator Troubleshooting        | 7  |
| Condenser Coils                                 | 8  |
| Ice Machine                                     | 9  |
| Oven Programming: Lang and Blodgett Models      | 11 |
| Heating and Air Conditioning                    | 18 |
| Plumbing                                        | 19 |
| Stem Thermometers                               | 21 |
| Blenders                                        | 22 |
| Airpot Brewer                                   | 23 |
| Fountain Beverage Machine                       | 25 |
| Yesterday's Batch                               |    |
| Yesterday's Batch Standards                     | 1  |



| MANUAL SECTION                            |    |
|-------------------------------------------|----|
| Yesterday's Packaging and Displaying      | 2  |
| Financial                                 |    |
| Financial Management Programs Disclaimer  | 1  |
| Introduction                              | 2  |
| Financial Terms Glossary                  | 3  |
| Understanding P&L Statements              | 6  |
| Sales Curve                               | 7  |
| Scheduling                                | 8  |
| Mandatory Records and Record Keeping      | 8  |
| Recommended Records and Record Keeping    | 15 |
| Food Safety                               |    |
| Introduction                              | 1  |
| Six Common Types of Foodborne Illnesses   | 2  |
| Personal Hygiene                          | 3  |
| Hand Washing                              | 4  |
| Critical Control Points                   | 5  |
| Critical Control Points Chart TCBY        | 7  |
| Critical Control Points Chart Mrs. Fields | 10 |
| Pest Control                              | 13 |
| Cleaning Procedures                       | 16 |
| Food Safety Training                      | 19 |
| Labor / Staffing                          |    |
| Staffing and Forecasting                  | 1  |
| Labor Management Checklist                | 2  |
| Defining Needs                            | 3  |
| What Can You Offer?                       | 4  |
| Staffing Plan                             | 5  |
| Recruiting                                | 6  |
| Interviewing                              | 7  |
| Reference Checks                          | 11 |
| Job Offer                                 | 12 |
| Team Member Job Description               | 13 |





| MANUAL CECTION                                      |    |
|-----------------------------------------------------|----|
| MANUAL SECTION                                      | 14 |
| Manager Job Description                             | 14 |
| Train and Retain                                    | 16 |
| Motivation                                          | 17 |
| Coaching                                            | 22 |
| Legal Compliance                                    |    |
| Introduction                                        | 1  |
| Development of Your Store/Maintaining Your Premises | 2  |
| Obligations Relating to Operations                  | 4  |
| Fees                                                | 7  |
| Marketing and Promotions                            | 9  |
| Use of The Marks and Confidential Information       | 11 |
| Transfers                                           | 12 |
| Default and Termination                             | 14 |
| Communication from your Franchisor                  | 17 |
| Loss Prevention                                     |    |
| Store & Product Security                            | 1  |
| Cash Handling                                       | 1  |
| Waste Control                                       | 9  |
| Ways to Control Costs                               | 11 |
| Marketing                                           |    |
| Marketing Terms                                     | 1  |
| Merchandising                                       | 2  |
| Packaging Standards                                 | 5  |
| Store image                                         | 6  |
| Sampling                                            | 8  |
| Marketing Emails                                    | 10 |
| Microsites                                          | 11 |
| Social Media                                        | 12 |
| Grand Opening                                       | 26 |
| Online Ordering                                     |    |
| Welcome to Online Ordering                          | 1  |
|                                                     |    |





| MANUAL GEORGIAN                                                                                                                                                                      |                            |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------|
| MANUAL SECTION                                                                                                                                                                       |                            |
| Accessing Your Store Manager Portal                                                                                                                                                  | 1                          |
| Navigating the Store Manager                                                                                                                                                         | 2                          |
| Managing Your Online Ordering Site                                                                                                                                                   | 2                          |
| My User Profile                                                                                                                                                                      | 3                          |
| Store Selection                                                                                                                                                                      | 4                          |
| Store User                                                                                                                                                                           | 4                          |
| Orders                                                                                                                                                                               | 6                          |
| Store Hours                                                                                                                                                                          | 6                          |
| Social Media Links                                                                                                                                                                   | 7                          |
| Pickup Times                                                                                                                                                                         | 8                          |
| Delivery                                                                                                                                                                             | 8                          |
| Products                                                                                                                                                                             | 9                          |
| Prep Your POS for Online Ordering                                                                                                                                                    | 13                         |
| Manage Online Orders on Your POS                                                                                                                                                     | 14                         |
| Open & Close                                                                                                                                                                         |                            |
| Store Daily/Weekly Operational Procedures                                                                                                                                            | 1                          |
| Opening Checklist                                                                                                                                                                    | 2                          |
| Closing Checklist                                                                                                                                                                    | 3                          |
| Weekly Checklists                                                                                                                                                                    | 4                          |
| Monthly Checklists                                                                                                                                                                   | 5                          |
| Placing and Receiving Orders                                                                                                                                                         |                            |
| Placing Orders                                                                                                                                                                       | 1                          |
| Receiving Orders                                                                                                                                                                     | 2                          |
| Tecolving Orders                                                                                                                                                                     | 2                          |
| Storing Orders                                                                                                                                                                       | 3                          |
|                                                                                                                                                                                      |                            |
| Storing Orders                                                                                                                                                                       | 3                          |
| Storing Orders  Defective Product Guidelines                                                                                                                                         | 3 4                        |
| Storing Orders  Defective Product Guidelines  First Order Guide                                                                                                                      | 3<br>4<br>5                |
| Storing Orders  Defective Product Guidelines  First Order Guide  Smallwares and Equipment Ordering                                                                                   | 3<br>4<br>5<br>6           |
| Storing Orders  Defective Product Guidelines  First Order Guide  Smallwares and Equipment Ordering  Tundra Credit Card Information                                                   | 3<br>4<br>5<br>6<br>7      |
| Storing Orders  Defective Product Guidelines  First Order Guide  Smallwares and Equipment Ordering  Tundra Credit Card Information  Tundra Frequently Asked Questions                | 3<br>4<br>5<br>6<br>7      |
| Storing Orders  Defective Product Guidelines  First Order Guide  Smallwares and Equipment Ordering  Tundra Credit Card Information  Tundra Frequently Asked Questions  Point of Sale | 3<br>4<br>5<br>6<br>7<br>8 |



| MANUAL SECTION                             |    |
|--------------------------------------------|----|
| Manager Procedures                         | 6  |
| Mrs. Fields Gift Card Program              | 7  |
| Setting Up Gift Cards                      | 8  |
| Selling a Gift Card                        | 9  |
| Frequently Asked Questions                 | 10 |
| Safety and Security (cont.)                |    |
| Statement of Policy on Safety              | 1  |
| New Team Member Orientation Policy         | 2  |
| Substance Abuse                            |    |
| General Safety Rules                       | 3  |
| Chemical Safety Hazard Communication       | 4  |
| Ongoing Safety Training                    | 5  |
| Safety/Housekeeping Inspections            | 5  |
| Accident Investigation/Injury Reporting    | 7  |
| Safe Work Practices & Procedures           | 8  |
| Workplace Fire Safety                      | 9  |
| Burns                                      | 10 |
| Slips, Trips, and Falls                    | 10 |
| Preventing Sprains & Strains               | 11 |
| Ladders                                    | 11 |
| Proper Lifting Techniques                  | 12 |
| Proper Chemical Storage                    | 13 |
| CO <sub>2</sub> Changing & Storage         | 13 |
| Proper Housekeeping                        | 14 |
| Prevent Cuts & Burns                       | 15 |
| Storage                                    | 15 |
| Utensils and Tools                         | 1  |
| Customer Accidents                         | 16 |
| Repetitive Motion                          | 17 |
| Electrical                                 | 17 |
| Seasonal Flu & Influenza Preparedness Plan | 18 |
| Sales Reporting                            |    |





| MANUAL SECTION                                        |            |  |  |  |
|-------------------------------------------------------|------------|--|--|--|
| Introduction                                          | 1          |  |  |  |
| Sales Reporting Guidelines                            | 2          |  |  |  |
| Using the Website                                     | 4          |  |  |  |
| Standards                                             |            |  |  |  |
| Communication Board                                   | 1          |  |  |  |
| Restroom Standards                                    | 2          |  |  |  |
| Floor standards                                       | 2          |  |  |  |
| Inventory Standards                                   | 2          |  |  |  |
| Refrigerator Standards                                | 3          |  |  |  |
| Freezer Standards                                     | 3          |  |  |  |
| Back Room Standards                                   | 4          |  |  |  |
| Dinning Room Standards                                | 5          |  |  |  |
| Uniform Standards                                     | 6          |  |  |  |
| <b>Quick Reference Sheets: Frosted Specialty Item</b> | s          |  |  |  |
| Cookie Card Designs                                   | Page 4-5   |  |  |  |
| Mini Cookie Cake Designs                              | Page 6-7   |  |  |  |
| Cookie Slice Designs                                  | Page 8-9   |  |  |  |
| Regular and Mini Cookie Cups Designs                  | Page 10-11 |  |  |  |
| Sugar Cookie Designs                                  | Page 12-13 |  |  |  |
| Frosted Brownie Designs                               | Page 14-15 |  |  |  |
| Cookie Sandwich Designs                               | Page 16    |  |  |  |
| 9" Heart Cookie Cakes                                 | Page 17-18 |  |  |  |
| <b>Quick Reference Sheets: Cookie Cakes</b>           |            |  |  |  |
| Message Only Cookie Cakes                             | Page 19    |  |  |  |
| Birthday Cookie Cakes                                 | Page 20-21 |  |  |  |
| Special Occasion Cookie Cakes                         | Page 22-23 |  |  |  |
| Holiday Cookie Cakes                                  | Page 24-26 |  |  |  |
| Baking and Decorating Procedures                      |            |  |  |  |
| Big Cookie Cake Introduction                          | Page 27    |  |  |  |
| General Messages                                      | Page 28-29 |  |  |  |
| Pre-Pressed Big Cookie Cakes                          | Page 30    |  |  |  |
| Hand-Pressed Big Cookie Cakes                         | Page 31-32 |  |  |  |



| MANUAL SECTION                                                      |             |
|---------------------------------------------------------------------|-------------|
| Cookie Card Productions                                             | Page 33     |
| Cookie Slice Production                                             | Page 34     |
| Mini Cookie Cake Production                                         | Page 35     |
| 9" Big Cookie Cake                                                  | Page 36     |
| Mixing Frosting Colors                                              | Page 37-38  |
| Pastry Bag Preparation                                              | Page 39     |
| Pastry Bag Holding & Care                                           | Page 40     |
| Decorating Borders                                                  | Page 41-42  |
| Decorating Writing                                                  | Page 43-45  |
| Stencils                                                            | Page 46     |
| Decorating Tips                                                     | Page 47-48  |
| Double Layer Big Cookie Cakes                                       | Page 49     |
| Packaging Big Cookie Cakes                                          | Page 50     |
| Order Pad                                                           | Page 50     |
| Cookie Cake Counter Mat                                             | Page 51     |
| Cookie Cake Brochure                                                | Page 51     |
| Microsite Designs on the POS                                        | Page 52     |
| Store Tablet                                                        | Page 52     |
| Cookie Cake Design S&P's                                            |             |
| Design PLU Numbers and Names - Designs are listed in order by PLU # | Page 53-181 |

#### EXHIBIT F

#### LIST OF CURRENT AND FORMER FRANCHISEES

#### MRS. FIELDS CURRENT FRANCHISEES

| Franchisee Owner Name        | Address                     | City              | State      | Phone Number   |
|------------------------------|-----------------------------|-------------------|------------|----------------|
| John Ploen                   | 1301 Gulf Shores Pkwy       | Gulf Shores       | Alabama    | (251) 968-8229 |
| Mohammad Ashraf              | 3111 WA. Chandler Blvd K204 | Chandler          | Arizona    | (480) 931-5696 |
| Narinder   Wife/MGR          | 72363 Baker Blvd            | Baker             | California | (760) 733-4505 |
| Dave Davis                   | 2701 Ming Ave               | Bakersfield       | California | (661) 832-8166 |
| Subhash Chand                | 2116 Brea Mall              | Brea              | California | (714) 255-1429 |
| Mansoor Ghatila              | 201 East Magnolia Blvd      | Burbank           | California | (818) 559-8662 |
| Quishon Walker               | 1233 W. Rancho Vista Blvd   | Palmdale          | California | (661) 526-4158 |
| Fatimah Safari               | 20700 S Avalon Blvd         | Carson            | California | (310) 324-4682 |
| Gulam Nabi                   | 239 Los Cerritos Center     | Cerritos          | California | (562) 410-1299 |
| Renee                        | 109 Sun Valley Mall         | Concord           | California | (925) 825-1780 |
| Frank Fu                     | 6000 Sepulveda Blvd         | Culver City       | California | (310) 398-2001 |
| Mike Mendoza                 | 321 Stonewood St            | Downey            | California | (562) 923-1896 |
| Sodexo                       | 3300 Broadway               | Eureka            | California | (707) 444-2130 |
| Cindy   Gee                  | 1350 Travis Blvd            | Fairfield         | California | (707) 429-4495 |
| Haroon Javed                 | 347 Southland Dr            | Hayward           | California | (510) 887-5606 |
| Sung Chang                   | 112 Great Mall Dr           | Milpitas          | California | (408) 649-3258 |
| Jong Sohn                    | 9301 Tampa Ave              | Northridge        | California | (818) 407-0404 |
| Jong Sohn                    | 72840 Highway 111           | Palm Desert       | California | (760) 346-4598 |
| Yousuf Nabi                  | 12505 N. Mainstreet #3265   | Rancho Cucamonga  | California | (909) 226-6055 |
| Andy Kurum                   | 925 Blossom Hill Road #9116 | San Jose          | California | (408) 550-6105 |
| Yosuf Nabi                   | 1 Mills Circle K116         | Ontario           | California | (909) 413-1761 |
| Kulbir Kaur                  | 1170 Galleria at Tyler      | Riverside         | California | (951) 354-5554 |
| Matt                         | 1151 Galleria Blvd          | Roseville         | California | (916) 878-5418 |
| Jong Sohn   MGR Jocelle      | 1150 El Camino Real         | San Bruno         | California | (650) 871-9688 |
| Yui Man Mak                  | 111 West Harbor Dr          | San Diego         | California | (619) 818-4469 |
| Cherie Coulter-Weith         | Pier 39 Bldg. B-06          | San Francisco     | California | (415) 989-1588 |
| Mehrad Lashkari Nejad        | 453 Hillsdale Mall          | San Mateo         | California | (650) 573-6481 |
| Harshad                      | 2800 N Main St              | Santa Ana         | California | (714) 972-1128 |
| AJ Jolly                     | 2855 Stevens Creek Blvd     | Santa Clara       | California | (408) 249-2489 |
| Claire Aquino                | 378 W Hillcrest Dr          | Thousand Oaks     | California | (805) 496-7602 |
| Dave Smith                   | 3301 E Main St              | Ventura           | California | (805) 658-8386 |
| Anubhav Jolly                | 603 Plaza Dr                | West Covina       | California | (626) 960-7011 |
| Gulam Nabi                   | 100 W. Broadway #G002       | Glendale          | California | (818) 919-9691 |
| Quishon Walker               | 24201 Valencia Blvd         | Valencia          | California | (661) 860-7998 |
| Julia Kim-Park               | 8505 Park Meadows Center Dr | Lonetree          | Colorado   | (303) 768-0842 |
| Abdul Mannan                 | 1531 Christiana Mall        | Newark            | Delaware   | (302) 283-1346 |
| David Duvall & Mike Dargenio | 451 E Altamonte Dr          | Altamonte Springs | Florida    | (407) 644-1400 |
| Mohammed Alamgir             | 19575 Biscayne Blvd         | Aventura          | Florida    | (305) 932-3310 |
| Tony Nguyen                  | 1700 W New Haven Ave        | Melbourne         | Florida    | (321) 727-7773 |

| Franchisee Owner Name     | Address                                 | City               | State         | Phone Number   |
|---------------------------|-----------------------------------------|--------------------|---------------|----------------|
| Mohammed Vaid             | 4200 Conroy Rd #239                     | Orlando            | Florida       | (407) 363-0670 |
| Hatim Hashwani            | 8001 S. Orange Blossom Trail #1206      | Orlando            | Florida       | (321) 247-5693 |
| Craig Ferrier             | 3101 PGA Blvd                           | Palm Beach Gardens | Florida       | (561) 626-4014 |
| Jerry Calvo               | 11401 Pines Blvd                        | Pembroke Pines     | Florida       | (954) 437-9705 |
| Paul Duback Street Eatery | Hwy 400 South                           | Dawsonville        | Georgia       | (706) 216-4131 |
| Yousuf Nabi               | 1450 Ala Moana Blvd #1063               | Honolulu           | Hawaii        | (808) 494-0929 |
| Kandie & Mark Dunning     | 1948 West Pullman Rd                    | Moscow             | Idaho         | (208) 883-1817 |
| Lomesh Amin               | 1650 Premium Outlets Rd                 | Aurora             | Illinois      | (630) 898-9909 |
| Lomesh Amin               | 5 Woodfield Mall #G119                  | Schaumburg         | Illinois      | (309) 663-8231 |
| Dave Thakkar              | 500 West Madison St                     | Chicago            | Illinois      | (312) 655-1931 |
| Peter Mario Spina         | 1905 Franciscan Way                     | Chicago            | Illinois      | (630) 520-0009 |
| Hassan Ismail             | 444 Chicago Ridge Mall Dr               | Chicago Ridge      | Illinois      | (708) 422-0123 |
| Peter Mario Spina         | 1700 N State St                         | Elgin              | Illinois      | (847) 741-5784 |
| Peter Mario Spina         | 1166 East State St                      | Geneva             | Illinois      | (630) 845-9300 |
| Peter Mario Spina         | 149 E Ogden Ave                         | Hinsdale           | Illinois      | (630) 455-5841 |
| Peter Mario Spina         | 3100 South Cicero                       | Cicero             | Illinois      | (708) 652-9025 |
| Peter Mario Spina         | 2095 Quentin Rd                         | Kildeer            | Illinois      | (847) 550-8392 |
| Necip Kandemir            | 121 Yorktown Center                     | Lombard            | Illinois      | (630) 627-9070 |
| Peter Mario Spina         | 1503 N Aurora Rd                        | Naperville         | Illinois      | (630) 995-3243 |
| Lomesh Amin               | 7501 W Cermak Rd                        | North Riverside    | Illinois      | (708) 447-5938 |
| Lomesh Amin               | 30 Oakbrook Center                      | Oak Brook          | Illinois      | (630) 368-1002 |
| Hassan Ismail             | 412 Orland Square Dr                    | Orland Park        | Illinois      | (708) 364-0123 |
| Peter Mario Spina         | 1400 W Algonquin Rd                     | Palatine           | Illinois      | (847) 907-9479 |
| Ayman Abdulhadi           | 7200 Harrison Ave #15                   | Rockford           | Illinois      | (815) 332-6707 |
| Peter Mario Spina         | 25W 420 Geneva Rd                       | Wheaton            | Illinois      | (630) 462-1300 |
| Lomesh Amin               | 5 Woodfield Mall F305A                  | Schaumburg         | Illinois      | (224) 520-8117 |
| Sammy Zabaneh             | 11531 W 95th St                         | Overland Park      | Kansas        | (913) 492-0767 |
| John Ploen                | 1704 East Bert Kouns Loop Industrial Dr | Shreveport         | Louisiana     | (318) 797-8229 |
| Craig Schwartz            | 1736 Gause Blvd E                       | Slidell            | Louisiana     | (985) 288-0340 |
| Thompson Sunderaj         | 50 Holyoke Street                       | Holyoke            | Massachusetts | (413) 322-9561 |
| Juliana Jong              | 257 Briarwood Cir                       | Ann Arbor          | Michigan      | (734) 996-0644 |
| Gus Shukeireh             | 4262 Baldwin                            | Auburn Hills       | Michigan      | (248) 636-1112 |
| Gus Shukeireh             | 18900 Michigan Ave                      | Dearborn           | Michigan      | (313) 271-9750 |
| Kamaljeet Batth           | 27500 Novi Rd                           | Novi               | Michigan      | (248) 380-5077 |
| Sumeer Dosanjh            | 2800 W Big Beaver Rd #Y321              | Troy               | Michigan      | (248) 614-5350 |
| Yousuf Nabi               | 1300 W Sunset Rd                        | Henderson          | Nevada        | (702) 547-6196 |
| Ted Yang                  | 4300 Meadows Ln #133A                   | Las Vegas          | Nevada        | (702) 877-9165 |
| Jerry Whitsett            | South Gate PO Box 11033 Terminal D      | Las Vegas          | Nevada        | (702) 837-8654 |
| Jerry Whitsett            | PO Box 11033 Terminal B                 | Las Vegas          | Nevada        | (702) 261-5858 |
| Gulam Nabi                | 3200 Las Vegas Blvd                     | Las Vegas          | Nevada        | (702) 866-0627 |

| Franchisee Owner Name        | Address                           | City            | State             | Phone Number   |
|------------------------------|-----------------------------------|-----------------|-------------------|----------------|
| Bill Bori                    | 146 Mulberry St                   | Newark          | New Jersey        | (609) 348-2800 |
| Bill Bori                    | 1 Scarlet Knight Way              | Piscataway      | New Jersey        | (609) 348-2800 |
| Bill Bori                    | 1 MetLife Stadium Dr              | East Rutherford | New Jersey        | (609) 348-2800 |
| Hesham El-Dewak              | 1053 Broadway                     | Bayonne         | New Jersey        | (201) 354-9272 |
| Christopher Garrow           | 2801 Pacific Ave                  | Atlantic City   | New Jersey        | (732) 577-1470 |
| Christopher Garrow           | 2301 Boardwalk                    | Atlantic City   | New Jersey        | (732) 577-1470 |
| Christopher Garrow           | 1 Convention Blvd                 | Atlantic City   | New Jersey        | (742) 933-8595 |
| Jacob Geffner                | 204 W. Englewood Ave              | Teaneck         | New Jersey        | (609) 373-2046 |
| Jacob Geffner                | 1 American Dream Way #A107        | East Rutherford | New Jersey        | (201) 448-7225 |
| Anne Sum                     | 3710 Route 9 South                | Freehold        | New Jersey        | (732) 577-1470 |
| Mohammed Aslam               | 8801 Queens Blvd                  | Elmhurst        | New York          | (718) 699-0780 |
| Jui Trivedi                  | One Walden Galleria               | Buffalo         | New York          | (716) 683-2192 |
| Sanjiv Chand                 | 458 7th Ave                       | New York        | New York          | (212) 594-5682 |
| Jui Trivedi                  | 7979 Pittsford-Victor Rd          | Victor          | New York          | (585) 223-6400 |
| Jui Trivedi                  | 665 Route 318                     | Waterloo        | New York          | (315) 539-2700 |
| Thompson Sunderaj            | 1 Crossgates Mall #P114           | Albany          | New York          | (518) 918-6255 |
| Jacob Geffner                | 2655 Richmond Ave                 | Staten Island   | New York          | (917) 607-7407 |
| Mohammad Aslam               | 630 Old Country Road              | Garden City     | New York          | (516) 695-1178 |
| Rakesh Sethi                 | 4325 Glenwood Ave                 | Raleigh         | North<br>Carolina | (919) 783-5702 |
| Samuel Batt                  | 11025 Carolina Place Parkway D-29 | Pineville       | North<br>Carolina | (980) 299-0003 |
| Vijay Siripurapu             | 4314 Milan Rd                     | Sandusky        | Ohio              | (419) 624-8383 |
| Terry (Susan-Wife) Lynn Duff | 4901 N Kicapoo St                 | Shawnee         | Oklahoma          | (405) 275-0434 |
| Dhruptipatel Patel           | 160 North Gulph Rd                | King Of Prussia | Pennsylvania      | (610) 265-9034 |
| Amita Jadeja                 | 200 Monroeville Mall              | Monroeville     | Pennsylvania      | (412) 372-6227 |
| Lauren Spindler              | 1900 Pine St                      | Abilene         | Texas             | (325) 670-2910 |
| Jake Tortorice               | 4440 Dowlen Rd                    | Beaumont        | Texas             | (409) 347-3400 |
| Jake Tortorice               | 3080 College St                   | Beaumont        | Texas             | (409) 212-7777 |
| Jake Tortorice               | 2596 Calder Ave                   | Beaumont        | Texas             | (409) 832-4342 |
| Clay Swindle                 | 8687 North Central Expy           | Dallas          | Texas             | (214) 750-5919 |
| Frank Shaikh                 | 6002 Slide Road Kiosk #116        | Lubbock         | Texas             | (806) 778-2099 |
| Jake Tortorice               | 3504 Highway 365                  | Nederland       | Texas             | (409) 729-7267 |
| Jake Tortorice               | 6915 Cypresswood F                | Spring          | Texas             | (281) 251-7267 |
| Blane Smith                  | 2070 Layton Hills Mall            | Layton          | Utah              | (609) 373-2046 |
| Lomesh (Dipesh) B Amin       | 5300 S 76th St                    | Greendale       | Wisconsin         | (414) 421-8849 |

#### MRS. FIELDS AGREEMENTS SIGNED BUT OUTLETS NOT OPENED

| Franchisee Name | Address                       | City    | State | Phone Number   | # of Agreements |
|-----------------|-------------------------------|---------|-------|----------------|-----------------|
| Faisal Shaikh   | 3911 126 <sup>th</sup> Street | Lubbock | TX    | (801) 946-1637 | 1               |

# FORMER FRANCHISEES WHO HAVE HAD AN OUTLET TERMINATED, CANCELLED, NOT RENEWED, OR OTHERWISE VOLUNTARILY OR INVOLUNTARILY CEASED TO DO BUSINESS UNDER THE FRANCHISE AGREEMENT AND FRANCHISEES THAT HAVE NOT COMMUNICATED WITH US WITHIN 10 WEEKS OF THE ISSUANCE DATE OF THIS DISCLOSURE DOCUMENT

| Franchisee Name   | City           | State      | Phone Number   |
|-------------------|----------------|------------|----------------|
| Chris French      | Anchorage      | Alaska     | (907) 350-8439 |
| Jong Sohn         | Daly City      | California | (650) 255-0868 |
| Subhash Chand     | Angeles        | California | (310) 562-8357 |
| Dave Smith        | Sacramento     | California | (916) 416-0369 |
| Thompson Sunderaj | Pembroke Pines | Florida    | (305) 433-0383 |
| Jeff Sheets       | Valparaiso     | Indiana    | (219) 738-2921 |
| Jeff Sheets       | Merrillville   | Indiana    | (219) 718-5333 |
| Sam Gerges        | Portland       | Maine      | (207) 939-4633 |
| Yong Kyu Kim      | Hanover        | Maryland   | (703) 732-9331 |
| Rick Pavia        | Missouri       | Missouri   | (440) 617-8931 |
| Sahlee Cabo       | Las Vegas      | Nevada     | (808) 542-6974 |
| Vijay Siripurapu  | Niles          | Ohio       | (412) 436-6010 |
| Lorena Jordan     | El Paso        | Texas      | (915) 873-4953 |

#### **TRANSFERS**

| Franchisee Name     | City            | State      | Phone Number   |
|---------------------|-----------------|------------|----------------|
| Usman Siddiqui      | North Riverside | Illinois   | (708) 447-5938 |
| Lomesh Amin         | Chicago         | Illinois   | (312) 655-1931 |
| Anthony Di Ambrosio | Freehold        | New Jersey | (732) 577-1470 |

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

#### EXHIBIT G

#### STATE ADMINISTRATORS / AGENTS FOR SERVICE OF PROCESS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for the franchising disclosure/registration laws. We may not yet be registered to sell franchises in any or all of these states.

#### **CALIFORNIA**

Department of Financial Protection & Innovation:

Toll Free: 1 (866) 275-2677

#### Los Angeles

Suite 750 320 West 4th Street Los Angeles, California 90013 (213) 576-7505

Sacramento

2101 Arena Blvd. Sacramento, CA 95834 (916) 445-7205

San Diego

1455 Frazee Road, Suite 315 San Diego, California 92108 619) 525-4233

#### San Francisco

One Sansome Street, Suite 600 San Francisco, California 94104 (415) 972-8559

#### **HAWAII**

(state administrator)

Business Registration Division Department of Commerce and Consumer Affairs P.O. Box 40 Honolulu, Hawaii 96810 (808) 586-2722

(agent for service of process)

Commissioner of Securities of the Department of Commerce and Consumer Affairs 335 Merchant Street Honolulu, Hawaii 96813 (808) 586-2722

#### **ILLINOIS**

Franchise Bureau Office of the Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465

#### **INDIANA**

(state administrator)

Indiana Secretary of State Securities Division, E-111 302 West Washington Street Indianapolis, Indiana 46204 (317) 232-6681

(agent for service of process)

Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, Indiana 46204 (317) 232-6531

#### **MARYLAND**

(state administrator)

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

(agent for service of process)

Maryland Securities Commissioner at the Office of the Attorney General Securities Division 200 St. Paul Place
Baltimore, Maryland 21202-2021
(410) 576-6360

#### **MICHIGAN**

(state administrator)

Michigan Attorney General's Office Consumer Protection Division Attn: Franchise Section G. Mennen Williams Building, 1st Floor 525 West Ottawa Street Lansing, Michigan 48933 (517) 335-7567

(agent for service of process)

Michigan Department of Commerce, Corporations and Securities Bureau P.O. Box 30054 6546 Mercantile Way Lansing, Michigan 48909

#### **MINNESOTA**

(state administrator)

Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600

(agent for service of process)

Minnesota Commissioner of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651)-539-1600

#### **NEW YORK**

(state administrator)

NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21<sup>st</sup> Floor New York, New York 10005 (212) 416-8222Phone (212) 416-6042 Fax

(agent for service of process)

New York Department of State One Commerce Plaza 99 Washington Avenue, 6<sup>th</sup> Floor Albany, NY 12231-0001 (518) 473-2492

#### NORTH DAKOTA

North Dakota Securities Department 600 East Boulevard Avenue State Capitol - Fifth Floor Bismarck, North Dakota 58505 (701) 328-4712

#### **OREGON**

Oregon Division of Financial Regulation 350 Winter Street NE, Suite 410 Salem, Oregon 97301 (503) 378-4140

#### **RHODE ISLAND**

Department of Business Regulation Division of Securities John O. Pastore Complex Building 69-1 1511 Pontiac Avenue Cranston, Rhode Island 02920 (401) 462-9645

#### **SOUTH DAKOTA**

Division of Insurance Securities Regulation 124 S. Euclid, Suite 104 Pierre, South Dakota 57501 (605) 773-3563

#### **VIRGINIA**

(state administrator)

State Corporation Commission Division of Securities and Retail Franchising 1300 East Main Street, Ninth Floor Richmond, Virginia 23219 (804) 371-9051

(agent for service of process)

Clerk, State Corporation Commission 1300 East Main Street Richmond, Virginia 23219 (804) 371-9733

#### **WASHINGTON**

(state administrator)

Department of Financial Institutions Securities Division P.O. Box 9033 Olympia, Washington 98507-9033 (360) 902-8760

(agent for service of process)

Director
Department of Financial Institutions Securities
Division
150 Israel Road, S.W.
Tumwater, Washington 98501

#### **WISCONSIN**

(for service of process)

Administrator, Division of Securities Department of Financial Institutions 4822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 266-2139

(state administrator)

Division of Securities Department of Financial Institutions 822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 266-9555

#### EXHIBIT H

#### STATE-SPECIFIC ADDENDA

## ADDITIONAL DISCLOSURES FOR THE FRANCHISE DISCLOSURE DOCUMENT OF MRS. FIELDS FRANCHISING, LLC

The following are additional disclosures for the Franchise Disclosure Document of Mrs. Fields Franchising, LLC required by various state franchise laws. Each provision of these additional disclosures will only apply to you if the applicable state franchise registration and disclosure law applies to you.

1. <u>No Waiver or Disclaimer of Reliance in Certain States</u>. The following provision applies only to franchisees and Franchises that are subject to the state franchise disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

#### **CALIFORNIA**

- 1. THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.
- 2. SECTION 31125 OF THE CALIFORNIA CORPORATIONS CODE REQUIRES US TO GIVE YOU A DISCLOSURE DOCUMENT, IN A FORM CONTAINING THE INFORMATION THAT THE COMMISSIONER MAY BY RULE OR ORDER REQUIRE, BEFORE A SOLICITATION OF A PROPOSED MATERIAL MODIFICATION OF AN EXISTING FRANCHISE.
- 3. OUR WEBSITE, <u>www.mrsfieldsfranchise.com</u> HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION AT <u>www.dfpi.ca.gov</u>.
  - 4. The following is added at the end of Item 3:

Neither we, our parent, predecessor or affiliates nor any person in Item 2 of the Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. Sections 78a et seq., suspending or expelling such persons from membership in that association or exchange.

5. The following is added at the end of Item 6:

The maximum interest rate allowed in California is 10%.

6. The following paragraphs are added at the end of Item 17:

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or nonrenewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, and the law applies, the law will control.

The Franchise Agreement contains a covenant not to compete that extends beyond termination of the agreement and franchise. This provision might not be enforceable under California law.

The Franchise Agreement provides for termination upon bankruptcy. This provision might not be enforceable under federal bankruptcy law (11 U.S.C.A. Sections 101 et seq.).

The Franchise Agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

The Franchise Agreement requires binding arbitration. The arbitration will occur in Delaware with costs being borne as described in the Franchise Agreement. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as the Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

The Franchise Agreement requires you to sign a general release of claims upon renewal or transfer of the Franchise Agreement. California Corporations Code Section 31512 provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of that law or any rule or order thereunder is void. Section 31512 might void a waiver of your rights under the Franchise Investment Law (California Corporations Code Section 31000 – 31516). Business and Professions Code Section 20010 might void a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 – 20043).

#### 7. The following paragraph is added to the end of Item 19:

The earnings claims figures do not reflect the costs of sales, operating expenses or other costs or expenses that must be deducted from gross revenue or gross sales figures to obtain your net income or profit. You should conduct an independent investigation of the costs and expenses you will incur in operating your Franchise Business. Franchisees or former franchisees listed in the Franchise Disclosure Document may be one source of this information.

#### **HAWAII**

THESE FRANCHISES WILL BE/HAVE BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF REGULATORY AGENCIES OR A FINDING BY THE DIRECTOR OF REGULATORY AGENCIES THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE

H-3 MFF 2025 FDD

FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE FRANCHISE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS FRANCHISE DISCLOSURE DOCUMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

#### **ILLINOIS**

- 1. Illinois law governs the franchise agreement.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 4. Franchisee's rights upon Termination and Non-Renewal are set forth in Sections 19 and 20 of the Illinois Franchise Disclosure Act.

#### **MARYLAND**

1. The following is added to the end of the "Summary" sections of Item 17(c), entitled Requirements for franchisee to renew or extend, and Item 17(m), entitled **Conditions for franchisor approval of transfer**:

However, any release required as a condition of renewal, relocation, sale and/or assignment/transfer will not apply to claims or liability arising under the Maryland Franchise Registration and Disclosure Law.

2. The following is added to the end of the "Summary" section of Item 17(h), entitled "Cause" defined – non-curable defaults:

The Franchise Agreement provides for termination upon bankruptcy. This provision might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.), but we will enforce it to the extent enforceable.

3. The following sentence is added to the end of the "Summary" section of Item 17(v), entitled **Choice of forum**, and 17(w), entitled **Choice of Law**:

You may bring suit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

4. The following language is added to the end of the chart in Item 17:

You must bring any claims arising under the Maryland Franchise Registration and Disclosure Law within 3 years after the grant of the franchise.

#### **MINNESOTA**

1. **Maximum Late Fee.** The following is added at the end of the chart in Item 6:

Notwithstanding the foregoing, the maximum late payment fee shall be \$30 as governed by Minnesota Statute 604.113.

2. **Renewal, Termination, Transfer and Dispute Resolution**. The following is added at the end of the chart in Item 17:

With respect to franchises governed by Minnesota law, we will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4 and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) of the Franchise Agreement and 180 days' notice for non-renewal of the Franchise Agreement.

Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400J prohibits us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Franchise Disclosure Document or Franchise Agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes 1984, Chapter 80C, or your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction. Those provisions also provide that no condition, stipulation or provision in the Franchise Agreement will in any way abrogate or reduce any of your rights under the Minnesota Franchises Law, including, if applicable, the right to submit matters to the jurisdiction of the courts of Minnesota.

### State Cover Page and Item 17, Additional Disclosures:

Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400(J) prohibit us from requiring litigation to be conducted outside of Minnesota, requiring waiver of a jury trial or requiring the franchisee to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Disclosure Document shall abrogate or reduce any of your rights as provided for in Minn. Stat. Sec. 80C, or your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction.

Franchisee cannot consent to the franchisor obtaining injunctive relief. The franchisor may seek injunctive relief. A court will determine if a bond is required.

### Item 6, Additional Disclosure:

NSF checks are governed by Minn. Stat. 604.113, which puts a cap of \$30 on service charges.

### Item 13, Additional Disclosures:

The Minnesota Department of Commerce requires that a franchisor indemnify Minnesota Franchisees against liability to third parties resulting from claims by third parties that the franchisee's use of the franchisor's trademark infringes upon the trademark rights of the third party. The franchisor does not indemnify against the consequences of a franchisee's use of a franchisor's

trademark except in accordance with the requirements of the franchise agreement, and as the condition to an indemnification, the franchise must provide notice to the franchisor of any such claim immediately and tender the defense of the claim to the franchisor. If the franchisor accepts tender of defense, the franchisor has the right to manage the defense of the claim, including the right to compromise, settle or otherwise resolve the claim, or to determine whether to appeal a final determination of the claim.

Minnesota considers it unfair to not protect the franchisee's right to use the trademarks. Refer to Minnesota Statute 80C.12.Subd. 1(G). The franchisor will protect the franchisee's rights to use the trademarks, service marks, trade names, logotypes or other commercial symbols or indemnify the franchisee from any loss, costs, or expenses arising out of any claim, suite or demand regarding the use of the name.

### Item 17, Additional Disclosures:

Any condition, stipulation or provision, including any choice of law provision, purporting to bind any person who, at the time of acquiring a franchise is a resident of the State of Minnesota or in the case of a partnership or corporation, organized or incorporated under the laws of the State of Minnesota, or purporting to bind a person acquiring any franchise to be operated in the State of Minnesota to waive compliance or which has the effect of waiving compliance with any provision of the Minnesota Franchise Law is void.

We will comply with Minn. Stat. Sec. 80C.14, subds. 3, 4 and 5, which requires, except in certain specified cases, that a franchisee be given 90 days notice of termination (with 60 days to cure), 180 days notice for nonrenewal of the Franchise Agreement, and that consent to the transfer of the franchise will not be unreasonably withheld.

Minnesota Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release, assignment, novation, or waiver that would relieve any person from liability imposed by Minnesota Statute \$80C.01 - 80C.22.

The limitations of claims section must comply with Minn. Stat. Sec. 80C.17, subd. 5.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

### **NEW YORK**

To the extent the New York General Business Law, Article 33, §§680 - 695 applies, the terms of this Addendum apply.

1. The following information is added to the cover page of the Franchise Disclosure Document.

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT F OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN

THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CAN NOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is to be added at the end of Item 3:

With the exception of what is stated above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal, or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.
- No such party has been convicted of a felony or pleaded <u>nolo contendere</u> to a felony charge or, within the 10-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation, or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

### Item 5, Additional Disclosures.

The initial franchise fee constitutes part of our general operating funds and will be used as such in our discretion.

The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for a franchisee to renew or extend," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied.

H-7 MFF 2025 FDD

4. The following language replaces the "Summary" section of Item 17(d), titled "Termination by franchisee":

You may terminate the agreement on any grounds available by law.

5. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum," and Item 17(w), titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchiser or the franchisee by Article 33 of the General Business Law of the State of New York

- 6. Franchise Questionnaires and Acknowledgements--No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts--Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchisee before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, ten (10) business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

### NORTH DAKOTA

- 1. <u>Lost Revenue Damages</u>. The Item 6 line item entitled <u>Lost Revenue Damages</u> will not be enforced to the extent prohibited by applicable law.
- 2. The following is added to the end of the "Summary" sections of Item 17(c), entitled **Requirements for franchisee to renew or extend**, and Item 17(m), entitled Conditions for franchisor approval of transfer:

However, any release required as a condition of renewal, sale and/or assignment/transfer will not apply to the extent prohibited by the North Dakota Franchise Investment Law.

3. The following is added to the end of the "Summary" section of Item 17(r), entitled **Non-competition covenants after the franchise is terminated or expires**:

Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota; however, we and you will enforce the covenants to the maximum extent the law allows.

4. The "Summary" section of Item 17(u), entitled <u>Dispute resolution by arbitration or mediation</u> is deleted and replaced with the following:

To the extent required by the North Dakota Franchise Investment Law (unless such requirement is preempted by the Federal Arbitration Act), arbitration will be at a site to which we and you mutually agree.

5. The "Summary" section of Item 17(v), entitled <u>Choice of forum</u>, is deleted and replaced with the following:

You must sue us in a Court in the State of Delaware, except that, subject to your arbitration obligation, and to the extent required by the North Dakota Franchise Investment Law, you may bring an action in North Dakota.

6. The "Summary" section of Item 17(w), entitled <u>Choice of law</u>, is deleted and replaced with the following:

Except as otherwise required by North Dakota law, the laws of the State of Delaware will apply.

### **RHODE ISLAND**

1. The following language is added to the end of the "Summary" sections of Item 17(v), entitled **Choice of forum**, and 17(w), entitled **Choice of law**:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act (the "Act") provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

### **VIRGINIA**

1. The following language is added to the end of the "Summary" section of Item 17(e), entitled **Termination by franchisor without cause**:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the Franchise Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

#### WASHINGTON

1. The following matter is added to Item 3:

In re: Franchise No Poaching Provisions (Mrs. Fields Franchising, LLC) (Case No. 19-2- 24791-6; State of Washington, King County Superior Court). Beginning in January 2018, the Washington Attorney General launched a sweeping investigation into the use of non-solicitation and no-hire provisions in franchise agreements, with the stated goal of "eliminating no-poach clauses nationwide." The Washington Attorney General asserted that such no-poach provisions violated the Washington Consumer Protection Act. We elected, on September 20, 2019, to enter into an Assurance of Discontinuance ("AOD") to avoid the time and expense of a protracted court action. Other than as a mechanism for the court to approve and enter the AOD, no court proceeding was initiated. Under the terms of the AOD, we agreed to: notify all franchisees that we entered into the AOD, not include no-poach provisions in our future agreements, not enforce such provisions in our existing franchise agreements, exercise reasonable commercial efforts to amend all existing franchise agreements with entities in Washington to remove any no-poach provisions in their existing agreement, and remove those provisions from existing agreements as they came up for renewal or renegotiation. Under its express terms, the AOD is not to be construed as an admission of law, fact, liability, misconduct, or wrongdoing on our part.

### 2. The following paragraph is added at the end of Item 17:

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed

\$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

# THE FOLLOWING PAGES IN THIS EXHIBIT ARE STATE-SPECIFIC RIDERS TO THE FRANCHISE AGREEMENT

MFF 2025 FDD H-11 65609236v3

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN ILLINOIS

a Delaware limited liability company whose address it 1717 S 4800 W. Salt Lake City. Utah 84104 ("we")

THIS RIDER is made and entered into by and between MRS. FIELDS FRANCHISING, LLC,

| and             |                         | , a(n) | whose principal                                                      |
|-----------------|-------------------------|--------|----------------------------------------------------------------------|
| business addres | ss is                   |        | ("you").                                                             |
| 1.              |                         |        | ertain Franchise Agreement dated is annexed to and forms part of the |
| Franchise Agre  |                         |        | relating to the Franchise Agreement                                  |
| was made or ac  | ecepted in Illinois and |        | ler the Franchise Agreement will be                                  |

2. **GOVERNING LAW**. Section 17.4 of the Franchise Agreement is deleted and replaced with the following:

All matters relating to arbitration will be governed by the United States Federal Arbitration Act (9 U.S.C. sections 1 et seq.). Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. sections 1051 et seq.), or other United States federal law, this Agreement, the franchise, and all claims arising from the relationship between us and you will be governed by the laws of the State of Illinois without regard to its conflict of laws rules.

3. <u>ILLINOIS FRANCHISE DISCLOSURE ACT</u>. The following language is added to the end of the Franchise Agreement:

In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.

In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

Franchisees rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

[signature page follows]

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Franchise Agreement.

FRANCHISEE OWNER:

### 

Date\*: \_\_\_\_\_ Name: \_\_\_\_\_

MRS. FIELDS FRANCHISING, LLC

Date: \_\_\_\_\_

Title: \_\_\_\_\_

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN MARYLAND

|                     | <b>DER</b> is made and entered into by and between <b>MRS. FII</b> liability company whose address it 1717 S. 4800 W., Sala | · · · · · · · · · · · · · · · · · · · |
|---------------------|-----------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| and                 | , a(n)                                                                                                                      | whose principal                       |
| business address is |                                                                                                                             | ("you").                              |
|                     | ACKGROUND. We and you are parties to that certa                                                                             |                                       |
| •                   | ent. This Rider is being signed because (a) you are domic operate under the Franchise Agreement will be located in          | •                                     |

2. **RELEASES.** The following is added to the end of Sections 3.1(c) and 12.3(e) of the Franchise Agreement:

However, any release required as a condition of renewal, sale and/or assignment/transfer will not apply to any claims or liability arising under the Maryland Franchise Registration and Disclosure Law.

3. **INSOLVENCY**. The following sentence is added to the end of Section 13.1(a) of the Franchise Agreement:

This Section 13.1(a) may not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.).

4. **CONSENT TO JURISDICTION**. The following language is added to the end of Section 17.5 of the Franchise Agreement:

You may bring an action in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

5. <u>LIMITATIONS OF CLAIMS; WAIVER OF CLASS ACTION</u>. The following sentence is added to the end of Section 17.7 of the Franchise Agreement:

You must bring any claims arising under the Maryland Franchise Registration and Disclosure Law within 3 years after we grant you the franchise.

6. **GOVERNING LAW**. Section 17.4 of the Franchise Agreement is deleted and replaced with the following:

All matters relating to arbitration will be governed by the United States Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). Except to the extent governed by the Federal Arbitration Act, the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. sections 1051 et seq.), or other federal law, this Agreement, the Franchise, and all claims arising from the relationship between us and you will be governed by the laws of Delaware, without regard to its conflict of laws rules, except that (1) any state law regulating the offer or sale of franchises or governing the relationship of a franchisor and its franchisee will not apply unless its jurisdictional requirements are met independently without reference to this section, and (2) Maryland law will apply to claims arising under the Maryland Franchise Registration and Disclosure Law.

7. **ACKNOWLEDGMENTS**. The following is added as a new Section 19 to the end of the Franchise Agreement:

### 8. **ACKNOWLEDGEMENTS.**

All representations requiring you to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Franchise Agreement.

### MRS. FIELDS FRANCHISING, LLC

### FRANCHISEE OWNER:

| By:                           | _ [Name] |
|-------------------------------|----------|
| Name:                         | By:      |
| Title:                        | Name:    |
| Date*:                        |          |
| (*This is the Effective Date) | Date:    |

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN MINNESOTA

THIS RIDER is made and entered into by and between MRS. FIELDS FRANCHISING, LLC,

| a Delaware limited liability company who and | ose address it 1717 S. 4800 W., S | •                                   |
|----------------------------------------------|-----------------------------------|-------------------------------------|
| business address is                          |                                   | ("you").                            |
|                                              | • •                               | tain Franchise Agreement dated      |
|                                              |                                   | s annexed to and forms part of the  |
| Franchise Agreement. This Rider is being     | ng signed because (a) the Store   | e that you will operate under the   |
| Franchise Agreement will be located in M     | Minnesota; and/or (b) any of the  | offering or sales activity relating |
| to the Franchise Agreement occurred in M     | Minnesota                         |                                     |

2. <u>INFRINGEMENT AND CLAIMS</u>. The following sentence is added to the end of Section 10.4 of the Franchise Agreement:

Provided you have complied with all provisions of this Agreement applicable to the Marks, we will protect your right to use the Marks and will indemnify you from any loss, costs or expenses arising out of any claims, suits or demands regarding your use of the Marks in accordance with Minn. Stat. Sec. 80C 12, Subd. 1(g).

- 3. <u>IN-TERM NON-COMPETE AND NON-SOLICITATION; POST-TERM NON-COMPETE AND NON-SOLICITATION</u>. Sections 11.1 and 11.2 of the Franchise Agreement will not be enforced to the extent prohibited by applicable law.
- 4. **RELEASES.** The following is added to the end of Sections 3.1(c) and 12.3(e) of the Franchise Agreement:

Any release required as a condition of renewal and/or assignment/transfer will not apply to the extent prohibited by the Minnesota Franchises Law.

5. **SUCCESSOR FRANCHISE AND OTHER RIGHTS AND REMEDIES**. The following is added to the end of Sections 3.1 and 13.2 of the Franchise Agreement:

However, with respect to franchises governed by Minnesota law, we will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4 and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice of non-renewal of this Agreement.

6. **LOST REVENUE DAMAGES**. The following language is added to the end of Section 14.3 of the Franchise Agreement

We and you acknowledge that certain parts of this provision might not be enforceable under Minn. Rule Part 2860.4400J. However, we and you agree to enforce the provision to the extent the law allows.

7. **INJUNCTIVE RELIEF.** Section 17.1 of the Franchise Agreement is deleted and replaced with the following:

Nothing in this Agreement bars our right to obtain specific performance of the provisions of this Agreement and seek injunctive relief against conduct that threatens to injure or harm us, the Marks or the System, under customary equity rules, including applicable rules for obtaining restraining orders and preliminary injunctions. You agree that we may seek such injunctive relief. You agree that your only remedy if an injunction is entered against you will be the dissolution of that injunction, if warranted, upon due hearing, and you hereby expressly waive any claim for damages caused by such injunction. A court will determine if a bond is required.

8. **GOVERNING LAW**. The following statement is added at the end of Section 17.4 of the Franchise Agreement:

Nothing in this Agreement will abrogate or reduce any of your rights under Minnesota Statutes Chapter 80C or your right to any procedure, forum or remedies that the laws of the jurisdiction provide.

9. **CONSENT TO JURISDICTION**. The following language is added to the end of Section 17.5 of the Franchise Agreement:

Notwithstanding the foregoing, Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400j prohibit us, except in certain specified cases, from requiring litigation to be conducted outside of Minnesota. Nothing in this Agreement will abrogate or reduce any of your rights under Minnesota statutes Chapter 80C or your rights to any procedure, forum or remedies that the laws of the jurisdiction provide.

- 10. **WAIVER OF PUNITIVE DAMAGES AND JURY TRIAL**. If and then only to the extent required by the Minnesota Franchises Law, Section 17.6 of the Franchise Agreement is deleted.
- 11. **LIMITATIONS OF CLAIMS; WAIVER OF CLASS ACTION**. The following is added to the end of Section 17.7 of the Franchise Agreement:

; provided, however, that Minnesota law provides that no action may be commenced under Minn. Stat. Sec. 80C.17 more than 3 years after the cause of action accrues.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Franchise Agreement.

FRANCHISEE OWNER:

|                               | [Name] |
|-------------------------------|--------|
| Ву:                           | By:    |
| Name:                         | Name:  |
| Title:                        | Title: |
| Date*:                        | Date:  |
| (*This is the Effective Date) |        |

MRS FIELDS FRANCHISING LLC

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN THE STATE OF NEW YORK

| a Delaware limited liability company whose addre and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | by and between <b>MRS. FIELDS FRANCHISING, LLC</b> , ass it 1717 S. 4800 W., Salt Lake City, Utah 84104 (" <b>we</b> "), a(n) whose principal                                          |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| business address is                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ("you").                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | are parties to that certain Franchise Agreement dated<br>e Agreement"). This Rider is being signed because (a) you                                                                     |
| are domiciled in the State of New York and the St                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | tore that you will operate under the Franchise Agreement the offering or sales activity relating to the Franchise                                                                      |
| 2. <b>TRANSFERS BY US</b> . The follow Franchise Agreement:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | owing language is added to the end of Section 12.1 of the                                                                                                                              |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | cable law, no transfer will be made except to an assignee villing and able to assume our obligations under this                                                                        |
| 3. <b>RELEASES</b> . The following lang the Franchise Agreement:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | guage is added to the end of Sections 3.1(c) and 12.3(e) of                                                                                                                            |
| favor from the provisions of Article 33 of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | enjoyed by you and any causes of action arising in your the General Business Law of the State of New York and emain in force to the extent required by the non-waiver 7.5, as amended. |
| 4. OTHER RIGHTS AND REMINE Section 13.2 of the Franchise Agreement:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | <b>EDIES</b> . The following language is added to the end of                                                                                                                           |
| You also may terminate this Agreement of Article 33 of the General Business Law of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | on any grounds available by law under the provisions of f the State of New York.                                                                                                       |
| 5. <b>GOVERNING LAW; CONSE</b> added at the end of Sections 17.4 and 17.5 of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | NT TO JURISDICTION. The following statement is Franchise Agreement:                                                                                                                    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | iver of any right conferred upon you by the provisions of<br>al Business Law, as amended, and the regulations issued                                                                   |
| <b>IN WITNESS WHEREOF</b> , the parties helow, to be effective as of the Effective Date of the | nave executed and delivered this Rider on the dates noted the Franchise Agreement.                                                                                                     |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | FRANCHISEE OWNER:                                                                                                                                                                      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | [Name]                                                                                                                                                                                 |
| By:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | By:                                                                                                                                                                                    |
| Name:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Name:                                                                                                                                                                                  |
| Title:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Title:                                                                                                                                                                                 |
| Date*:(*This is the Effective Date)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | Date:                                                                                                                                                                                  |

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN NORTH DAKOTA

| THIS RIDER is made and entered into by and between MRS. FIELDS FRANCHISING, LLC                                                                                                                                     |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| a Delaware limited liability company whose address it 1717 S. 4800 W., Salt Lake City, Utah 84104 ("we") and, a(n) whose principal                                                                                  |
| business address is("you").                                                                                                                                                                                         |
|                                                                                                                                                                                                                     |
| 1. <b>BACKGROUND.</b> We and you are parties to that certain Franchise Agreement dated                                                                                                                              |
| , 20, (the "Franchise Agreement"). This Rider is annexed to and forms part of the                                                                                                                                   |
| Franchise Agreement. This Rider is being signed because (a) you are a resident of North Dakota <u>and</u> the Store that you will operate under the Franchise Agreement will be located or operated in North Dakota |
| and/or (b) any of the offering or sales activity relating to the Franchise Agreement occurred in North Dakota                                                                                                       |
| and of (b) any of the offering of sales activity feating to the Franchise rigidement occurred in North Barkon                                                                                                       |
| 2. <b>RELEASES</b> . The following is added to the end of Sections 3.1(c) and 12.3(e) of the Franchise Agreement:                                                                                                   |
| A                                                                                                                                                                                                                   |
| Any release required as a condition of renewal, and/or assignment/transfer will not apply to the extent prohibited by the North Dakota Franchise Investment Law.                                                    |
| 3. IN-TERM NON-COMPETE AND NON-SOLICITATION AND POST-TERM                                                                                                                                                           |
| NON-COMPETE AND NON-SOLICITATION. The following is added to the end of Sections 11.1 and                                                                                                                            |
| 11.2 of the Franchise Agreement:                                                                                                                                                                                    |
|                                                                                                                                                                                                                     |
| Covenants not to compete and non-solicitation provisions such as those mentioned above are                                                                                                                          |
| generally considered unenforceable in the State of North Dakota; however, we will enforce the covenants to the maximum extent the law allows.                                                                       |
| 4. <b>LOST REVENUE DAMAGES</b> . The following language is added to the end of Section                                                                                                                              |
| 14.3 of the Franchise Agreement:                                                                                                                                                                                    |
|                                                                                                                                                                                                                     |
| We and you acknowledge that certain parts of this provision might not be enforceable under the                                                                                                                      |
| North Dakota Franchise Investment Law. However, we and you agree to enforce the provision to                                                                                                                        |
| the extent the law allows.                                                                                                                                                                                          |
| 5. <b>ARBITRATION</b> . The first paragraph of Section 17.8 of the Franchise Agreement i                                                                                                                            |
| amended to read as follows:                                                                                                                                                                                         |
|                                                                                                                                                                                                                     |
| We and you agree that all controversies, disputes, or claims between us or our affiliates, and ou                                                                                                                   |
| and their respective owners, officers, directors, agents, and employees, on the one hand, and you                                                                                                                   |
| (and your owners, guarantors, Affiliates, and employees), on the other hand, arising out of or related                                                                                                              |
| to:                                                                                                                                                                                                                 |
| (1) this Agreement or any other agreement between you (or any of your owners) and                                                                                                                                   |
| us (or any of our affiliates);                                                                                                                                                                                      |
|                                                                                                                                                                                                                     |
| (2) our relationship with you;                                                                                                                                                                                      |
| (3) the scope or validity of this Agreement or any other agreement between you (o                                                                                                                                   |
| any of your owners) and us (or any of our affiliates) or any provision of any of such agreement                                                                                                                     |

(including the validity and scope of the arbitration provision under this Section 17.8, which we and you acknowledge is to be determined by an arbitrator, not a court); or

- (4) any System Standard must be submitted for binding arbitration, on demand of either party, to the American Arbitration Association. The arbitration proceedings will be conducted by one arbitrator and, except as this Section otherwise provides, according to the American Arbitration Association's then-current Commercial Arbitration Rules. All proceedings will be conducted at a suitable location chosen by the arbitrator that is within Wilmington, Delaware; provided, however, that to the extent otherwise required by the North Dakota Franchise Investment Law (unless such a requirement is preempted by the Federal Arbitration Act), arbitration shall be held at a site to which we and you mutually agree. All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. §§ 1 et seq.). The interim and final awards of the arbitrator shall be final and binding upon each party, and judgment upon the arbitrator's awards may be entered in any court of competent jurisdiction.
- **GOVERNING LAW.** Section 17.4 of the Franchise Agreement is deleted and replaced 6. with the following:

All matters relating to arbitration will be governed by the Federal Arbitration Act (9 U.S.C. Sections 1 et seq.). Except to the extent governed by the Federal Arbitration Act, the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. sections 1051 et seq.), or other federal law, and except as otherwise required by North Dakota law, this Agreement, the Franchise, and all claims arising from the relationship between us and you will be governed by the laws of Delaware, without regard to its conflict of laws rules, except that any state law regulating the offer or sale of franchises or governing the relationship of a franchisor and its franchisee will not apply unless its jurisdictional requirements are met independently without reference to this section.

**CONSENT TO JURISDICTION**. The following is added to the end of Section 17.5 of the Franchise Agreement:

Notwithstanding the foregoing, to the extent required by the North Dakota Franchise Investment Law, and subject to your arbitration obligations, you may bring an action in North Dakota for claims arising under the North Dakota Franchise Investment Law.

- WAIVER OF PUNITIVE DAMAGES AND JURY TRIAL. To the extent required by the North Dakota Franchise Investment Law, Section 17.6 of the Franchise Agreement is deleted.
- **LIMITATIONS OF CLAIMS; WAIVER OF CLASS ACTION**. Section 17.7 of the Franchise Agreement is deleted in its entirety and replaced with the following paragraph:

Except for claims arising from your non-payment or underpayment of amounts you owe us, any and all claims arising out of or relating to this Agreement or our relationship with you will be barred unless a judicial or arbitration proceeding is commenced in accordance with this Agreement within one (1) year from the date on which the party asserting the claim knew or should have known of the facts giving rise to the claims. The statutes of limitations under North Dakota Law applies with respect to claims arising under the North Dakota Franchise Investment Law. We and you agree that any proceeding will be conducted on an individual basis and that any proceeding between us and any of our Affiliates, or our and their respective owners, officers, directors, agents, and employees, on the one hand, and you (or your owners, guarantors, affiliates, and employees), on the other hand, may not be: (i) conducted on a class-wide basis, (ii) commenced, conducted or consolidated with any other proceeding, (iii) joined with any claim of an unaffiliated third-party, or (iv) brought on

H-20MFF 2025 FDD

your behalf by any association or agent. No previous course of dealing shall be admissible to explain, modify, or contradict the terms of this Agreement. No implied covenant of good faith and fair dealing shall be used to alter the express terms of this Agreement.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Franchise Agreement.

| MRS. FIELDS FRANCHISING, LLC        | FRANCHISEE OWNER: |  |
|-------------------------------------|-------------------|--|
|                                     | [Name]            |  |
| By:                                 | By:               |  |
| Name:                               | Name:             |  |
| Title:                              | Title:            |  |
| Date*:(*This is the Effective Date) | Date:             |  |

### RIDER TO THE FRANCHISE AGREEMENT FOR USE IN RHODE ISLAND

| a Delaware limited liability company whose add and                                                                                                                                                              |                                                                                            | •                                                                 |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-------------------------------------------------------------------|
| business address is                                                                                                                                                                                             |                                                                                            | ("you").                                                          |
|                                                                                                                                                                                                                 | Agreement"). This Rider is a                                                               | annexed to and forms part of the                                  |
| Franchise Agreement. This Rider is being sign. Store that you will operate under the Franchise of the offering or sales activity relating to the Franchise                                                      | Agreement will be located                                                                  | in Rhode Island; and/or (b) any                                   |
| 2. <b>GOVERNING LAW / CONS</b> added to the end of Sections 17.4 and 17.5 of the                                                                                                                                |                                                                                            | <u><b>DN</b></u> . The following language is                      |
| Section 19-28.1-14 of the Rhode Island franchise agreement restricting jurisdic application of the laws of another state this act." To the extent required by appunder the Rhode Island Franchise Investigation | tion or venue to a forum or<br>is void with respect to a cla<br>blicable law, Rhode Island | utside this state or requiring the im otherwise enforceable under |
| <b>IN WITNESS WHEREOF</b> , the parties below, to be effective as of the Effective Date of                                                                                                                      |                                                                                            |                                                                   |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                    | FRANCHISEE OV                                                                              | VNER:                                                             |
|                                                                                                                                                                                                                 | [Name]                                                                                     |                                                                   |
| Ву:                                                                                                                                                                                                             | By:                                                                                        |                                                                   |
| Name:                                                                                                                                                                                                           | Name:                                                                                      | _                                                                 |
| Title:                                                                                                                                                                                                          | Title:                                                                                     |                                                                   |
| Date*:(*This is the Effective Date)                                                                                                                                                                             | Date:                                                                                      |                                                                   |

## WASHINGTON RIDER TO THE FRANCHISE AGREEMENT, REPRESENTATIONS AND ACKNOWLEGMENT " STATEMENT, AND RELATED AGREEMENTS

| a Delaware lim | nited liability company whose address it 1717 S. 4800 W., Salt | Lake City, Utah 84104 (" <b>we</b> "), |
|----------------|----------------------------------------------------------------|----------------------------------------|
| and            | , a(n)                                                         | whose principal                        |
| business addre | ss is                                                          | ("you").                               |
| 1              | BACKGROUND. We and you are parties to that certain             | n Franchica Agraamant datad            |
| 1.             | , 20, (the "Franchise Agreement"). This Rider is an            | <u>C</u>                               |
| Franchise Agre | eement. This Rider is being signed because (a) you are domic   | -                                      |

THIS RIDER is made and entered into by and between MRS. FIELDS FRANCHISING, LLC,

2. <u>WASHINGTON LAW</u>. The following paragraphs are added to the end of the Franchise Agreement:

the Store that you will operate under the Franchise Agreement will be located or operated in Washington; and/or (c) any of the offering or sales activity relating to the Franchise Agreement occurred in Washington.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained

MFF 2025 FDD H-23

in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Franchise Agreement.

| MRS. FIELDS FRANCHISING, LLC        | FRANCHISEE OWNER: |  |
|-------------------------------------|-------------------|--|
|                                     | [Name]            |  |
| By:                                 | By:               |  |
| Name:                               | Name:             |  |
| Title:                              | Title:            |  |
| Date*:(*This is the Effective Date) | Date:             |  |

# THE FOLLOWING PAGES IN THIS EXHIBIT ARE STATE-SPECIFIC RIDERS TO THE AREA DEVELOPMENT AGREEMENT

MFF 2025 FDD H-25 65609236v3

## RIDER TO THE AREA DEVELOPMENT AGREEMENT FOR USE IN ILLINOIS

| andbusiness address is                                                                                                                                                             | , a(n)                                                                                            | whose principa                                                                                            |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|
| business address is                                                                                                                                                                |                                                                                                   | ("you").                                                                                                  |
| 1. <b>BACKGROUND</b> . We are dated, 20 (the forms part of the Area Development Agreement was under the Area Development Agreement v                                               | 'Area Development Agreemen<br>tement. This Rider is being sig-<br>made or accepted in Illinois ar | ned because (a) the offer relating the Store that you will operate                                        |
| 2. <u>ILLINOIS FRANCHISI</u> end of the Area Development Agreement:                                                                                                                |                                                                                                   | ollowing language is added to th                                                                          |
| In conformance with Section 4 development agreement that desi Illinois is void. However, a development of Illinois.                                                                | gnates jurisdiction and venue                                                                     | in a forum outside the State of                                                                           |
| In conformance with Section 41 or or provision purporting to bind ar Illinois Franchise Disclosure Act                                                                             | ny person acquiring any franch                                                                    | ise to waive compliance with th                                                                           |
| Franchisees rights upon Terminat Illinois Franchise Disclosure Act.                                                                                                                | ion and Non-Renewal are set f                                                                     | Forth in sections 19 and 20 of th                                                                         |
| No statement, questionnaire, or active commencement of the franch under any applicable state franch reliance on any statement made by us. This provision supersedes any Franchise. | ise relationship shall have the ise law, including fraud in the us, any franchise seller, or any  | effect of (i) waiving any claim<br>e inducement, or (ii) disclaimin<br>y other person acting on behalf of |
| IN WITNESS WHEREOF, the protection of the Effective as of the Effective as of the Effective ARS. FIELDS FRANCHISING, LLC                                                           | ctive Date of the Area Develop                                                                    | ment Agreement.                                                                                           |
|                                                                                                                                                                                    | [Name]                                                                                            |                                                                                                           |
| Ву:                                                                                                                                                                                | By:                                                                                               |                                                                                                           |
| Name:                                                                                                                                                                              | Name:                                                                                             |                                                                                                           |
| Title:                                                                                                                                                                             | Title :                                                                                           |                                                                                                           |
| Date*:(*This is the Effective Date)                                                                                                                                                | Date:                                                                                             |                                                                                                           |

MFF 2025 FDD H-26

### RIDER TO THE AREA DEVELOPMENT AGREEMENT FOR USE IN MARYLAND

| THIS RIDER is made and entered into be a Delaware limited liability company whose addressed and                                                                                                                                                       | ess it 1717 S. 4800 W., Salt Lake City, U                                                                                    | Jtah 84104 (" <b>we</b> "),         |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------|
| business address is                                                                                                                                                                                                                                   |                                                                                                                              |                                     |
| 1. <b>BACKGROUND</b> . We and you dated, 20 (the "Area I forms part of the Area Development Agreement. in Maryland, and/or (b) the Store that you will clocated in Maryland.                                                                          | This Rider is being signed because (a)                                                                                       | is annexed to and you are domiciled |
| 2. <b>RELEASES.</b> The following is Development Agreement:                                                                                                                                                                                           | s added to the end of Section 8(B                                                                                            | )(6) of the Area                    |
| However, any release required as a cond apply to any claims or liability arising ur Law.                                                                                                                                                              |                                                                                                                              |                                     |
| 3. <b>ACKNOWLEDGMENTS</b> . The Area Development Agreement:                                                                                                                                                                                           | following is added as a new Section 1                                                                                        | 2 to the end of the                 |
| All representations requiring you to ass<br>intended to nor shall they act as a releas<br>Maryland Franchise Registration and Dis                                                                                                                     | se, estoppel or waiver of any liability i                                                                                    |                                     |
| No statement, questionnaire, or acknowledgement commencement of the franchise relationship shat applicable state franchise law, including fraud statement made by us, any franchise seller, or a supersedes any other term of any document execution. | all have the effect of (i) waiving any<br>in the inducement, or (ii) disclaiming<br>any other person acting on behalf of the | claims under any g reliance on any  |
| <b>IN WITNESS WHEREOF</b> , the parties below, to be effective as of the Effective Date of                                                                                                                                                            |                                                                                                                              | on the dates noted                  |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                                                          | AREA DEVELOPER:                                                                                                              |                                     |
|                                                                                                                                                                                                                                                       | [Name]                                                                                                                       |                                     |
| By:                                                                                                                                                                                                                                                   | By:                                                                                                                          |                                     |
| Name:                                                                                                                                                                                                                                                 | Name:                                                                                                                        |                                     |
| Title:                                                                                                                                                                                                                                                | Title :                                                                                                                      |                                     |
| Date*:(*This is the Effective Date)                                                                                                                                                                                                                   | Date:                                                                                                                        |                                     |

## RIDER TO THE AREA DEVELOPMENT AGREEMENT FOR USE IN MINNESOTA

| THIS RIDER is made and entered into be a Delaware limited liability company whose addreand                                                                                                                                                    | ss it 1717 S. 4800 W., Salt Lake City, U                                                                                 | Jtah 84104 (" <b>we</b> "),                                    |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------|
| business address is                                                                                                                                                                                                                           | , a(n)                                                                                                                   | (" <b>you</b> ").                                              |
| 1. <b>BACKGROUND</b> . We and you a dated, 20 (the "Area D forms part of the Area Development Agreement." will operate under the Area Development Agreement offering or sales activity relating to the Area Development Agreement.            | Development Agreement"). This Rider<br>This Rider is being signed because (a)<br>ment will be located in Minnesota; and  | is annexed to and<br>the Store that you<br>d/or (b) any of the |
| 2. <b>RELEASES.</b> The following is Development Agreement:                                                                                                                                                                                   | added to the end of Section 8(B                                                                                          | )(6) of the Area                                               |
| Any release required as a condition of reextent prohibited by the Minnesota France                                                                                                                                                            |                                                                                                                          | ll not apply to the                                            |
| 3. <b>DEFAULT AND TERMINATI</b> the Area Development Agreement:                                                                                                                                                                               | <b>ON</b> . The following is added to the end                                                                            | of Section 7(B) of                                             |
| However, with respect to franchises gove Sec. 80C.14, Subds. 3, 4 and 5 which req 90 days' notice of termination (with 60 d Agreement.                                                                                                        | uire, except in certain specified cases,                                                                                 | that you be given                                              |
| No statement, questionnaire, or acknowledgeme commencement of the franchise relationship sha applicable state franchise law, including fraud statement made by us, any franchise seller, or a supersedes any other term of any document execu | all have the effect of (i) waiving any<br>in the inducement, or (ii) disclaiming<br>any other person acting on behalf of | claims under any g reliance on any                             |
| <b>IN WITNESS WHEREOF</b> , the parties helow, to be effective as of the Effective Date of t                                                                                                                                                  |                                                                                                                          | on the dates noted                                             |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                                                  | AREA DEVELOPER:                                                                                                          |                                                                |
|                                                                                                                                                                                                                                               | [Name]                                                                                                                   |                                                                |
| By:                                                                                                                                                                                                                                           | By:                                                                                                                      |                                                                |
| Name:                                                                                                                                                                                                                                         | Name:                                                                                                                    |                                                                |
| Title:                                                                                                                                                                                                                                        | Title:                                                                                                                   |                                                                |
| Date*:                                                                                                                                                                                                                                        | Date:                                                                                                                    |                                                                |

(\*This is the Effective Date)

### RIDER TO THE AREA DEVELOPMENT AGREEMENT FOR USE IN THE STATE OF NEW YORK

| THIS RIDER is made and entered into a Delaware limited liability company whose add and                                                                                  | lress it 1717 S. 4800 W., Salt Lake Cit                                   | y, Utah 84104 (" <b>we</b> "),              |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|---------------------------------------------|
| business address is                                                                                                                                                     |                                                                           | ("you").                                    |
|                                                                                                                                                                         | few York, and/or (b) any of the offer                                     | Rider is being signed perate under the Area |
| 2. <b>TRANSFERS BY US</b> . The fol Area Development Agreement:                                                                                                         | llowing language is added to the end                                      | of Section 8(A) of the                      |
| However, to the extent required by app<br>who, in our good faith judgment, is<br>Agreement.                                                                             |                                                                           |                                             |
| 3. <b>RELEASES</b> . The following la Development Agreement:                                                                                                            | inguage is added to the end of Section                                    | n 8(B)(6) of the Area                       |
| Notwithstanding the foregoing all right<br>favor from the provisions of Article 33<br>the regulations issued thereunder shall<br>provisions of GBL Sections 687.4 and 6 | of the General Business Law of the S remain in force to the extent requir | tate of New York and                        |
| 4. OTHER RIGHTS AND REM of the Area Development Agreement:                                                                                                              | <b>MEDIES</b> . The following language is                                 | added as Section 7(F)                       |
| You also may terminate this Agreemen Article 33 of the General Business Law                                                                                             |                                                                           | nder the provisions of                      |
| IN WITNESS WHEREOF, the parties below, to be effective as of the Effective Date of                                                                                      |                                                                           | der on the dates noted                      |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                            | AREA DEVELOPER:                                                           |                                             |
|                                                                                                                                                                         | [Name]                                                                    |                                             |
| By:                                                                                                                                                                     | By:                                                                       |                                             |
| Name:                                                                                                                                                                   | Name:                                                                     |                                             |
| Title:                                                                                                                                                                  | Title:                                                                    |                                             |
| Date*:(*This is the Effective Date)                                                                                                                                     | Date:                                                                     |                                             |

### RIDER TO THE AREA DEVELOPMENT AGREEMENT FOR USE IN NORTH DAKOTA

| <b>THIS RIDER</b> is made and entered into by a Delaware limited liability company whose address and                                                                                                                                                    | s it 1717 S. 4800 W., Salt Lake City, U                                                                                   | Jtah 84104 (" <b>we</b> "),                                     |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------|
| business address is                                                                                                                                                                                                                                     |                                                                                                                           |                                                                 |
| 1. <b>BACKGROUND</b> . We and you are dated, 20, (the "Area Doforms part of the Franchise Agreement. This Rider Dakota and the Store that you will operate under the North Dakota; and/or (b) any of the offering or sale in North Dakota.              | is being signed because (a) you are a<br>he Franchise Agreement will be loca                                              | is annexed to and<br>a resident of North<br>ated or operated in |
| 2. <b>RELEASES</b> . The following is Development Agreement:                                                                                                                                                                                            | added to the end of Section 8(B                                                                                           | )(6) of the Area                                                |
| Any release required as a condition of rerextent prohibited by the North Dakota Fran                                                                                                                                                                    |                                                                                                                           | ll not apply to the                                             |
| No statement, questionnaire, or acknowledgement commencement of the franchise relationship shall applicable state franchise law, including fraud in statement made by us, any franchise seller, or an supersedes any other term of any document execute | I have the effect of (i) waiving any<br>in the inducement, or (ii) disclaiming<br>by other person acting on behalf of the | claims under any g reliance on any                              |
| IN WITNESS WHEREOF, the parties habelow, to be effective as of the Effective Date of the                                                                                                                                                                |                                                                                                                           | on the dates noted                                              |
| MRS. FIELDS FRANCHISING, LLC                                                                                                                                                                                                                            | AREA DEVELOPER:                                                                                                           |                                                                 |
|                                                                                                                                                                                                                                                         | [Name]                                                                                                                    |                                                                 |
| By:                                                                                                                                                                                                                                                     | _ By:                                                                                                                     |                                                                 |
| Name:                                                                                                                                                                                                                                                   | _ Name:                                                                                                                   |                                                                 |
| Title:                                                                                                                                                                                                                                                  | _ Title:                                                                                                                  |                                                                 |
| Date*·                                                                                                                                                                                                                                                  | Dota                                                                                                                      |                                                                 |

(\*This is the Effective Date)

### WASHINGTON RIDER TO THE AREA DEVELOPMENT AGREEMENT, REPRESENTATIONS AND ACKNOWLEDGEMENT STATEMENT, AND RELATED AGREEMENTS

| and                    | , a(n)                                               | whose principal                  |
|------------------------|------------------------------------------------------|----------------------------------|
| business address is    |                                                      | ("you").                         |
| 1. <b>BACI</b>         | KGROUND. We and you are parties to that certain      | Area Development Agreement       |
| dated                  | , 20, (the "Area Development Agreement               | "). This Rider is annexed to and |
| forms part of the Area | Development Agreement. This Rider is being signed    | ed because (a) you are domiciled |
| in Washington; and/or  | (b) the Store that you will operate under the Area I | Development Agreement will be    |
| located or operated i  | n Washington; and/or (c) any of the offering or      | sales activity relating to Area  |
| Development Agreem     | ent occurred in Washington.                          |                                  |

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

WASHINGTON LAW. The following paragraphs are added to the end of the Area

RCW 19.100.180 may supersede the development agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the development agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by an area developer may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained

MFF 2025 FDD H-31

Development Agreement:

in the development agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting an area developer from (i) soliciting or hiring any employee of an area developer of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the area development agreement or elsewhere are void and unenforceable in Washington.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the Franchise.

**IN WITNESS WHEREOF**, the parties have executed and delivered this Rider on the dates noted below, to be effective as of the Effective Date of the Area Development Agreement.

| MRS. FIELDS FRANCHISING, LLC        | AREA DEVELOPER: |  |  |  |
|-------------------------------------|-----------------|--|--|--|
|                                     | [Name]          |  |  |  |
| By:                                 | By:             |  |  |  |
| Name:                               | Name:           |  |  |  |
| Title:                              | Title:          |  |  |  |
| Date*:(*This is the Effective Date) |                 |  |  |  |

### EXHIBIT I

### SAMPLE RELEASE FORM

### MRS. FIELDS FRANCHISING, LLC

### GRANT OF FRANCHISOR CONSENT AND FRANCHISEE RELEASE

MRS. FIELDS FRANCHISING, LLC ("we," "us," or "our") and the undersigned [franchisee],

| ("you"  | or "your"), | currently    | are parties to  | a certain Fra | anchise A  | Agreement (the | e "Franchise | Agreement"   |
|---------|-------------|--------------|-----------------|---------------|------------|----------------|--------------|--------------|
| dated _ |             |              | _, 20 You       | have asked u  | s to take  | the following  | action or to | agree to the |
| followi | ng request: | [insert as a | appropriate for | renewal or tr | ansfer sit | tuation]       |              |              |
|         | 0 1         |              |                 |               |            | -              |              |              |
|         |             |              |                 |               |            |                |              |              |

We have the right under the Franchise Agreement to obtain a general release from you (and, if applicable, your owners) as a condition of taking this action or agreeing to this request. Therefore, we are willing to take the action or agree to the request specified above if you (and, if applicable, your owners) give us the release and covenant not to sue provided below in this document. You (and, if applicable, your owners) are willing to give us the release and covenant not to sue provided below as partial consideration for our willingness to take the action or agree to the request described above.

Consistent with the previous introduction, you, on your own behalf and on behalf of your successors, heirs, executors, administrators, personal representatives, agents, assigns, owners, directors, managers, officers, principals, employees, and affiliated entities (collectively, the "Releasing Parties"), hereby forever release and discharge us and our affiliates, and our and their current and former officers, directors, owners, principals, employees, agents, representatives, affiliated entities, successors, and assigns (collectively, the "Mrs. Fields Parties") from any and all claims, damages (known and unknown), demands, causes of action, suits, duties, liabilities, and agreements of any nature and kind (collectively, "Claims") that you and any of the other Releasing Parties now has, ever had, or, but for this document, hereafter would or could have against any of the Mrs. Fields Parties, including without limitation, Claims (1) arising out of or related to the Mrs. Fields Parties' obligations under the Franchise Agreement or (2) otherwise arising from or related to your and the other Releasing Parties' relationship, from the beginning of time to the date of your signature below, with any of the Mrs. Fields Parties. You, on your own behalf and on behalf of the other Releasing Parties, further covenant not to sue any of the Mrs. Fields Parties on any of the Claims released by this paragraph and represent that you have not assigned any of the Claims released by this paragraph to any individual or entity who is not bound by this paragraph.

We also are entitled to a release and covenant not to sue from your owners. By his, her, or their separate signatures below, your owners likewise grant to us the release and covenant not to sue provided above.

IF THE FRANCHISE YOU OPERATE UNDER THE FRANCHISE AGREEMENT IS LOCATED IN CALIFORNIA OR IF YOU ARE A RESIDENT OF CALIFORNIA, THE FOLLOWING SHALL APPLY:

A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor.

### IF THE FRANCHISE IS LOCATED IN MARYLAND OR IF YOU ARE A RESIDENT OF MARYLAND, THE FOLLOWING SHALL APPLY:

Any general release provided for hereunder shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

The parties hereto have signed this release on the date stated on the first page hereof.

### MRS. FIELDS FRANCHISING, LLC

| Print Name:      |       |
|------------------|-------|
| Title:           |       |
| By:              | Date: |
| FRANCHISEE       |       |
| Print Name:      |       |
| Title:           |       |
| By:              |       |
| FRANCHISEE OWNER | Date: |
| Print Name:      |       |
| Title:           |       |
| By:              | Date: |
| Print Name:      |       |
| Title:           | Date: |
| By:              |       |

### **EXHIBIT J**

### DISCLOSURE ACKNOWLEDGEMENT AGREEMENT

### REPRESENTATIONS AND ACKNOWLEDGMENT STATEMENT

The purpose of this Statement is to demonstrate to Mrs. Fields Franchising, LLC ("Franchisor") that the person(s) signing below ("I," "me" or "my"), whether acting individually or on behalf of any legal entity established to acquire the franchise rights, (a) fully understands that the purchase of a Mrs. Fields Cookie Store is a significant long-term commitment, complete with its associated risks, and (b) is not relying on any statements, representations, promises or assurances that are not specifically set forth in Franchisor's Franchise Disclosure Document and Exhibits (collectively, the "FDD") in deciding to purchase the franchise.

In that regard, I represent to Franchisor and acknowledge that:

| I understand that buying a franchise is not a guarantee of success. Purchasing or establishing any business is risky, and the success or failure of the franchise is subject to many variables such as my skills and abilities (and those of my partners, officers, employees), the time my associates and I devote to the business, competition, interest rates, the economy, inflation, operation costs, location, lease terms, the market place generally and other economic and business factors. I am aware of and am willing to undertake these business risks. I understand that the success or failure of my business will                                                                 | INITIAL: |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------|
| depend primarily upon my efforts and not those of Franchisor.  I received a copy of the FDD, including the Franchise Agreement, at least 14 calendar                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | INITIAL: |
| days (10 business days in Michigan and New York) before I executed the Franchise Agreement. I understand that all of my rights and responsibilities and those of Franchisor in connection with the franchise are set forth in these documents and only in these documents. I acknowledge that I have had the opportunity to personally and carefully review these documents and have, in fact, done so. I have been advised to have professionals (such as lawyers and accountants) review the documents for me and to have them help me understand these documents. I have also been advised to consult with other franchisees regarding the risks associated with the purchase of the franchise. |          |
| Neither the Franchisor nor any of its officers, employees or agents (including any franchise broker) has made a statement, promise or assurance to me concerning any matter related to the franchise (including those regarding advertising, marketing, training, support service or assistance provided by Franchisor) that is contrary to, or different from, the information contained in the FDD.                                                                                                                                                                                                                                                                                              | INITIAL: |
| My decision to purchase the franchise has not been influenced by any oral representations, assurances, warranties, guarantees or promises whatsoever made by the Franchisor or any of its officers, employees or agents (including any franchise broker), including as to the likelihood of success of the franchise.                                                                                                                                                                                                                                                                                                                                                                              | INITIAL: |
| I have made my own independent determination as to whether I have the capital necessary to fund the business and my living expenses, particularly during the start-up phase.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | INITIAL: |

### SPECIAL REPRESENTATION REGARDING RECEIPT OF FINANCIAL INFORMATION.

PLEASE READ THE FOLLOWING QUESTION CAREFULLY. THEN SELECT YES OR NO AND PLACE YOUR INITIALS WHERE INDICATED. Have you received any information from the Franchisor or any of its officers, employees or agents (including any franchise broker) concerning actual, average, projected or forecasted sales, revenues, income, profits or earnings of the franchise business (including any statement, promise or assurance concerning the likelihood of success) other than information contained in the FDD? □ Yes  $\square$  No (INSERT INITIAL HERE:\_\_\_\_) If you selected "Yes," please describe the information you received on the lines below: Do not sign this Representations and Acknowledgment Statement if you are a Hawaii resident, or the franchise is to be located in Hawaii. Do not sign this Representations and Acknowledgement Statement if you are a Maryland resident, or the franchise is to be located in Maryland. Do not sign this Representations and Acknowledgement Statement if you are a Washington resident, or the franchise is to be located in Washington. FRANCHISEE: Sign here if you are taking the franchise as a Sign here if you are taking the franchise as an CORPORATION, LIMITED LIABILITY **COMPANY OR PARTNERSHIP** INDIVIDUAL(S) ☐ (Note: use these blocks if you are an Print Name of Legal Entity individual or a partnership but the partnership is not a separate legal entity) By: \_\_\_\_\_ Signature Signature Print Name: Print Name: Title: Date: ☐ Signature

Print Name: Date:

### EXHIBIT K

### STATE EFFECTIVE DATES AND RECEIPT PAGES

### **State Effective Dates**

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

| State        | Effective Date |
|--------------|----------------|
| California   | Pending        |
| Hawaii       | Pending        |
| Illinois     | Pending        |
| Indiana      | Pending        |
| Maryland     | Pending        |
| Michigan     | Pending        |
| Minnesota    | Pending        |
| New York     | Pending        |
| North Dakota | Pending        |
| Rhode Island | Pending        |
| South Dakota | Pending        |
| Virginia     | Pending        |
| Washington   | Pending        |
| Wisconsin    | Pending        |

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

### RECEIPT (OUR COPY)

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Mrs. Fields Franchising, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, Mrs. Fields Franchising, LLC or an affiliate in connection with the proposed franchise sale, or sooner if required by applicable state law. Under Iowa law, we must give you this disclosure document at the earlier of our 1st personal meeting or 14 calendar days before you sign an agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale. Under Michigan law, we must give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. Under New York law, we must provide this disclosure document at the earlier of the 1st personal meeting or 10 business days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

If Mrs. Fields Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit G.

The franchisor is Mrs. Fields Franchising LLC, 1717 S. 4800 W., Salt Lake City, Utah 84104; (801) 412-8890. The franchise seller for this offering is: ☐ Rich Hankins ☐ Paul Correale ☐ Name of Franchise Seller: Mrs. Fields Franchising, LLC Mrs. Fields Franchising, LLC 1717 S. 4800 W. 1717 S. 4800 W. Principal Business Address: Salt Lake City, Utah 84104 Salt Lake City, Utah 84104 (801) 412-8890 (480) 434-1230 Telephone No.: Issuance Date: April 30, 2025. See Exhibit G for Mrs. Fields Franchising, LLC's registered agents authorized to receive service of process. I have received a disclosure document dated April 30, 2025, that included the following Exhibits: Exhibit A Financial Statements and Exhibit G State Administrators / Agents for and Guarantee of Performance Service of Process Franchise Agreement State-Specific Addenda Exhibit B Exhibit H Exhibit C Area Development Agreement Exhibit I Sample Release Form Exhibit D Form Deposit Agreement Exhibit J Disclosure Acknowledgment Agreement Table of Contents of Operations Manual State Effective Dates and Receipt Pages Exhibit E Exhibit K Exhibit F Lists of Current and Former Franchisees Date Signature Printed Name Date Signature Printed Name

Please sign this copy of the receipt, print the date on which you received this disclosure document, and return it, by mail or email, to Mrs. Fields Franchising, LLC, 1717 S. 4800 W., Salt Lake City, Utah 84104. Phone: (801) 412-8890, Email: <a href="mailto:Rhankins@famousbrandsintl.com">Rhankins@famousbrandsintl.com</a>.

### RECEIPT (YOUR COPY)

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Mrs. Fields Franchising, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to Mrs. Fields Franchising, LLC or an affiliate in connection with the proposed franchise sale, or sooner if required by applicable state law. Under Iowa law, we must give you this disclosure document at the earlier of our 1st personal meeting or 14 calendar days before you sign an agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale. Under Michigan law, we must give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. Under New York law, we must provide this disclosure document at the earlier of the 1st personal meeting or 10 business days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

If Mrs. Fields Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit G.

The franchisor is Mrs. Fields Franchising LLC, 1717 S. 4800 W., Salt Lake City, Utah 84104; (801) 412-8890. The franchise seller for this offering is:

| ☐ Rich     | Hankins                             |       | Paul Correale                         |             |        | Name of Franchise Seller:             |
|------------|-------------------------------------|-------|---------------------------------------|-------------|--------|---------------------------------------|
|            | Fields Franchising, LLC             |       | Mrs. Fields Franch                    | nising, LLC | 2      | D: : 1D :                             |
|            | S. 4800 W.                          |       | 1717 S. 4800 W.                       | ab 04104    |        | Principal Business Address:           |
|            | Lake City, Utah 84104<br>) 412-8890 |       | Salt Lake City, Uta<br>(480) 434-1230 | an 84104    |        |                                       |
| (001)      | 7 412 0070                          |       | (400) 454 1250                        |             |        | Telephone No.:                        |
| Issuance I | Date: April 30, 2025.               |       |                                       |             |        |                                       |
| See Exhib  | oit G for Mrs. Fields Fran          | chis  | sing, LLC's registe                   | ered agent  | s auth | orized to receive service of process  |
| I have rec | eived a disclosure docu             | men   | t dated April 30, 2                   | 025, that i | includ | ed the following Exhibits:            |
| Exhibit A  | Financial Statements and            | l     |                                       | Exhibit C   | G Sta  | te Administrators / Agents for        |
|            | and Guarantee of Perfor             | man   | ce                                    |             | Ser    | vice of Process                       |
| Exhibit B  | Franchise Agreement                 |       |                                       | Exhibit F   | I Sta  | te-Specific Addenda                   |
| Exhibit C  | Area Development Agree              | emei  | nt                                    | Exhibit I   | Saı    | nple Release Form                     |
| Exhibit D  | Form Deposit Agreemen               | t     |                                       | Exhibit J   | Di     | sclosure Acknowledgment Agreemen      |
| Exhibit E  | Table of Contents of Ope            | ratio | ons Manual                            | Exhibit K   | K Sta  | ate Effective Dates and Receipt Pages |
| Exhibit F  | Lists of Current and For            | mer   | Franchisees                           |             |        |                                       |
|            |                                     |       |                                       |             |        |                                       |
| Date       | Signature                           |       |                                       |             | Printe | d Name                                |
| Date       | Signature                           |       |                                       |             | Printe | d Name                                |

PLEASE SIGN THIS COPY OF THE RECEIPT, PRINT THE DATE ON WHICH YOU RECEIVED THIS DISCLOSURE DOCUMENT AND KEEP IT FOR YOUR RECORDS.